Exposure and Fetal Growth-Associated miRNA Alterations in the Human Placenta

By Matthew Alan Maccani

B.A., Dartmouth College, 2007

A Dissertation Submitted in Partial Fulfillment of the Requirements for Degree of Doctor of Philosophy in the Division of Biology and Medicine at Brown University

> Providence, Rhode Island May 2011

© Copyright 2011 by Matthew Alan Maccani

This dissertation by Matthew Alan Maccani is accepted in its present form by the Division of Biology and Medicine as satisfying the dissertation requirements for the Degree of Doctor of Philosophy.

Data		
Date		

Carmen J. Marsit, PhD, Advisor

Recommended to the Graduate Council

Date		
		Richard K. Miller, PhD (Outside Reader)
Date		
		Joseph M. Bliss, MD/PhD (Reader; Chair)
Date		
		James F. Padbury, MD (Reader)
Date		
		Jared Robins, MD (Reader)
	Approved by the	Graduate Council

Date_____

Peter M. Weber, PhD

Dean of the Graduate School

Curriculum Vitae

Matthew Alan Maccani was born in Vineland, NJ, USA on January 8th 1985. He graduated from Washington Township High School in Sewell, NJ in June 2003 with high honors as valedictorian and went on to achieve his B.A. at Dartmouth College in Hanover, NH, USA, double-majoring in "Cell and Developmental Biology and Genetics" and "German Studies". While still a student at Dartmouth, Matthew won an internship at the Maximilian Planck Institute for Chemical Ecology in Jena, Germany, and worked for a semester in the lab of Dr. Ian Baldwin on characterizing the biological and biochemical properties of genetically-modified tobacco. Back at Dartmouth, Matthew worked in the lab of Dr. Mardi Crane-Godreau and conducted research on the effects of cigarette smoke exposure on antimicrobial peptide production and function in human airway epithelial cells; Matthew had the great opportunity to present his research at the New England Immunology Conference in Woods Hole, MA in October 2006 and at the 94th Annual Meeting of the American Association of Immunologists in Miami Beach, FL in May 2007. After graduating from Dartmouth in June 2007, Matthew began his PhD work in the Pathobiology Graduate Program at Brown University in July 2007 with a summer rotation in the lab of Dr. Carmen Marsit. Matthew did his second rotation in the lab of Dr. Joseph Bliss in which he worked on a project characterizing phagocytosis of *Candida* parapsilosis by human neutrophils. In January 2008, Matthew officially joined the Marsit lab to continue his doctoral training, working to characterize epigenetic alterations in the human placenta associated with *in utero* exposures and fetal growth. After passing his preliminary exam in May 2009, Matthew became a doctoral candidate and continued his research in the Marsit lab. Matthew has presented his work at a number of scientific

iii

conferences, including the 57th Annual Meeting of the Society for Gynecologic Investigation in Orlando, FL in March 2010, the 36th Annual New England Conference on Perinatal Research in Chatham, MA in October 2010, the Clinical Epigenetics International Meeting (CLEPSO) in Homburg/Saar, Germany, in March 2011, and the Epigenetics and Developmental Programming Conference in Newcastle upon Tyne, UK in March 2011. While a PhD student, Matthew won a number of awards, including a Pre-Doctoral Training Fellowship in Environmental Pathology from the National Institute of Environmental Health Sciences, the Mander Research Pre-Doctoral Fellowship for FY 2010 for excellence in genomics and proteomics research at Brown University, the Wellcome Trust Young Investigator Bursary Award from the Epigenetics and Developmental Programming Conference in March 2011, and a Sheridan Teaching Certificate I from Brown University. Matthew has been an invited speaker at a number of conferences and research colloquia, mentored a number of undergraduates in the lab, and authored several peer-reviewed original manuscripts and reviews.

Preface and Acknowledgments

The work presented in this PhD dissertation, including the experiments, analysis, and discussion, have been executed by me. Critical collaborations have been acknowledged appropriately throughout.

This thesis describes four years of drive, desire, and dedication geared at better understanding alterations of miRNA expression and function in the placenta associated with *in utero* exposures and fetal growth. I believe this work represents an important addition to the understanding of how miRNA may be playing important roles in placental function and ultimately, how aberrant miRNA expression in the placenta may have important downstream effects on the fetal programming of the infant.

It is a humbling honor to be achieving a doctorate degree. There are a number of individuals whose guidance, mentoring, and support has made such an awesome endeavor not only possible but fruitful along the journey. My training at Brown has provided me with opportunities to grow and develop, both inside and outside of the laboratory. My advisor and mentor, Dr. Carmen Marsit, has been an exceptional mentor, challenging me to strive for excellence while at the same time keeping my best interests in mind along the way. From the early days of my summer rotation in 2007, I knew that if blessed with the opportunity to work on my thesis in the Marsit lab, it would be an amazing experience. The opportunities that Carmen has given me have broadened my horizons and have continued to cultivate my scientific and creative mind. It has been an honor to work under the guidance of such a driven, intelligent, and accomplished individual. My thesis committee, chaired by Dr. Joseph Bliss, and including Dr. James Padbury and Dr. Jared Robins, has offered guidance, helpful suggestions, and

v

constructive criticism which have proved very beneficial throughout the process.

Members of my committee have also invited me to engage in opportunities to enrich both my public speaking skills as well as to engage in scientific dialogue with colleagues and collaborators – such as presenting my work on a number of occasions at the Pediatric Research Colloquium at Women & Infants' Hospital in Providence – for which I am very grateful. My outside reader, Dr. Richard Miller, has also contributed his expertise in the areas of placental toxicology and placental biology, and I am very thankful for his contributions. I would also like to thank the members of the Marsit lab, past and present, for without their assistance – whether working out the nuances of a new protocol or aiding in the collection of samples and questionnaire data compilation – this work would be nearly impossible.

My journey as a scientist began very early in my life. My love of nature was instilled in me by my grandparents and great-grandmother, and I enjoyed exploring the wilds of the natural world surrounding our home in rural southern New Jersey. As I continued through young adulthood, my parents and family nurtured this love and supported me as I strove to make my dreams and aspirations a reality. My family also instilled in me the value of hard work; for this and for their continued love, guidance, and support, I am forever grateful. Following high school, I attended Dartmouth where I trained in the lab of Dr. Mardi Crane-Godreau. Her creativity both inside and outside of the laboratory has been an inspiration for me; Mardi's mentoring while I was a student in her lab and beyond, as well as her friendship, has been truly wonderful and has enriched my life greatly.

vi

Last but definitely not least, I would like to thank my girlfriend, Jennifer Z. Joukhadar, for her friendship, love, and support. She is truly a partner for the journey – and has made my life all the sweeter. Many thanks to my friends and family all over the globe, most especially Kiran Parkhe, Vedant Mehra, Lizzie Newton, Kristin Krogh, Iden Sinai, Andrew Robson, Weizhi Zhang, Allison Citro, Nathan Chung, and many more. Your friendship has meant the world to me.

My PhD dissertation experience was, for me, yet another example which once more underscores that timeless adage: "Life is a journey, not a destination."

Table of Contents

Abstract1
Chapter 1 – Introduction
Thesis Overview4
Fetal Programming5
The Placenta
Epigenetics12
DNA Methylation13
Imprinting15
Histone Modification17
Small Regulatory RNAs and microRNAs (miRNA)18
Placental miRNA
The Placenta as a Functional Model of Epigenetics in Fetal Programming24
References
Chapter 2 – Maternal Cigarette Smoking During Pregnancy is Associated with
Downregulation of <i>miR-16</i> , <i>miR-21</i> , and <i>miR-146a</i> in the Placenta41
Abstract43
Introduction
Results
Discussion
Materials and Methods54
Funding57
Acknowledgments

Keleiences	.59
Chapter 3 – <i>miR-16</i> and <i>miR-21</i> Expression in the Placenta is Associated with Feta	al
Growth	.71
Abstract	.73
Introduction	.74
Results	.76
Discussion	.79
Materials and Methods	.82
Acknowledgments	84
Funding	.85
References	.86
Chapter 4 – Functional Consequences of Overexpression of miRNA in Placental	
Cell Lines	.96
Abstract	.97
Introduction	00
	.98
Results1	.98 101
Results	.98 101 106
Results 1 Discussion 1 Materials and Methods 1	.98 101 106 112
Results. 1 Discussion. 1 Materials and Methods. 1 Acknowledgments. 1	.98 101 106 112 116
Results. 1 Discussion. 1 Materials and Methods. 1 Acknowledgments. 1 Funding. 1	.98 101 106 112 116
Results 1 Discussion 1 Materials and Methods 1 Acknowledgments 1 Funding 1 References 1	.98 101 106 1112 1116 .17
Results 1 Discussion 1 Materials and Methods 1 Acknowledgments 1 Funding 1 References 1	.98 101 106 112 116 17 118
Results 1 Discussion 1 Materials and Methods 1 Acknowledgments 1 Funding 1 References 1 Chapter 5 – Discussion 1	.98 101 106 112 116 117 118 .33

Appendix – Birthweight is associated with DNA promoter methylation of the		
glucocorticoid receptor in human placenta	155	
Abstract	157	
Introduction		
Results		
Discussion	163	
Materials and Methods	167	
Acknowledgments	171	
Funding		
References	173	

List of Tables

Chapter	2
Table 1.	Demographics of the study population
Table 2.	Multivariable linear regression model of individual miRNA expression68
Table 3.	miRNA targets predicted and/or experimentally validated for miR-16, mir-21,
and miR-	<i>146a</i> 70

Chapter 3

Table 1. Demographics of the study population (n=107)91
Table 2. Expression of miR-16, miR-21, miR-93, miR-135b, miR-146a, and miR-182
determined through qRT-PCR in 107 primary human term placenta samples92
Table 3. Logistic regression for the association between individual miRNA expression
and SGA status94
Table 4. Logistic regression to examine the interaction of low <i>miR-16</i> and low <i>miR-21</i>
expression on the association with SGA status

Chapter 4

Table 1.	Predicted and empirically-confirmed gene targets	of <i>miR-16</i> , <i>miR-21</i> , and <i>miR-</i>
146a		124

Appendix

Table 1.	Demographics of the	study population	
----------	---------------------	------------------	--

Supplemental Table 1. Pearson correlation coefficients and corresponding p-values
examining the correlation of each of the covariates with the log GR mean methylation
extent
Table 2. Multivariable linear regression model of log-transformed mean GR methylation
extent

List of Figures

Chapter 1

Figure 1. Basic development of the human embryonic and extraembryonic tissues.....40

Chapter 2

Figure 1.	Maternal cigarette smoking during pregnancy is associated with	
downregul	lation of <i>miR-16</i> , <i>miR-21</i> , and <i>miR-146a</i>	.67
Figure 2.	miR-146a is downregulated in TCL-1 cells exposed to nicotine and	
benzo(a)p	yrene	.69

Chapter 3

Figure 1. Distribution of infant birthweights (y-axis) by primary term human placenta miRNA expression quartiles (x-axis) determined through qRT-PCR for (A) *miR-16* (p=0.04), (B) *miR-21* (p=0.02), (C) *miR-93* (p=0.88), (D) *miR-135b* (p=0.84), (E) *miR-146a* (p=0.46), and (F) *miR-182* (p=0.55). Black bars indicate median of birthweight percentile within each quartile. Kruskal-Wallis tests revealed that birthweight percentile significantly differed across quartiles of *miR-16* and *miR-21* expression (p<0.05)......93

Chapter 4

Supplemental Figure 1. Migration assays using the technique of staining with hematoxylin and cell counting described previously to confirm results trending significance (p<0.2) as determined using MTS to quantify migrated cells (see Figure 5).
3A cells overexpressing *miR-21* exhibited a trend of increased invasion, 150% versus control (p<0.08), confirming previous results using MTS to quantify migrated cells....130

Appendix

Figure 1. Comparison of the mean extent of methylation of the 13 CpG sites in the GR
exon 1F promoter region in SGA (n=102), AGA (n=343), and LGA (n=35) placentas
determined by bisulfite pyrosequencing181
Figure 2. Correlation matrix describing Pearson correlations of methylation status
among the 13 CpG sites within the exon 1F promoter region of the GR in all placenta
samples (n=480)
Figure 3. Examination of the effects of 5-az-2' deoxycytidine (5-azaC) on placenta
choriocarcinoma cell line JEG-3184

Abstract of Exposure and Fetal Growth-Associated miRNA Alterations in the Human Placenta, by Matthew Alan Maccani, Ph.D., Brown University, May 2011

Exposure of the developing fetus to harmful chemicals, such as xenoestrogens, heavy metals, alcohol, and cigarette smoke, is associated with poor fetal and developmental outcomes. The overarching hypothesis of this work is that environment-associated differential expression of placental miRNA has consequences for the growth and development of the placenta and fetus and plays a role in fetal programming. In 25 human placentas, qRT-PCR revealed that miR-16, miR-21, and miR-146a were significantly downregulated in cigarette smoke-exposed placentas (n=17) compared to controls (n=8), and TCL-1 cells exposed to both nicotine and benzo(a)pyrene exhibited significant, downregulation of *miR-146a*. In an independent, population-based birth cohort, 107 human placenta samples were examined for the expression of 6 miRNA expressed in the placenta and previously shown to regulate cell growth and development pathways – miR-16, miR-21, miR-93, miR-135b, miR-146a, and miR-182 – and were investigated for associations of fetal growth and miRNA expression. The expression of miR-16 and miR-21 was markedly reduced in infants with the lowest birthweights (p<0.05). Logistic regression models suggested that low expression of *miR-16* in the placenta predicts an over 4-fold increased odds of small for gestational age (SGA) status (p=0.009). Moreover, having both low miR-16 and low miR-21 expression in the placenta predicts a greater increase in odds for SGA than having just low miR-16 or miR-21 expression (p<0.02), suggesting an additive effect of both of these miRNA. Placental cell lines transfected to overexpress miRNA of interest were utilized to analyze changes in cell viability, growth, proliferation, migration, and invasion, as well as to empirically

validate miRNA targets predicted using in *silico* bioninformatic approaches.

Overexpression of *miR-21* in 3A cells resulted in a trend to increased cell migration. TCL-1 cells overexpressing *miR-21* exhibited a 50% reduction in protein levels of PTEN (p<0.05), and 3A cells overexpressing *miR-146a* exhibited a 25% reduction in STAT-1 protein levels (p<0.05). Taken collectively, this work further elucidates the association of placental miRNA expression with *in utero* exposures and fetal growth and seeks to better understand how an adverse intrauterine environment acts through epigenetic mechanisms to alter infant and child health. Chapter 1

Thesis Overview and Introduction

Thesis Overview

The study of epigenetics is focused on understanding the control of gene expression beyond what is encoded in the DNA sequence. Central to growing interest in the field is the hope that more can be learned about the epigenetic regulatory mechanisms underlying processes of human development and disease. Researchers have begun to examine epigenetic alterations to learn more about epigenetic regulation in the placenta. A number of studies are now making important links between alterations to appropriate epigenetic regulation and diseases of gestation and early life; promising links have been examined between environmental exposures and altered expression of microRNA (miRNA), small RNAs involved in epigenetic mechanisms of gene regulation. Examining changes in miRNA expression associated with environmental exposures and fetal growth will prove especially important in providing critical insights into the biology of development and response to in utero exposure. Furthermore, miRNA expression profiles in the placenta may aid in determination of *in utero* exposures, as well as future disease risk assessment. This thesis aims to analyze changes in miRNA expression in the human placenta associated with *in utero* exposure and fetal growth and seeks to determine the cellular consequences of such changes in miRNA expression to ultimately understand potential downstream effects on infant growth and development.

This thesis begins with a broad introduction to the placenta and to the endocrine and chemical processes it controls and regulates. The introduction to the thesis continues with a definition of epigenetics, a description of the field, the main modes of epigenetic gene regulation, and the strides that are being made in the study of placental epigenetics. The introduction continues with a discussion of miRNA and work that has been done to

better characterize placental miRNA and their crucial involvement in ensuring the proper growth and development of the fetus. Chapter two details the results of a study which sought to characterize the modulation of placental miRNA by maternal cigarette smoking during pregnancy by analyzing the expression of four candidate miRNA for associations with maternal cigarette smoking during pregnancy. Chapter three describes the results of a study which investigated associations of the expression of six candidate miRNA in the placenta with fetal growth and attempted to elucidate some of the implications of aberrant expression on the downstream growth and development of the fetus. Chapter four shows data from work aimed at better understanding the effects of aberrant miRNA expression on the placental cell level, further suggesting that miRNA play important roles in regulating cell processes crucial for the proper function of placental cells during various stages of pregnancy. Finally, Chapter five summarizes the conclusions of the previous chapters, highlights the relevance of these findings, and discusses future directions for this work.

Fetal Programming

Our understanding of the interplay between genes and the environment is being greatly enhanced in the post-genome era. There are only a few settings where the importance of this gene-environment interface is more profound than during intrauterine development, where the "critical windows" are narrower and where disruption or modification can influence fetal development as well as lead to programming of health throughout the life course. Over the years, many have speculated that adverse intrauterine conditions might play a role in determining not only the growth and development of the fetus but also the health of the child and adult.

A number of epidemiological studies exploring links between adverse intrauterine conditions and diseases or conditions later in life have used as their cohort of interest the Dutch Famine Birth Cohort, which consists of men and women born as term singletons in Amsterdam, The Netherlands (1, 2). The formation of the Dutch Famine Birth Cohort resulting from the Dutch Famine of 1944-45 provided researchers with an opportunity to study starvation during pregnancy and a number of health and developmental outcomes potentially associated with this intrauterine exposure. As described by Stein, et al.(2), the Nazi food blockade and embargo of part of the Netherlands leading to the Dutch Famine was quite remarkable in a number of respects. Extensive data existed on a number of conditions of the social environment at the time of the famine, the definitive demarcation of the famine in both time and place, and the degree of nutritional deprivation was known with a great degree of precision (2). Throughout the 6 months of famine, the population's daily rations were 400-800 calories and at the height of the famine, the official food rations of the famine area were 450 calories per day, considered only 25% of the minimum standard while cities and towns outside of the famine area had daily caloric intake of approximately 1300 calories (1, 2). In developing the Dutch Famine Birth Cohort, these nutrition levels as well as a number of other key confounders were assessed and included in the data. Associations between adverse intrauterine environment as represented by famine and a number of diseases and conditions, including but not limited to type 2 diabetes mellitus, cardiovascular disease, other metabolic disorders, and decreased cognitive function later in life (1, 3, 4) have been reported in the years since

the creation of the Dutch Famine Birth Cohort. In addition, researchers found that maternal weight loss or moderate to low weight gain was significantly associated with infant birthweight, length, and ponderal index, as well as with trimester of exposure to famine, controlled for confounders (5). Furthermore, it was noted that among women whose weight changed more than 0.5 kg/week the "association between third-trimester weight change and birthweight among mother-daughter pairs exposed to famine in early or mid-pregnancy was stronger than the association observed among the unexposed cohort or among those exposed only late in pregnancy" (5). Additional observations of increased risk of affective disorders among males exposed to famine during their second trimester of gestation have been characterized (6), further underscoring the notion that timing of prenatal exposure to famine as well as potential confounding elements, such as gender of the infant, may be playing important roles in modulating the programming of Birth cohorts such as the Dutch Famine Birth Cohort have provided the fetus. researchers with many of the tools necessary to investigate epidemiological associations between adverse intrauterine conditions and postnatal health and disease.

Observations and associations reported in studies on the Dutch Famine Birth Cohort led many to further hypothesize that adverse intrauterine conditions could play a part in programming the fetus. The phenomenon now known as "fetal programming" is a model of gene-environment interaction and can inform the mechanistic basis of the synergistic effect(s) of the environment and the molecular character of development (4, 7). One of the forefathers of the theory of fetal programming is David Barker who, starting in the 1980s and 1990s, proposed and tested the hypothesis that an adverse fetal environment but plentiful food in adulthood might be a major factor associated with a number of adult diseases (8). This "Barker Hypothesis" as it became known further hypothesized that adverse intrauterine conditions may result in a negative or poor maternal forecast, commonly manifested in small for gestational age (SGA) status or reduced infant birthweight. This poor maternal forecast predicts that the child will be born into a postnatal environment in which resources are scarce, and thus, the child has been forecasted or programmed to thrive in such a poor environment. Poor maternal forecasts can prove incorrect if a child is born into what is or soon becomes a nutrientrich environment.

Several studies suggest that incorrect maternal forecasts may be linked to a number of diseases later in life, including type 2 diabetes mellitus, cardiovascular disease, other metabolic disorders, and decreased cognitive function later in life (1, 3, 4). A growing body of evidence suggests the importance of prenatal growth and development on later morbidity, further suggesting a link between developmental responses to *in utero* environments and adult biological outcomes (9). Current and future work in the developmental origins of health and disease is continuing to investigate the molecular mechanisms of fetal programming, both in epidemiological studies and model systems, often focusing on one of the most important tissues responsible for regulating *in utero* development – the placenta.

The Placenta

Throughout the *in utero* development of the fetus, the placenta is of utmost importance to ensure proper growth and development. The placenta begins development at the time of fertilization (10). Four days after fertilization, the morula enters the uterus. As uterine fluid penetrates its outer layer, the zona pellucida, a blastocyst with a cavity is formed; further penetration results in the hatching of the blastocyst into two parts, the inner cell mass and the trophoblast. The blastocyst is then bathed in uterine secretions that provide the embryo with oxygen and key metabolites; soon this becomes inadequate for further embryonic development and the embryo must implant in the uterine wall (10). Invasion of the uterine wall begins at this stage (11). Proper invasion is crucial to further placental development in a normal pregnancy.

Following implantation of the blastocyst, trophoblast cells begin to differentiate, migrate, and invade into the uterine stroma. The cytotrophoblast stem cells either fuse to form the syncytiotrophoblast or aggregate to form anchoring villous trophoblast cells. Anchoring villous trophoblast cells give rise to a sub-population known as extravillous trophoblast (EVT) cells which invade the uterine wall and its blood vessels, particularly the spiral arteries. Both interstitial invasion of the stroma and endovascular trophoblast invasion of the spiral arteries occur (11). EVT cells are generated from cell columns at the tips of anchoring villi with resulting cells migrating deeper into maternal tissues (11). The EVT are responsible for invasion, thereby anchoring the placenta to the decidua and myometrium. Additionally, the EVT cells remodel the maternal spiral arteries in a process that spans the first few weeks in pregnancy; while the specifics of this process remain controversial, spiral artery remodeling seems to follow a mechanism displacing smooth muscle and endothelial cells, in order to produce a blood vessel capable of allowing increased blood flow and reduced resistance (12). This is an essential step in establishing and maintaining a normal pregnancy and is necessary for the higher blood requirement of the fetus later in pregnancy (12). Shallow or incomplete trophoblast

invasion and lack of or limited spiral artery modification have been implicated in complications of pregnancy such as preeclampsia and intrauterine growth restriction (12, 13).

Formed from fused cytotrophoblast stem cells, the syncytiotrophoblast is a multinucleated but continuous cell layer which ultimately covers the surface of the placenta and is a site of metabolic activity resulting in one function of the placenta as a barrier (though not completely impervious to insult) (10). The syncytiotrophoblast is a specialized epithelium covering the villous tree and has several functions, such as transport of gases, nutrients, and waste products and synthesis of peptide and steroid hormones that regulate placental, fetal, and maternal systems. Alterations in villous trophoblast differentiation are seen in various pathophysiological situations and may underlie several pregnancy disorders (10).

Fetal circulation begins around 6 weeks (14) and maternal flow starts as intravascular plugs are dissolved and endovascular trophoblast migration into spiral arteries is completed (15). It is at this stage that most consider the placenta as fully functional with both fetal and maternal circulations established and functioning (11). During the stages of further placental growth and development, intermediate and terminal villi branch off from the main stem villi, forming what is considered to be a system of villous trees (11). Additionally, the placenta subdivides into functional units called placental lobules (11). As an overall summary, basic development of the human embryonic and extraembryonic tissues is described schematically by **Figure 1** (16).

Called by some a hallmark of mammalian development (17), the functional placenta provides the fetus with nutrients, allows for waste to be transferred and

ultimately excreted by the mother, and protects the fetus from what would otherwise be a type of immune detection which would result in an attack of the placenta by the maternal immune system. In addition, the placenta has a degree of metabolic and endocrine activity, is involved in secreting hormones responsible for maintaining and regulating various stages of pregnancy, and performs biochemical reactions to protect, whenever possible, the fetus from exposure to toxicants or other harmful chemicals (18). All of these functions of the placenta as well as placental gene expression thus respond to and are marked by environmental insults (18, 19), and in many ways, the placenta can serve as a record of *in utero* exposure and pathology (20). Various compounds and drugs, including but not limited to alcohol (21), nicotine (22), cocaine (23) (24), lead (25), and phthalates (26), have been shown to cross the placenta and alter placental gene expression; some even accumulate in the placental tissue. Depending on the timing and type of environmental insult, such insults may affect various aspects of placental development and function and can have a variety of ramifications on the developing fetus.

One of the most common environmental exposures during pregnancy is maternal cigarette smoking during pregnancy. Maternal cigarette smoking has been reported to be associated with increased risk for spontaneous abortion (27) and preterm delivery (27-29). Previous studies have shown that there are placental complications linked to cigarette smoke exposure during pregnancy, including alterations to the development and function of the placenta (30). There are more than 4,000 chemicals in a cigarette, including nicotine, benzo(a)pyrene, and carbon monoxide; more than 43 of these chemicals are known carcinogens (31). Nicotine readily crosses the placenta and can

result in fetal concentrations that are 15% higher than maternal concentrations (32). While a number of studies have shown a decrease in overall prevalence of smoking in women in the past 20 years, the prevalence of smoking in young pregnant women has increased (33, 34). Additional studies have reported that 12-15% of all women smoke during their pregnancies (35, 36). Taken collectively, these observations suggest that maternal smoking during pregnancy remains an important common exposure that can have major ramifications on not only the normal growth and development of the fetus but also on fetal programming.

Studies continue to reveal how developmental plasticity and the developmental origins of health and disease may be mechanistically linked to the placenta (37). Also, data have linked the placenta to fetal cardiovascular adaptations(38), fetal body composition(39, 40), and fetal endocrinology and metabolism(40). Investigations are underway to determine changes in the genetics and epigenetics of the placenta which are characteristic of such exposures and pathological responses resulting from these exposures. Many of these investigations are revealing the epigenetic mechanisms underlying molecular fetal programming.

Epigenetics

Epigenetics is broadly defined as the field of research which studies changes in gene expression that are not caused by changes in the sequence of DNA (41), and the field has seen relatively rapid growth over the past few decades, accelerated by advancements in molecular biology, biotechnology, and genomics. The emergence of a new field – namely, "environmental epigenetics" (42) – combines the traditional ways of

studying epigenetics with the understanding that environmental exposures affect such Research in fetal programming and many other epigenetic mechanisms as well. disciplines is now focusing on the paradigm that gene regulation occurs beyond the DNA sequence. Most of the acquired adverse effects related to intrauterine environment cannot be due to genetic alterations. This critical role of *epigenetic* regulation, the mitotically and meiotically heritable control of gene expression not related to DNA sequence, during development is becoming increasingly appreciated. Thus, an understanding of changes to the cellular epigenome is at the interface of the interaction between genes and environment, and can provide a mechanistic basis for the synergistic effects. Research in model systems and now expanding to human studies has suggested that the causes and consequences of a variety of pathologies are related to environmental influence on epigenetic regulation. Examination of the specific molecular character of these epigenetic alterations in perinatal development has been less comprehensive. As the field of epigenetics continues to grow and be defined, there remains a central focus on examination of 4 main modes of epigenetic regulation: DNA methylation, imprinting, histone modification and small RNA-mediated control, specifically miRNAs.

DNA Methylation

DNA methylation has become the most heavily studied mode of epigenetic regulation (41). In brief, DNA methylation is carried out by one of a variety of DNA methyltransferases responsible for adding a methyl group to cytosine residues in cytosine/guanine-rich regions of DNA (called "CpG islands"). A general rule (one that is usually, but not always, true) is that when a given stretch of cytosines in a CpG island or islands located in the promoter region of a gene is methylated, that gene will be effectively silenced by methylation; such a CpG island would be termed "hypermethylated". Conversely, when a given stretch of cytosines in a CpG island or islands located in the promoter region of a gene is not methylated, that gene will not be silenced by methylation; the CpG island in this case would be said to be "hypomethylated". It should also be noted that it is not the methylation of DNA itself which contributes to transcriptional repression but rather the binding of various elements (proteins that act as transcriptional repressors, proteins that block the movement of RNA polymerase, etc.) to methylated stretches of DNA that most greatly contribute to the transcriptional repression characteristic of genes with methylated CpG islands in their promoters. Researchers continue to work on attempting to decipher a type of chromatin code – one that may give scientists clues as to what degree of promoter methylation as well as interaction with histone post-translation modifications may be necessary to silence a particular gene.

Throughout the development of the embryo, important resetting of methylation patterns of germline and somatic lineages occur. Methylation throughout the genome of the zygote is almost completely removed during the cleavage phase of development; in between the implantation and gastrulation phases of development, *de novo* methylation reestablishes the developing organism's methylation patterns which, under normal conditions, are maintained throughout the rest of the organism's life (17, 43-45). This patterning is not limited to the embryo, but also occurs in a specific fashion in the extraembryonic lineages, although the overall levels of methylation in extraembryonic cell lineages are significantly lower than that in the somatic lineage (17). Crucial for the

health and survival of the organism is the need for the appropriate removal and resetting of methylation patterns during development thereby making this period a critical window during which the environment can have profound effects on the epigenetic pattern of the offspring.

Imprinting

Epigenetic regulation is central to the phenomenon of genomic imprinting, the parent-of-origin, allele-specific expression of genes. Genes controlled through imprinting are often located and regulated coordinately in clusters. Imprinted genes are theorized to be controlled at differentially methylated regions (DMRs) by DNA methylation(46). One type of DMR is one that is differentially methylated in all tissues throughout development and is commonly called an imprinting control region (ICR) because such ICRs are hypothesized to be key regulators of imprinting in their particular chromosomal domains (47). The other type of DMR is one that has differential patterns of tissue-specific methylation during stages of somatic development (47).

Both non-coding RNAs and changes in DNA methylation at sites in DMRs are responsible for the regulation of the imprint. Although DNA methylation is involved, it does not function in a manner similar to that seen in promoter regions, but instead functions to alter the binding of specific transcription factor and/or enhancer elements which control the allele-specific expression of the region (48, 49). The marks of imprinting are erased in germline cells, and re-established dependent on the sex of the individual (i.e. in sperm, paternal imprints become established and in oocytes, maternal imprints) (50).

Imprinting has been theorized to be one of the mechanisms involved in the socalled "parent conflict" theory (51). The "parent conflict theory" suggests that paternally expressed genes strongly favor using maternal resources to benefit offspring while maternally expressed genes attempt to preserve such maternal resources and thus, are in direct conflict with one another (51). In such a way, one could argue that paternally expressed (and maternally imprinted) genes would work to foster the growth of offspring while maternally expressed (and paternally imprinted) genes would function to better ensure that each offspring has approximately the same access to maternal resources as its siblings (52).

Imprinted genes are thought to function in the control of embryonic development, including placental development (53), as well as in functions later in life such as behaviors and metabolism (54). Alterations to normal imprinting patterns lead to well characterized syndromes, including Praeder-Willi and Angelman Syndromes related to inappropriate imprinting at chromosome 15q11. Developmental environment has also been suggested to affect the appropriate establishment of imprinted genes, particularly at chromosome 11p15 leading to Beckwith-Wiedemann Syndrome, which has been linked to the use of assisted reproductive technologies (55-57). Although methylation patterns are critical in the regulation of genomic imprinting, their utility to examine imprinted genes is limited by the ability to determine allele-specific methylation patterns. Most often, imprinting status is determined using allele specific PCR reactions to examine allele specific expression (58), although new methods based on genome-wide SNP arrays are allowing for the examination of allelic-specific expression and alterations to normal imprinting status on the genomic scale (59, 60), and may be useful for the examination of

imprinting patterns in the placenta and their association with normal fetal growth and development.

Histone modification

Modifications of the chromatin environment play key roles in epigenetic regulation of gene expression as well. One of these epigenetic regulatory mechanisms involves the acetylation, methylation, phosphorylation, and ubiquitinylation of histones, leading to regulation of gene expression (61, 62). The combined effects of the modification of the amino-terminal tails of core histones by acetylation, phosphorylation, and methylation play a major role in determining gene activity (63-65). A number of classes of histone methyltransferases – key enzymes involved in the transfer of methyl groups to histones – have been discovered (64), including the H3-K4 methyltransferase (66, 67) and five H3-K9 methyltransferases (68-71). Additionally, researchers have identified a number of transcription co-activators that have characteristic histone acetyltransferase (HAT) activity and histone deacetylases (HDACs), both of which play important roles in histone modification (72).

Histone modifications can be established in particular reactions or in a sequential order (63, 65); recent studies have indicated that the sequential order of modifications may be gene specific (73, 74). Other research has suggested that site-specific combinations of covalent histone modifications may comprise a type of histone code that can not only affect the structure of chromatin but can also affect targeting of transcriptional complexes (75-77). Such histone modifications can lead to gene activation or gene silencing, depending on the effects on transcriptional complexes.

Alterations to patterns of histone modification can have a number of negative consequences, such as developmental dysregulation, X-chromosome inactivation, or might lead to a number of diseases (72). Research is continuing to better define how the patterns of histone modifications are utilized by the cell to control gene expression, as well as how these marks are involved in regulating additional epigenetic processes (78).

Small regulatory RNAs and microRNAs (miRNA)

In the early 1990s, researchers first published observations characterizing two small regulatory RNAs, known as lin-4 and let-7, which were shown to control the timing of larval development in C. elegans (79, 80). These RNAs, initially termed "lin-4 and *let-7* RNAs", were initially suggested to represent a class of endogenous RNAs found in worms, flies, and mammals and since have been renamed "microRNAs (miRNAs)" (81-83). Subsequent work suggested that these small regulatory RNAs could be found in plants, mammals, green algae, and viruses (84). Other classes of small RNAs have been found in plants, animals and fungi; small interfering RNAs (siRNAs) (85, 86) and Piwiinteracting RNAs (piRNAs) (87) are two examples. miRNAs are different from these other classes of small RNAs in that they are formed from transcripts that have been shown to fold back on themselves, generating characteristic hairpin structures (88); other small RNA classes are formed from longer hairpins (siRNAs) or from precursor forms lacking a double-stranded nature (piRNAs) (88). Generally, as these small regulatory RNA molecules can alter gene and protein expression without altering the underlying genetic code, they too are considered critical mechanisms in epigenetic regulation.

miRNA are transcribed by RNA Polymerase II as part of transcripts called primary miRNAs (pri-miRNAs) and include 5' caps and 3' poly(A) tails (89-91). The miRNA portion of the pri-miRNA then forms a hairpin (89). The pri-miRNA is then digested by the dsRNA-RNA-specific ribonuclease Drosha, and the hairpin that is ultimately released is called precursor miRNA (pre-miRNA) (92). pre-miRNA has been characterized to be 70-75 nucleotides of RNA in length with 1-4 nucleotide 3' overhangs, 25-30 base pair stems, and small loops (92, 93). Data have also suggested that Drosha processes either the 5' or 3' terminus of the mature miRNA, depending on which strand of the pre-miRNA associates with the RNA-induced silencing complex (RISC) (92, 93). The pre-miRNA is then exported from the nucleus to the cytoplasm by a complex containing Exportin-5 (Exp5) (93, 94). After arrival in the cytoplasm, the pre-miRNA is cleaved by Dicer, an RNase III superfamily member (92, 93). After cleavage by Dicer, the resulting double-stranded RNA has short 3' overhangs at either end (94). It should be noted that only one of the two strands in the post-Dicer-processed dsRNA is the true mature miRNA; some mature miRNAs are formed from the leading strand of the miRNA strand while others are formed from the lagging strand, in a mature miRNA-specific fashion (94). In order to effectively control the translation of target mRNAs, the dsRNA that Dicer has processed must be separated into two strands, and the single-stranded mature miRNA has to associate with the RISC in order to be trafficked to its mRNA target (95). Researchers have shown that determination of the active strand in the ds-RNA has a direct relationship with the stability of the ends of the dsRNA (96, 97); in brief, their work revealed that the strand with less stable base pairing of the 2-4

nucleotides at the 5' end of the duplex associates with RISC and ultimately takes on the role as the active miRNA strand (96).

Data have suggested that miRNA regulate gene expression by base-pairing to a target mRNA transcript; the exact mechanism for this post-transcriptional gene regulation varies depending on a number of factors, the most noteworthy of which seems to be the degree of complementarity of the miRNA to its target mRNA sequence (81). The active strand of the mature miRNA associates more specifically with the Argonaute protein of the RISC and upon trafficking to the target mRNA, participates in post-transcriptional repression (95, 98). As a general rule, a miRNA with perfect complementarity to its target mRNA will cause the degradation of the mRNA transcript through Argonautecatalyzed mRNA cleavage (95, 99, 100) while miRNA with imperfect complementarity to a target mRNA will cause translational repression by blocking or altering the normal function of machinery that would otherwise aid in translation of mRNA into protein (81). Mechanisms by which translational repression is carried out include inhibition of translation initiation and poly(A) shortening (101). Some groups have also shown data which suggest that miRNA can use a combination of both mRNA degradation and disruption of translation to carry out post-transcriptional repression (102).

miRNA have been shown to carry out important roles in a number of responses to stress and disease. Work by van Rooij and colleagues demonstrated that a cardiacspecific miRNA, miR-208, was crucial for hypertrophy of cardiomyocytes and fibrosis in response to stress and hypothyroidism (103). Findings such as this have led several to hypothesize that there may be important miRNA-based therapies for heart disease that have yet to be developed (104). Several groups have suggested roles for miRNA in
preventing or even contributing to the development and progression of cancer (105). miR-21, shown to be upregulated in human brain tumor glioblastoma, has also been shown to have anti-apoptotic properties in human glioblastoma cells (106). Several other miRNAs have been described as having oncogenic or even tumor-suppressive characteristics, and many have been selected for use in developing cancer-specific therapies (105).

There exist a number of technologies and methods that have given researchers the power to measure even the smallest of changes in miRNA expression. Reverse transcription quantitative real-time PCR has been used to determine changes in the expression of particular miRNA associated with particular exposures or diseases. Highthroughput assays, including miRNA microarrays, have been used to interrogate the expression of thousands of reported and predicted miRNA sequences in tissues that have a particular exposure or disease. Downstream effects of miRNA on post-transcriptional gene regulation remain more challenging and require use of bioinformatics approaches to first predict mRNA targets of specific miRNA and then confirm the effects of over- or underexpression of miRNA on that particular target. Due to the miRNAs' ultimate role in controlling protein translation, true confirmation of targets requires examination of the proteins of interest using specific antibodies, or through in-vitro approaches coupling targeted miRNA binding regions to reporter constructs. The understanding of miRNAs' role in the human placenta is in its infancy but due to the known critical role of miRNA in human development, it is certainly an attractive and exciting field of study.

21

Placental miRNA

Several groups have published data demonstrating that miRNA expression is tissue-specific and that several miRNAs are expressed in the human placenta (102, 107). Since such discoveries, much interest has been generated for investigating the involvement of miRNA in placental gene regulation and the possible utility of discovering placental miRNA which can serve as clinical biomarkers of exposure or disease.

Pineles and colleagues measured miRNA levels in placentas to investigate whether placental miRNA expression patterns are associated with preeclampsia or small for gestational age diagnoses (108). This work revealed differential expression of *miR*-*210* and *miR-182* in preeclampsia versus control patients and showed key associations in miRNA expression patterns and preeclampsia, furthering efforts to find a miRNA biomarker of preeclampsia pathology (108). Zhu and coworkers found that thirty four miRNAs were differentially expressed in preeclamptic placentas compared to normal placentas further suggesting a role for miRNA in the pathogenesis of preeclampsia (109). Additional studies have been conducted linking differential miRNA expression in placental tissue with preeclampsia (110). Such biomarkers would prove especially useful in giving clinicians and pathologists sensitive molecular tools to better diagnose preeclampsia, a disorder that has great impacts on both pregnant women and their newborns.

It has long been speculated that varying oxygen levels have effects on the genetic and epigenetic control mechanisms involved in placental growth and cell survival – and ultimately, on the health and survival of the developing fetus. To investigate this further,

22

Donker and coworkers analyzed the relationships between expression of Argonaute 2, an important RNAi enzyme, and other miRNA in trophoblasts and in environments with varying oxygen level (111). Donker and colleagues' data showed that not only is the miRNA processing machinery present and functional in human trophoblasts but that varying expression of *miR-93* and *miR-424* is associated with different levels of oxygen (111). Such data may prove especially helpful in determining whether particular placental abnormalities – and ultimately, fetal abnormalities – may be associated with aberrant levels of oxygen at particular critical windows of development.

Some groups have observed and detected placental miRNA in maternal plasma and have conducted comparative studies on circulating miRNA and other circulating nucleic acids (CAN) in the maternal sera of pregnant and non-pregnant women (112, 113). Chim and colleagues showed that placental miRNAs (*miR-141*, *miR-149*, *miR-299-5p*, and *miR-135b*) were highly expressed in maternal plasma in pregnant mothers and suggested that such expression patterns could serve as important biomarkers for monitoring pregnancy (112). Gilad and coworkers demonstrated that placental miRNA levels in sera from pregnant women were higher than those in sera from non-pregnant women; moreover, levels of miRNA in sera of pregnant women correlated with pregnancy stage (113). Thus, placental miRNA levels detectable in maternal serum may serve as important clinical biomarkers of pregnancy, pregnancy stage, and other pregnancy-related outcomes.

The Placenta as a Functional Model of Epigenetics in Fetal Programming

Data have suggested that environmental exposures can alter miRNA expression in in-vitro systems (114) and various tissues (42) and that altered levels of miRNA expression can be associated with particular diseases or risk factor for disease (115). As reviewed above, work has also been done to investigate biomarkers of diseases, such as cancer, growth retardation, and other records of *in utero* environment in placental patterns of DNA methylation, imprinting, histone modification, and miRNA expression. Studies aimed at further revealing the importance of miRNA in responding to environmental exposure and to disease may prove very useful in better understanding how such exposures and diseases affect the body. Current and past research into diseases of pregnancy – such as finding miRNA associated with preeclampsia (108, 109) or genespecific placental methylation patterns associated with gestational trophoblastic disease (GTD) (116) – might be expanded to include work to better understand associations between epigenetic factors in the placenta and diseases such as choriocarcinoma, childhood cancers, and other diseases of childhood and adolescence, as well as alterations to normal placental immunology and function.

Using the placenta as a reference tissue allows researchers to utilize an important residual tissue whose respective miRNA expression and DNA methylation patterns may prove to be powerful biomarkers possessing predictive capability for a number of diseases or disease progression. Modern advances in bioinformatics – such as pathway analysis tools and target gene prediction software – as well as advances in technology, such as microarray technology, have given researchers and clinicians tools to better detect patterns in DNA methylation, imprinting, histone modification and miRNA

24

expression that are associated with particular exposures and diseases. Aberrant patterns of miRNA expression or DNA methylation may ultimately serve as biomarkers for exposure, disease burden, or even as "early indicator" diagnostics of increased risk for developing future disease or disorders.

Important advances in placental epigenetics continue to elucidate a better understanding of the epigenetic regulatory mechanisms of in the placenta. Knowledge of such epigenetic mechanisms may be useful in identifying novel biomarkers for exposure, burden, or risk for disease. Such biomarkers may prove essential for developing new diagnostics for early diagnosis of risk factor and levels of exposure. Additionally, these aberrant patterns of miRNA expression, imprinting, DNA methylation, or histone modification may identify previously unknown pathways targeted for alteration, which, in turn, may serve as targets for novel drug treatment or prevention strategies. These epigenetic biomarkers can be brought from the benchtop to the bedside and will be useful in helping clinicians better diagnose and prevent the onset of disease.

REFERENCES

- de Rooij, S.R., Wouters, H., Yonker, J.E., Painter, R.C. and Roseboom, T.J.
 (2010) Prenatal undernutrition and cognitive function in late adulthood. *Proc Natl Acad Sci U S A*, **107**, 16881-16886.
- 2 Stein, Z., Susser, M., Saenger, G. and Marolla, F. (1972) Nutrition and mental performance. *Science*, **178**, 708-713.
- 3 Argente, J., Mehls, O. and Barrios, V. (2010) Growth and body composition in very young SGA children. *Pediatr Nephrol*, **25**, 679-685.
- 4 Barker, D.J. and Clark, P.M. (1997) Fetal undernutrition and disease in later life. *Reviews of reproduction*, **2**, 105-112.
- 5 Stein, A.D., Ravelli, A.C. and Lumey, L.H. (1995) Famine, third-trimester pregnancy weight gain, and intrauterine growth: the Dutch Famine Birth Cohort Study. *Hum Biol*, **67**, 135-150.
- Brown, A.S., Susser, E.S., Lin, S.P., Neugebauer, R. and Gorman, J.M. (1995)
 Increased risk of affective disorders in males after second trimester prenatal
 exposure to the Dutch hunger winter of 1944-45. *Br J Psychiatry*, **166**, 601-606.
- Hales, C.N. and Barker, D.J. (1992) Type 2 (non-insulin-dependent) diabetes
 mellitus: the thrifty phenotype hypothesis. *Diabetologia*, 35, 595-601.
- 8 Schulz, L.C. (2010) The Dutch Hunger Winter and the developmental origins of health and disease. *Proc Natl Acad Sci U S A*, **107**, 16757-16758.
- 9 Hochberg, Z., Feil, R., Constancia, M., Fraga, M., Junien, C., Carel, J.C., Boileau,
 P., Le Bouc, Y., Deal, C.L., Lillycrop, K. *et al.* (2010) Child Health,
 Developmental Plasticity, and Epigenetic Programming. *Endocr Rev*.

- Benirschke, K. and Kaufmann, P. (1995) *Pathology of the human placenta*.Springer-Verlag, New York.
- Pijnenborg, R., Vercruysse, L. and Brosens, I. (2011) Deep placentation. *Best Pract Res Clin Obstet Gynaecol.*
- 12 Cartwright, J.E., Kenny, L.C., Dash, P.R., Crocker, I.P., Aplin, J.D., Baker, P.N. and Whitley, G.S. (2002) Trophoblast invasion of spiral arteries: a novel in vitro model. *Placenta*, 23, 232-235.
- Whitley, G.S. and Cartwright, J.E. (2010) Cellular and molecular regulation of spiral artery remodelling: lessons from the cardiovascular field. *Placenta*, **31**, 465-474.
- Jauniaux, E., Watson, A.L., Hempstock, J., Bao, Y.P., Skepper, J.N. and Burton,
 G.J. (2000) Onset of maternal arterial blood flow and placental oxidative stress. A
 possible factor in human early pregnancy failure. *Am J Pathol*, **157**, 2111-2122.
- Hustin, J. and Schaaps, J.P. (1987) Echographic [corrected] and anatomic studies of the maternotrophoblastic border during the first trimester of pregnancy. *Am J Obstet Gynecol*, **157**, 162-168.
- 16 (2001). Department of Health and Human Services.
- Jaenisch, R. (1997) DNA methylation and imprinting: why bother? *Trends Genet*,13, 323-329.
- 18 Sood, R., Zehnder, J.L., Druzin, M.L. and Brown, P.O. (2006) Gene expression patterns in human placenta. *Proc Natl Acad Sci U S A*, **103**, 5478-5483.
- 19 Guo, L., Choufani, S., Ferreira, J., Smith, A., Chitayat, D., Shuman, C., Uxa, R.,Keating, S., Kingdom, J. and Weksberg, R. (2008) Altered gene expression and

methylation of the human chromosome 11 imprinted region in small for gestational age (SGA) placentae. *Dev Biol*, **320**, 79-91.

- 20 Maccani, M.A. and Marsit, C.J. (2009) Epigenetics in the placenta. *Am J Reprod Immunol*, **62**, 78-89.
- Beattie, J.O. (1986) Transplacental alcohol intoxication. *Alcohol Alcohol*, 21, 163-166.
- 22 Bush, P.G., Mayhew, T.M., Abramovich, D.R., Aggett, P.J., Burke, M.D. and Page, K.R. (2000) Maternal cigarette smoking and oxygen diffusion across the placenta. *Placenta*, **21**, 824-833.
- 23 Bailey, D.N. (1987) Cocaine detection during toxicology screening of a university medical center patient population. *J Toxicol Clin Toxicol*, **25**, 71-79.
- 24 Pastrakuljic, A., Derewlany, L.O. and Koren, G. (1999) Maternal cocaine use and cigarette smoking in pregnancy in relation to amino acid transport and fetal growth. *Placenta*, **20**, 499-512.
- 25 Rastogi, S., Nandlike, K. and Fenster, W. (2007) Elevated blood lead levels in pregnant women: identification of a high-risk population and interventions. *J Perinat Med*, **35**, 492-496.
- Mose, T., Knudsen, L.E., Hedegaard, M. and Mortensen, G.K. (2007)
 Transplacental transfer of monomethyl phthalate and mono(2-ethylhexyl)
 phthalate in a human placenta perfusion system. *Int J Toxicol*, 26, 221-229.
- Castles, A., Adams, E.K., Melvin, C.L., Kelsch, C. and Boulton, M.L. (1999)
 Effects of smoking during pregnancy. Five meta-analyses. *Am J Prev Med*, 16, 208-215.

- 28 Shah, N.R. and Bracken, M.B. (2000) A systematic review and meta-analysis of prospective studies on the association between maternal cigarette smoking and preterm delivery. *Am J Obstet Gynecol*, **182**, 465-472.
- Kaddar, T., Rouault, J.P., Chien, W.W., Chebel, A., Gadoux, M., Salles, G.,
 Ffrench, M. and Magaud, J.P. (2009) Two new miR-16 targets: caprin-1 and
 HMGA1, proteins implicated in cell proliferation. *Biol Cell*, **101**, 511-524.
- 30 Einarson, A. and Riordan, S. (2009) Smoking in pregnancy and lactation: a review of risks and cessation strategies. *Eur J Clin Pharmacol*, **65**, 325-330.
- 31 Thielen, A., Klus, H. and Muller, L. (2008) Tobacco smoke: unraveling a controversial subject. *Exp Toxicol Pathol*, **60**, 141-156.
- 32 Lambers, D.S. and Clark, K.E. (1996) The maternal and fetal physiologic effects of nicotine. *Semin Perinatol*, **20**, 115-126.
- 33 Mohsin, M. and Bauman, A.E. (2005) Socio-demographic factors associated with smoking and smoking cessation among 426,344 pregnant women in New South Wales, Australia. *BMC Public Health*, **5**, 138.
- Jaakkola, N., Jaakkola, M.S., Gissler, M. and Jaakkola, J.J. (2001) Smoking during pregnancy in Finland: determinants and trends, 1987-1997. *Am J Public Health*, **91**, 284-286.
- Cnattingius, S. (2004) The epidemiology of smoking during pregnancy: smoking prevalence, maternal characteristics, and pregnancy outcomes. *Nicotine Tob Res*,
 6 Suppl 2, S125-140.

- Goodwin, R.D., Keyes, K. and Simuro, N. (2007) Mental disorders and nicotine dependence among pregnant women in the United States. *Obstet Gynecol*, **109**, 875-883.
- Godfrey, K.M. (2002) The role of the placenta in fetal programming-a review.*Placenta*, 23 Suppl A, S20-27.
- Behrman, R.E., Lees, M.H., Peterson, E.N., De Lannoy, C.W. and Seeds, A.E.
 (1970) Distribution of the circulation in the normal and asphyxiated fetal primate. *Am J Obstet Gynecol*, **108**, 956-969.
- 39 Jansson, T., Wennergren, M. and Illsley, N.P. (1993) Glucose transporter protein expression in human placenta throughout gestation and in intrauterine growth retardation. *J Clin Endocrinol Metab*, **77**, 1554-1562.
- Langley-Evans, S.C., Welham, S.J., Sherman, R.C. and Jackson, A.A. (1996)
 Weanling rats exposed to maternal low-protein diets during discrete periods of gestation exhibit differing severity of hypertension. *Clin Sci (Lond)*, **91**, 607-615.
- 41 Bird, A. (2007) Perceptions of epigenetics. *Nature*, **447**, 396-398.
- 42 Reamon-Buettner, S.M., Mutschler, V. and Borlak, J. (2008) The next innovation cycle in toxicogenomics: environmental epigenetics. *Mutat Res*, **659**, 158-165.
- Monk, M., Boubelik, M. and Lehnert, S. (1987) Temporal and regional changes in
 DNA methylation in the embryonic, extraembryonic and germ cell lineages
 during mouse embryo development. *Development*, 99, 371-382.
- Kafri, T., Ariel, M., Brandeis, M., Shemer, R., Urven, L., McCarrey, J., Cedar, H.
 and Razin, A. (1992) Developmental pattern of gene-specific DNA methylation in
 the mouse embryo and germ line. *Genes Dev*, 6, 705-714.

- 45 Razin, A. and Shemer, R. (1995) DNA methylation in early development. *Hum Mol Genet*, 4 Spec No, 1751-1755.
- Haycock, P.C. and Ramsay, M. (2009) Exposure of Mouse Embryos to Ethanol
 During Preimplantation Development: Effect on DNA Methylation in the H19
 Imprinting Control Region. *Biol Reprod*.
- 47 Reik, W. and Walter, J. (2001) Genomic imprinting: parental influence on the genome. *Nat Rev Genet*, **2**, 21-32.
- 48 Constancia, M., Pickard, B., Kelsey, G. and Reik, W. (1998) Imprinting mechanisms. *Genome Res*, **8**, 881-900.
- 49 Wood, A.J. and Oakey, R.J. (2006) Genomic imprinting in mammals: emerging themes and established theories. *PLoS Genet*, **2**, e147.
- 50 Swales, A.K. and Spears, N. (2005) Genomic imprinting and reproduction. *Reproduction*, **130**, 389-399.
- 51 Tilghman, S.M. (1999) The sins of the fathers and mothers: genomic imprinting in mammalian development. *Cell*, **96**, 185-193.
- Badcock, C. and Crespi, B. (2006) Imbalanced genomic imprinting in brain development: an evolutionary basis for the aetiology of autism. *J Evol Biol*, 19, 1007-1032.
- 53 Isles, A.R. and Holland, A.J. (2005) Imprinted genes and mother-offspring interactions. *Early Hum Dev*, 81, 73-77.
- 54 Tycko, B. and Morison, I.M. (2002) Physiological functions of imprinted genes. J *Cell Physiol*, **192**, 245-258.

- Chang, A.S., Moley, K.H., Wangler, M., Feinberg, A.P. and Debaun, M.R. (2005)
 Association between Beckwith-Wiedemann syndrome and assisted reproductive
 technology: a case series of 19 patients. *Fertil Steril*, 83, 349-354.
- 56 DeBaun, M.R., Niemitz, E.L. and Feinberg, A.P. (2003) Association of in vitro fertilization with Beckwith-Wiedemann syndrome and epigenetic alterations of LIT1 and H19. *Am J Hum Genet*, **72**, 156-160.
- Maher, E.R., Brueton, L.A., Bowdin, S.C., Luharia, A., Cooper, W., Cole, T.R.,
 Macdonald, F., Sampson, J.R., Barratt, C.L., Reik, W. *et al.* (2003) BeckwithWiedemann syndrome and assisted reproduction technology (ART). *J Med Genet*,
 40, 62-64.
- 58 Uejima, H., Lee, M.P., Cui, H. and Feinberg, A.P. (2000) Hot-stop PCR: a simple and general assay for linear quantitation of allele ratios. *Nat Genet*, **25**, 375-376.
- Bjornsson, H.T., Albert, T.J., Ladd-Acosta, C.M., Green, R.D., Rongione, M.A.,
 Middle, C.M., Irizarry, R.A., Broman, K.W. and Feinberg, A.P. (2008) SNP specific array-based allele-specific expression analysis. *Genome Res*, 18, 771-779.
- Kerkel, K., Spadola, A., Yuan, E., Kosek, J., Jiang, L., Hod, E., Li, K., Murty,
 V.V., Schupf, N., Vilain, E. *et al.* (2008) Genomic surveys by methylation sensitive SNP analysis identify sequence-dependent allele-specific DNA
 methylation. *Nat Genet*, 40, 904-908.
- Turner, B.M. (2002) Cellular memory and the histone code. *Cell*, **111**, 285-291.
- 62 Kimura, A.P., Liebhaber, S.A. and Cooke, N.E. (2004) Epigenetic modifications at the human growth hormone locus predict distinct roles for histone acetylation and methylation in placental gene activation. *Mol Endocrinol*, **18**, 1018-1032.

32

- Jenuwein, T. and Allis, C.D. (2001) Translating the histone code. *Science*, 293, 1074-1080.
- Zhang, Y. and Reinberg, D. (2001) Transcription regulation by histone
 methylation: interplay between different covalent modifications of the core
 histone tails. *Genes Dev*, 15, 2343-2360.
- 65 Strahl, B.D. and Allis, C.D. (2000) The language of covalent histone modifications. *Nature*, **403**, 41-45.
- Wang, H., Cao, R., Xia, L., Erdjument-Bromage, H., Borchers, C., Tempst, P. and
 Zhang, Y. (2001) Purification and functional characterization of a histone H3 lysine 4-specific methyltransferase. *Mol Cell*, 8, 1207-1217.
- Nishioka, K., Chuikov, S., Sarma, K., Erdjument-Bromage, H., Allis, C.D.,
 Tempst, P. and Reinberg, D. (2002) Set9, a novel histone H3 methyltransferase
 that facilitates transcription by precluding histone tail modifications required for
 heterochromatin formation. *Genes Dev*, 16, 479-489.
- Rea, S., Eisenhaber, F., O'Carroll, D., Strahl, B.D., Sun, Z.W., Schmid, M.,
 Opravil, S., Mechtler, K., Ponting, C.P., Allis, C.D. *et al.* (2000) Regulation of
 chromatin structure by site-specific histone H3 methyltransferases. *Nature*, 406, 593-599.
- Yang, L., Xia, L., Wu, D.Y., Wang, H., Chansky, H.A., Schubach, W.H.,
 Hickstein, D.D. and Zhang, Y. (2002) Molecular cloning of ESET, a novel
 histone H3-specific methyltransferase that interacts with ERG transcription factor. *Oncogene*, 21, 148-152.

- Schultz, D.C., Ayyanathan, K., Negorev, D., Maul, G.G. and Rauscher, F.J., 3rd
 (2002) SETDB1: a novel KAP-1-associated histone H3, lysine 9-specific
 methyltransferase that contributes to HP1-mediated silencing of euchromatic
 genes by KRAB zinc-finger proteins. *Genes Dev*, 16, 919-932.
- Ogawa, H., Ishiguro, K., Gaubatz, S., Livingston, D.M. and Nakatani, Y. (2002)
 A complex with chromatin modifiers that occupies E2F- and Myc-responsive
 genes in G0 cells. *Science*, **296**, 1132-1136.
- Uhlmann, K., Brinckmann, A., Toliat, M.R., Ritter, H. and Nurnberg, P. (2002)
 Evaluation of a potential epigenetic biomarker by quantitative methyl-single
 nucleotide polymorphism analysis. *Electrophoresis*, 23, 4072-4079.
- Agalioti, T., Lomvardas, S., Parekh, B., Yie, J., Maniatis, T. and Thanos, D.
 (2000) Ordered recruitment of chromatin modifying and general transcription factors to the IFN-beta promoter. *Cell*, **103**, 667-678.
- 74 Cosma, M.P., Tanaka, T. and Nasmyth, K. (1999) Ordered recruitment of transcription and chromatin remodeling factors to a cell cycle- and developmentally regulated promoter. *Cell*, **97**, 299-311.
- Rice, J.C. and Allis, C.D. (2001) Histone methylation versus histone acetylation:new insights into epigenetic regulation. *Curr Opin Cell Biol*, **13**, 263-273.
- Berger, S.L. (2002) Histone modifications in transcriptional regulation. *Curr Opin Genet Dev*, 12, 142-148.
- Lachner, M., O'Sullivan, R.J. and Jenuwein, T. (2003) An epigenetic road map for histone lysine methylation. *J Cell Sci*, **116**, 2117-2124.

- Calvanese, V., Horrillo, A., Hmadcha, A., Suarez-Alvarez, B., Fernandez, A.F.,
 Lara, E., Casado, S., Menendez, P., Bueno, C., Garcia-Castro, J. *et al.* (2008)
 Cancer genes hypermethylated in human embryonic stem cells. *PLoS ONE*, 3, e3294.
- Lee, R.C., Feinbaum, R.L. and Ambros, V. (1993) The C. elegans heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14. *Cell*, 75, 843-854.
- Reinhart, B.J., Slack, F.J., Basson, M., Pasquinelli, A.E., Bettinger, J.C., Rougvie,
 A.E., Horvitz, H.R. and Ruvkun, G. (2000) The 21-nucleotide let-7 RNA
 regulates developmental timing in Caenorhabditis elegans. *Nature*, 403, 901-906.
- Lagos-Quintana, M., Rauhut, R., Lendeckel, W. and Tuschl, T. (2001)
 Identification of novel genes coding for small expressed RNAs. *Science*, 294, 853-858.
- Lau, N.C., Lim, L.P., Weinstein, E.G. and Bartel, D.P. (2001) An abundant class of tiny RNAs with probable regulatory roles in Caenorhabditis elegans. *Science*, 294, 858-862.
- Lee, R.C. and Ambros, V. (2001) An extensive class of small RNAs in
 Caenorhabditis elegans. *Science*, **294**, 862-864.
- Griffiths-Jones, S., Saini, H.K., van Dongen, S. and Enright, A.J. (2008)
 miRBase: tools for microRNA genomics. *Nucleic Acids Res*, 36, D154-158.
- 85 Reinhart, B.J. and Bartel, D.P. (2002) Small RNAs correspond to centromere heterochromatic repeats. *Science*, **297**, 1831.

- Ambros, V., Lee, R.C., Lavanway, A., Williams, P.T. and Jewell, D. (2003)
 MicroRNAs and other tiny endogenous RNAs in C. elegans. *Curr Biol*, **13**, 807-818.
- 87 Aravin, A.A., Hannon, G.J. and Brennecke, J. (2007) The Piwi-piRNA pathway provides an adaptive defense in the transposon arms race. *Science*, **318**, 761-764.
- Bartel, D.P. (2004) MicroRNAs: genomics, biogenesis, mechanism, and function.
 Cell, **116**, 281-297.
- Lee, Y., Jeon, K., Lee, J.T., Kim, S. and Kim, V.N. (2002) MicroRNA
 maturation: stepwise processing and subcellular localization. *EMBO J*, 21, 4663-4670.
- 90 Smalheiser, N.R. (2003) EST analyses predict the existence of a population of chimeric microRNA precursor-mRNA transcripts expressed in normal human and mouse tissues. *Genome Biol*, **4**, 403.
- 91 Cai, X., Hagedorn, C.H. and Cullen, B.R. (2004) Human microRNAs are processed from capped, polyadenylated transcripts that can also function as mRNAs. *RNA*, **10**, 1957-1966.
- Lee, Y., Ahn, C., Han, J., Choi, H., Kim, J., Yim, J., Lee, J., Provost, P.,
 Radmark, O., Kim, S. *et al.* (2003) The nuclear RNase III Drosha initiates
 microRNA processing. *Nature*, 425, 415-419.
- Yi, R., Qin, Y., Macara, I.G. and Cullen, B.R. (2003) Exportin-5 mediates the nuclear export of pre-microRNAs and short hairpin RNAs. *Genes Dev*, 17, 3011-3016.

- 94 Lund, E., Guttinger, S., Calado, A., Dahlberg, J.E. and Kutay, U. (2004) Nuclear export of microRNA precursors. *Science*, **303**, 95-98.
- Hutvagner, G. and Zamore, P.D. (2002) A microRNA in a multiple-turnover
 RNAi enzyme complex. *Science*, 297, 2056-2060.
- Schwarz, D.S., Hutvagner, G., Du, T., Xu, Z., Aronin, N. and Zamore, P.D.
 (2003) Asymmetry in the assembly of the RNAi enzyme complex. *Cell*, 115, 199-208.
- 97 Khvorova, A., Reynolds, A. and Jayasena, S.D. (2003) Functional siRNAs and miRNAs exhibit strand bias. *Cell*, **115**, 209-216.
- Mourelatos, Z., Dostie, J., Paushkin, S., Sharma, A., Charroux, B., Abel, L.,
 Rappsilber, J., Mann, M. and Dreyfuss, G. (2002) miRNPs: a novel class of
 ribonucleoproteins containing numerous microRNAs. *Genes Dev*, 16, 720-728.
- 99 Song, J.J., Smith, S.K., Hannon, G.J. and Joshua-Tor, L. (2004) Crystal structure of Argonaute and its implications for RISC slicer activity. *Science*, **305**, 1434-1437.
- Yekta, S., Shih, I.H. and Bartel, D.P. (2004) MicroRNA-directed cleavage of HOXB8 mRNA. *Science*, **304**, 594-596.
- 101 Filipowicz, W., Bhattacharyya, S.N. and Sonenberg, N. (2008) Mechanisms of post-transcriptional regulation by microRNAs: are the answers in sight? *Nat Rev Genet*, 9, 102-114.
- Lim, L.P., Lau, N.C., Garrett-Engele, P., Grimson, A., Schelter, J.M., Castle, J.,Bartel, D.P., Linsley, P.S. and Johnson, J.M. (2005) Microarray analysis shows

that some microRNAs downregulate large numbers of target mRNAs. *Nature*,433, 769-773.

- 103 van Rooij, E., Sutherland, L.B., Qi, X., Richardson, J.A., Hill, J. and Olson, E.N.
 (2007) Control of stress-dependent cardiac growth and gene expression by a microRNA. *Science*, **316**, 575-579.
- 104 van Rooij, E., Marshall, W.S. and Olson, E.N. (2008) Toward microRNA-based therapeutics for heart disease: the sense in antisense. *Circ Res*, **103**, 919-928.
- 105 Mocellin, S., Pasquali, S. and Pilati, P. (2009) Oncomirs: from tumor biology to molecularly targeted anticancer strategies. *Mini Rev Med Chem*, **9**, 70-80.
- 106 Chan, J.A., Krichevsky, A.M. and Kosik, K.S. (2005) MicroRNA-21 is an antiapoptotic factor in human glioblastoma cells. *Cancer Res*, **65**, 6029-6033.
- Barad, O., Meiri, E., Avniel, A., Aharonov, R., Barzilai, A., Bentwich, I., Einav,
 U., Gilad, S., Hurban, P., Karov, Y. *et al.* (2004) MicroRNA expression detected
 by oligonucleotide microarrays: system establishment and expression profiling in
 human tissues. *Genome Res*, 14, 2486-2494.
- 108 Pineles, B.L., Romero, R., Montenegro, D., Tarca, A.L., Han, Y.M., Kim, Y.M., Draghici, S., Espinoza, J., Kusanovic, J.P., Mittal, P. *et al.* (2007) Distinct subsets of microRNAs are expressed differentially in the human placentas of patients with preeclampsia. *Am J Obstet Gynecol*, **196**, 261 e261-266.
- 109 Zhu, X.M., Han, T., Sargent, I.L., Yin, G.W. and Yao, Y.Q. (2009) Differential expression profile of microRNAs in human placentas from preeclamptic pregnancies vs normal pregnancies. *Am J Obstet Gynecol.*

- 110 Noack, F., Ribbat-Idel, J., Thorns, C., Chiriac, A., Axt-Fliedner, R., Diedrich, K. and Feller, A.C. (2011) miRNA expression profiling in formalin-fixed and paraffin-embedded placental tissue samples from pregnancies with severe preeclampsia. *J Perinat Med*.
- 111 Donker, R.B., Mouillet, J.F., Nelson, D.M. and Sadovsky, Y. (2007) The expression of Argonaute2 and related microRNA biogenesis proteins in normal and hypoxic trophoblasts. *Mol Hum Reprod*, **13**, 273-279.
- Chim, S.S., Shing, T.K., Hung, E.C., Leung, T.Y., Lau, T.K., Chiu, R.W. and Lo,
 Y.M. (2008) Detection and characterization of placental microRNAs in maternal
 plasma. *Clin Chem*, 54, 482-490.
- Gilad, S., Meiri, E., Yogev, Y., Benjamin, S., Lebanony, D., Yerushalmi, N.,
 Benjamin, H., Kushnir, M., Cholakh, H., Melamed, N. *et al.* (2008) Serum
 microRNAs are promising novel biomarkers. *PLoS ONE*, 3, e3148.
- 114 Marsit, C.J., Eddy, K. and Kelsey, K.T. (2006) MicroRNA responses to cellular stress. *Cancer Res*, 66, 10843-10848.
- 115 Callis, T.E. and Wang, D.Z. (2008) Taking microRNAs to heart. *Trends Mol Med*, 14, 254-260.
- Zhang, H.J., Siu, M.K., Wong, E.S., Wong, K.Y., Li, A.S., Chan, K.Y., Ngan,
 H.Y. and Cheung, A.N. (2008) Oct4 is epigenetically regulated by methylation in
 normal placenta and gestational trophoblastic disease. *Placenta*, 29, 549-554.



Figure 1. Basic development of the human embryonic and extraembryonic tissues. Figure adapted from (16).

Chapter Two

Maternal Cigarette Smoking During Pregnancy is Associated with Downregulation

of miR-16, miR-21, and miR-146a in the Placenta

Matthew A. Maccani, Michele Avissar-Whiting, Carolyn E. Banister, Bethany

McGonnigal, James F. Padbury, and Carmen J. Marsit

Epigenetics; October 1, 2010; 5(7): 583-589

Maternal Cigarette Smoking During Pregnancy is Associated with Downregulation of *miR-16*, *miR-21*, and *miR-146a* in the Placenta

Maccani, MA¹, Avissar-Whiting, M¹, Banister, CE¹, McGonnigal, B², Padbury, JF², and Marsit, CJ^{1,3}.

¹Department of Pathology and Laboratory Medicine, Brown University, Providence, RI,

USA

² Department of Pediatrics, Women and Infants' Hospital, Providence, RI, USA

³ Department of Community Health, Center for Environmental Health and Technology Brown University, Providence, RI USA

Corresponding Author: Carmen J. Marsit, Assistant Professor, Department of Pathology and Laboratory Medicine, Brown University, Box G-E537, Providence, RI, 02908, USA; phone; 401-863-6508; fax: 401-863-9008; email: carmen_marsit@brown.edu

Matthew_Maccani@brown.edu; Michele_Avissar@brown.edu; Carolyn_Banister@brown.edu; BMcGonnigal@wihri.org; JPadbury@wihri.org; Carmen_Marsit@brown.edu

CONFLICT OF INTEREST: The authors declare that there are no conflicts of interest.

ABSTRACT

Maternal cigarette smoking during pregnancy is associated with poor fetal outcome and aberrant miRNA expression is associated with adverse pregnancy outcomes. In 25 human placentas, we analyzed the expression of four candidate miRNA previously implicated in growth and developmental processes: *miR-16, miR-21, miR-146a,* and *miR-182,* and used three immortalized placental cell lines to identify if specific components of cigarette smoke were responsible for alterations to miRNA expression. *miR-16, miR-21,* and *miR-146a* were significantly downregulated in cigarette smoke-exposed placentas compared to controls. TCL-1 cells exposed to both nicotine and benzo(a)pyrene exhibited significant, dose-dependent downregulation of *miR-146a.* These results suggest that *miR-146a* is particularly responsive to exposures, and that smoking may elicit some of its downstream effects through alteration of miRNA expression.

KEY WORDS: miRNA, placenta, cigarette smoking, nicotine, benzo(a)pyrene, epigenetics

RUNNING TITLE: Cigarette smoking downregulates miRNA in placenta

INTRODUCTION

Gene-environment interactions occur throughout the lifetime but there are few times when such interactions are more important than during intrauterine development when perturbations can affect fetal growth and development. Further, there can be lasting effects due to altered fetal programming. During in utero development, the placenta plays a critical role supporting normal growth and development, providing the fetus with nutrients, assisting in waste removal, and protecting the fetus from both maternal immune rejection and from other environmental insults. Additionally, the placenta is a metabolic and endocrine center, producing and secreting hormones which support each stage of pregnancy. The placenta also expresses metabolic compounds responsible for the reactions which ultimately protect the fetus from exposure to A number of drugs and toxicants, including nicotine,¹ alcohol,² and toxicants. benzo(a)pyrene,³ have been found to accumulate in placenta tissue and affect placental gene expression. Maternal cigarette smoking has been reported to be associated with increased risk for spontaneous abortion⁴ and preterm delivery.⁴⁻⁶ Previous studies have shown that there are placental complications linked to cigarette smoke exposure during pregnancy, including alterations to the development and function of the placenta.⁷ There are more than 4,000 chemicals in a cigarette, including nicotine, benzo(a)pyrene, and carbon monoxide; more than 43 of these chemicals are known carcinogens.⁸ Nicotine readily crosses the placenta and can result in fetal concentrations that are 15% higher than maternal concentrations.¹ While a number of studies have shown a decrease in overall prevalence of smoking in women in the past 20 years, the prevalence of smoking in young pregnant women has increased.^{9, 10} Additional studies have reported that 12-15%

of all women smoke during their pregnancies.^{11, 12} Taken collectively, these observations suggest that maternal smoking during pregnancy remains an important common exposure that can have major ramifications on not only the normal growth and development of the fetus but also on fetal programming.

The mechanisms by which exposures, such as cigarette smoke, affect the complex regulatory mechanisms of the placenta are still being characterized. One such mode of toxicity may be through the altered expression of microRNA (miRNA), small ~22 nucleotide-long noncoding RNA molecules, which are highly ubiquitous and possess conservation across many species.¹³ MicroRNA posttranscriptionally regulate gene expression by base-pairing to the 3'-untranslated region of a target mRNA resulting in either translational repression or direct degradation of the mRNA, the exact mechanism of which depends largely on the degree of complementarity of the miRNA to its mRNA target. Because partial complementarity of a miRNA to an mRNA target can still lead to translational repression, a single miRNA has the capability of regulating a large number of genes.¹⁴ By negatively regulating their mRNA targets, miRNA have been implicated in regulating cell proliferation, growth and differentiation, and apoptosis.¹⁵ Highly conserved clusters of primate-specific miRNA are expressed in the placenta and other tissues.¹⁶⁻¹⁹ Additionally, multiple groups have further characterized a number of placental miRNA whose aberrant expression is associated with derangements in the intrauterine environment or maternal condition.²⁰⁻²² These discoveries have generated much interest in the involvement of miRNA in placental gene regulation and the possible utility of discovering placental miRNA which can serve as clinical biomarkers of exposure or disease.

As these miRNA play critical roles in development both of the fetus and placenta, and because toxicant exposures and stress can alter the expression of miRNA,^{23, 24} we sought to characterize the modulation of placental miRNA by maternal cigarette smoke, by analyzing the expression of four candidate miRNA for associations with maternal cigarette smoking during pregnancy.

RESULTS

Twenty-five placentas, 8 with a history of maternal cigarette smoking during pregnancy and 17 non-smoking controls, were analyzed for expression of candidate miRNA. Table 1 illustrates the demographics of the sample population. There were no significant differences in the gender, gestational age, birth weight, or maternal age between exposed and unexposed infants. Quantitative RT-PCR analysis revealed downregulation of *miR-16* (p<0.0001), *miR-21* (p<0.0001), and *miR-146a* (p<0.01) associated with maternal cigarette smoking during pregnancy (Figure 1). miR-182 expression showed no differential expression by maternal cigarette smoking during pregnancy. To control for potential confounding effects of clinical variables in these results, we utilized multivariable linear regression to examine the association between *miR-16, miR-21,* and *miR-146a* expression with maternal smoking during pregnancy, controlled for gestational age, birth weight, gender, and maternal age. The respective regression coefficients and p-values are provided in Table 2. These models suggest that maternal smoking leads to a statistically significant 3.9% decrease in relative expression of miR-146a, a 5.8% decrease in the relative expression of miR-16, and a 9.4% decrease in the relative expression of *miR-21*, each independent of infant gender, gestational age, maternal age, and infant birth weight.

Three immortalized placental cell lines representing three different stages and aspects of placental development were chosen to examine the effects of nicotine and benzo(a)pyrene specifically on the expression of the three miRNA found to be significantly altered in the primary tissues in cells from these different stages and aspects of placental development. We sought to investigate the effects of nicotine and benzo(a)pyrene exposure on the expression of these three miRNA in the first trimester villous 3A cells, the first trimester extravillous HTR8 cells, and the third trimester extravillous TCL-1 cells. As shown in Figure 2A, qRT-PCR analysis revealed that *miR-146a* was significantly altered across nicotine exposures in TCL-1 (ANOVA p<0.03), with a specific significant downregulation of approximately 2.5 fold in TCL-1 exposed to 1 μ M nicotine compared to mock control (p<0.02). No differential expression of *miR-16* or *miR-21* occurred in TCL-1, 3A, and HTR8 cells exposed to nicotine (data not shown). Similar to nicotine exposure, qRT-PCR analysis revealed that *miR-146a* was downregulated in TCL-1 cells across all benzo(a)pyrene exposure levels (Fig 2B, ANOVA p<0.006), with 1 μ M exposures leading to a greater than 50% downregulation, and 10 μ M an almost 75% downregulation of the miRNA. Again, no differential expression of *miR-16* or *miR-16* or *miR-16* or *miR-21* occurred in TCL-1, 3A, and HTR8 cells exposed to benzo(a)pyrene (data not shown).

Table 3 shows the targets as predicted by using the three-algorithm miRNA target prediction approach described, and confirmed targets with respective sources^{6, 25-29} for the three candidate miRNA with significant differential expression in primary placenta samples exposed to cigarette smoking *in-utero*. Predicted targets were as follows: for *miR-16*, BCL2L2 and EDA; for *miR-21*, PLAG1 and SATB1; and for *miR-146a*, TRAF6. Minimum free energies (MFE) are listed for each respective miRNA and target mRNA duplex and were determined using *RNA-hybrid*.

DISCUSSION

Taken as a whole, our work identifies the association of cigarette smoking during pregnancy with the aberrant expression of several miRNA involved in critical cell processes. A number of groups have previously demonstrated that nicotine¹ and benzo(a)pyrene³ interact with the placenta and may affect placental growth and development. Previous work demonstrated the effects of toxicants and agents of cell stress on expression of *miR-146a*,^{23, 24, 30, 31} as well as on *miR-16*³² and *miR-21*.^{33, 34} While our observations are limited by incomplete information regarding the duration of cigarette smoking during pregnancy, cigarette per day usage, or more extensive environmental exposure information (such as alcohol usage, environmental pollutant exposure, or secondhand cigarette smoke exposure), our data comprise an important first step in determining associations between maternal cigarette smoking during pregnancy and aberrant miRNA expression in the placenta.

To further investigate the effects of cigarette smoke, we utilized three placental cell lines from different stages of placental development to investigate mechanisms of aberrant miRNA expression associated with cigarette smoke exposure. Downregulation of *miR-146a* in TCL-1 cells treated with nicotine and benzo(a)pyrene suggests that *miR-146a* may be especially sensitive to agents of cellular stress. Moreover, this result suggests that two components of cigarette smoke which affect the expression of *miR-146a* in term placentas may be nicotine and benzo(a)pyrene. The lack of differential expression of *miR-16* and *miR-21* in cells treated with nicotine or benzo(a)pyrene does not necessarily rule out that their expression is not modulated by cigarette smoke. Other components of cigarette smoke may modulate the expression of these miRNA.

Furthermore, the effect of the mixed exposures encountered with the myriad chemicals in cigarette smoke may be truly responsible for the modulation of expression of these miRNA. Experiments conducted with different components of cigarette smoke and complex mixtures of these components may provide more extensive information on the mechanistic effects of cigarette smoke on miRNA expression.

Previous studies on benzo(a)pyrene-exposed human choriocarcinoma cell lines BeWo and JEG-3 showed that placental cell line proliferation was largely unchanged following exposure to 10 and 50 μ M doses of benzo(a)pyrene in serum-containing medium.³⁵ Genbacev and colleagues used nicotine-exposed placental explants to conclude that nicotine is an important molecule that can drastically placental growth and development.³⁶ In our cell culture experiments, we saw no detectable change in cell growth or morphology in any of our cell lines treated with any of the doses of benzo(a)pyrene or nicotine higher than the ones we used may have led to increased cell toxicity and alterations in cell growth due to increased cell stress. Future investigations using higher doses may discover miRNA whose expression and function is altered by higher levels of benzo(a)pyrene, nicotine, and other agents of environmental stress.

One of the more difficult challenges is miRNA target prediction; complementarity between miRNAs and their target mRNAs is generally far from perfect, and is not required for miRNA to functionally silence the expression of a protein product. As a result of this complexity, a simple search for sequence complementarity between the miRNA and its mRNA target, the basis of many target prediction algorithms, can be expected to produce many false-positive hits. Moreover, most miRNAs are thought to have potentially hundreds of targets, some of which will be targeted more strongly than others, and in cell-type specific contexts. Thus, it is essential to devise a strategy which predicts targets based not solely on sequence homology, but which also incorporates additional characterizations when choosing which targets should be examined with further testing.

Our target prediction strategy suggested that PLAG1 and SATB1 were targets of miR-21. Both of these genes are transcription factors implicated in tissue-specific control of the cell cycle and cellular proliferation, and PLAG1 has been previously demonstrated to be a target of miR-21.²⁶ Downregulation of miR-21 due to cigarette smoke exposure might lead to overproduction of PLAG1 and SATB1 protein, resulting in the overexpression of these key genes involved in cell cycling and proliferation.

Our target prediction strategy predicted BCL2L2 and EDA as targets for *miR-16*. BCL2L2, or B-cell CLL/lymphoma 2 like 2 protein, is a pro-survival molecule previously reported to be targeted by *miR-133b*.³⁷ Like other anti-apoptotic proteins in the BCL-2 family, BCL2L2 has been shown to be regulated both pre-transcriptionally and post-transcriptionally.³⁸ The prediction that BCL2L2 is a target of *miR-16* supports hypotheses that by downregulating *miR-16*, cigarette smoke upregulates BCL2L2 and may enhance signaling through anti-apoptotic pathways. The EDA gene is mutated in anhidrotic ectodermal dysplasia and under normal circumstances, produces the protein ectodysplasin, a member of the TNF superfamily that is involved in activation of the NF- $\kappa\beta$ signaling pathway.³⁹ Again, *miR-16* downregulation by cigarette smoke exposure, could lead to an upregulation of EDA protein thereby enhancing NF- $\kappa\beta$ signaling and increased tendency toward survival.

Our *in-silico* analysis predicted TRAF6 as a target for *miR-146a*, suggesting that downregulation of *miR-146a* could result in an upregulation of TRAF6, a protein also important for signaling through both the NF $\kappa\beta$ signaling pathway⁴⁰ as well as a mediator of inflammation in the toll-like receptor 4 (TLR4) pathway.^{41, 42} Previous reports using different target prediction strategies have hypothesized TRAF6 as a target of *miR-146a*²⁸ and have validated TRAF6 as a target in murine macrophages.²⁹ TLR4 mediates the inflammatory response, and aberrant TLR-4 signaling is associated with inflammation-induced preterm delivery.⁴³ Such a consequence could be the result of dysregulated TRAF6 production as a result of downregulation of *miR-146a*. TRAF6 overexpression due to downregulation of *miR-146a* could result in overactive TLR signaling which could have a number of downstream consequences for both placenta and fetus. Further investigation into such relationships would prove vital into better understanding both the role of TRAF6 in TLR4 signaling, as well as the association of aberrant expression of *miR-146a* with adverse pregnancy outcomes.

As with *miR-16*, downregulation of *miR-146a* can lead to enhanced signaling through the anti-apoptotic and pro-survival NF $\kappa\beta$ pathway, resulting in placental cells that avoid apoptosis and prolong their survival. Components of cigarette smoke have been suggested to function through the NF $\kappa\beta$ pathway,⁴⁴ and in light of our results with both placental tissues as well as cell culture experiments, both benzo(a)pyrene and nicotine may be acting through the NF $\kappa\beta$ signaling pathway through downregulation of *miR-146a*. Studies by other groups have demonstrated that an NF $\kappa\beta$ -modulated pathway upregulates *miR-146a* when *miR-146a* expression is induced with IL-1 β .⁴⁵ While this may be true under normal circumstances under which conditions of cellular stress are

held to a minimum, our data suggest that agents of cell stress, such as nicotine and benzo(a)pyrene, may downregulate *miR-146a* expression which, by consequently upregulating TRAF6 expression, enhances signaling through NF $\kappa\beta$ and promotes cell survival. The complexities of this potential duality of effect, namely the notion that *miR-146a* both negatively regulates and can be regulated by the NF $\kappa\beta$ pathway, remain to be further elucidated and future experiments involving knock-downs of various components of the pathway, including the miRNA acting as negative regulators of the pathway, may clarify the mechanisms at play. Our results suggest then that TRAF6, and therefore the NF $\kappa\beta$ and TLR4 signaling pathways, may be important mediators of the effects of cigarette smoke exposure on the placenta.

Our results indicate a potential cascade of molecular changes that occur in the placenta upon exposure to cigarette smoke. We suggest that cigarette smoke exposure during pregnancy is associated with downregulation of *miR-16*, *miR-21*, and *miR-146a* in the placenta. Collectively, aberrant repression of these miRNA upregulates the targets of these miRNA and may affect cell cycle regulation, growth, immunomodulation, and development in the placenta. These changes in target gene expression may have further effects downstream for both placenta and fetus, ultimately resulting in altered fetal programming. Future studies are ongoing to further characterize the effects of environmental exposures on placental miRNA, on understanding the downstream phenotypes of aberrant miRNA expression, and in investigating the placenta as a record of the intrauterine environment.

MATERIALS AND METHODS.

Placenta Samples. Placenta samples were collected within two hours of delivery at Women and Infants' Hospital in Providence, RI. An approximately 1 g biopsy of placenta was excised, free of maternal decidua, from the maternal side of the placenta 2 cm from the umbilical cord insertion site, and the sample was placed immediately in RNAlater and stored at 4°C. At least 72 hours later, placenta samples were removed from the RNAlater, blotted dry, aliquoted, and stored in sample tubes at -80°C until needed for examination. Medical information, including cigarette smoking during pregnancy, available on patient charts was also collected. All samples were collected in under appropriate IRB protocols for Women and Infants' Hospital, and Brown University.

RNA extraction. RNA was extracted from placenta samples and cultured cells using the miRvana miRNA Isolation Kit (Ambion) following the manufacturer protocols. For tissue samples, a 200 mg piece was cut and placed in ice cold, sterile PBS and homogenized using a PowerGen 125 Tissue Homogenizer (Fisher Scientific). The homogenized sample was removed from PBS, and RNA was extracted using manufacturer protocols. Extracted RNA was quantified using a Nanodrop spectrophotometer and then aliquoted into single-use aliquots and stored at -80°C.

Quantitative RT-PCR (qRT-PCR) for mature miRNA. Expression of mature miRNAs was measured using commercially available TaqMan MicroRNA Assays or TaqMan Gene Expression Assays (Applied Biosystems, Valencia, CA) on an Applied Biosystems 7500 Real Time PCR system, and analyzed with 7500 System Software All reactions were run in triplicate, with RNU44 serving as the referent for miRNA expression. In addition, a no-RT control was run with each plate.

Cell culture. 3A and TCL-1 placental cell lines were cultured in RPMI 1640 medium (GIBCO) supplemented with 10% FBS and 1% Pen-Strep. HTR8 cells were cultured in RPMI 1640 medium (GIBCO) supplemented with 5% FBS. These three placental cell lines represent three different stages and aspects of placental development and were selected to examine the effects of components of cigarette smoke in models of different time-points during pregnancy. 3A cells are first-trimester villous cells, HTR-8s are first trimester extravillous cells and TCL-1s are third trimester extravillous cells. Placental cell lines were exposed to $0 \mu M$, $1 \mu M$, $5 \mu M$, and $10 \mu M$ doses of nicotine (Sigma-Aldrich) in medium for 6 days, with 1 μ M being closest to a physiologically relevant dose and higher doses used to investigate the effects of more concentrated doses of nicotine on placental cells.^{46, 47} Placental cell lines were exposed to 0.1 µM, 1 µM, and 10 µM doses of benzo(a)pyrene (Sigma-Aldrich) or DMSO (< 0.1%) alone in medium for 6 days based on previously published methods and physiologically-relevant doses.^{23, 48} None of the doses of nicotine or benzo(a)pyrene led to increased cell toxicity compared to controls. Cells were cultured for 6 days, with exposure medium refreshed on days 2 and 4. The 6day period of exposure was chosen to mimic a long-term chronic exposure, but to avoid sub-culturing of the cells, and has been previously described.²³ All experimental and control conditions were performed in triplicate. Following exposure, cells were harvested and RNA was extracted for analysis by qRT-PCR as described above.

miRNA target prediction. 3 target prediction algorithms were used to predict targets for miRNA of interest, and a fourth algorithm was used to evaluate predicted targets for base-pairing and minimum free energy. The three algorithms used for target prediction were *miRanda* (September 2008 release, available online at

http://www.microrna.org/microrna/home.do), *PicTar* (as cited by ⁴⁹ and available online at <u>http://pictar.mdc-berlin.de/cgi-bin/new_PicTar_vertebrate.cgi</u>), and *TargetScan 5.1* (available online at <u>http://www.targetscan.org/</u>). In order to be considered a predicted target for further investigation, the target must have appeared in the top 100 targets in all three prediction algorithms. The *RNA-hybrid* algorithm was used to evaluate predicted targets for information on secondary structure and thermodynamic stability of the miRNA-mRNA duplex.⁵⁰ *RNA-hybrid* is available online at <u>http://bibiserv.techfak.unibielefeld.de/rnahybrid/submission.html</u>.

Statistical analysis. Two-tailed *t* tests were used to compare miRNA expression levels determined by real-time PCR in human tissue samples. In order to examine the association between exposure to cigarette smoke and miRNA expression, controlling for potential confounders, we employed multivariable linear regression modeling, in SAS 9.1 (SAS Institute, Cary, NC). ANOVA was used to determine differential expression of miRNA across treatments of nicotine and benzo(a)pyrene on cells in cell culture experiments.
FUNDING

This work was funded by NIH grants from the NCRR-P20RR018728, the NIEHS P42ES013660 and T32ES007272 (MAM), and the Flight Attendants Medical Research Institute Young Clinical Scientist Award.

ACKNOWLEDGEMENTS

Many thanks to Devin Koestler, Charlotte Wilhelm, Amanda Filiberto, and Luc Gagne for their insightful comments and criticisms during the preparation of this manuscript and to Keila Veiga and Alyse LaLiberte for placenta sample and medical history collection.

REFERENCES

- Lambers DS, Clark KE. The maternal and fetal physiologic effects of nicotine. Semin Perinatol 1996; 20:115-26.
- 2. Beattie JO. Transplacental alcohol intoxication. Alcohol Alcohol 1986; 21:163-6.
- Guyda HJ, Mathieu L, Lai W, Manchester D, Wang SL, Ogilvie S, et al. Benzo(a)pyrene inhibits epidermal growth factor binding and receptor autophosphorylation in human placental cell cultures. Mol Pharmacol 1990; 37:137-43.
- Castles A, Adams EK, Melvin CL, Kelsch C, Boulton ML. Effects of smoking during pregnancy. Five meta-analyses. Am J Prev Med 1999; 16:208-15.
- Shah NR, Bracken MB. A systematic review and meta-analysis of prospective studies on the association between maternal cigarette smoking and preterm delivery. Am J Obstet Gynecol 2000; 182:465-72.
- Kaddar T, Rouault JP, Chien WW, Chebel A, Gadoux M, Salles G, et al. Two new miR-16 targets: caprin-1 and HMGA1, proteins implicated in cell proliferation. Biol Cell 2009; 101:511-24.
- Einarson A, Riordan S. Smoking in pregnancy and lactation: a review of risks and cessation strategies. Eur J Clin Pharmacol 2009; 65:325-30.
- Thielen A, Klus H, Muller L. Tobacco smoke: unraveling a controversial subject. Exp Toxicol Pathol 2008; 60:141-56.
- Mohsin M, Bauman AE. Socio-demographic factors associated with smoking and smoking cessation among 426,344 pregnant women in New South Wales, Australia. BMC Public Health 2005; 5:138.

- Jaakkola N, Jaakkola MS, Gissler M, Jaakkola JJ. Smoking during pregnancy in Finland: determinants and trends, 1987-1997. Am J Public Health 2001; 91:284-6.
- Cnattingius S. The epidemiology of smoking during pregnancy: smoking prevalence, maternal characteristics, and pregnancy outcomes. Nicotine Tob Res 2004; 6 Suppl 2:S125-40.
- Goodwin RD, Keyes K, Simuro N. Mental disorders and nicotine dependence among pregnant women in the United States. Obstet Gynecol 2007; 109:875-83.
- Lee RC, Feinbaum RL, Ambros V. The C. elegans heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14. Cell 1993; 75:843-54.
- Du T, Zamore PD. Beginning to understand microRNA function. Cell Res 2007; 17:661-3.
- Miska EA. How microRNAs control cell division, differentiation and death. Curr Opin Genet Dev 2005; 15:563-8.
- Bentwich I, Avniel A, Karov Y, Aharonov R, Gilad S, Barad O, et al.
 Identification of hundreds of conserved and nonconserved human microRNAs.
 Nat Genet 2005; 37:766-70.
- Lehnert S, Van Loo P, Thilakarathne PJ, Marynen P, Verbeke G, Schuit FC.
 Evidence for co-evolution between human microRNAs and Alu-repeats. PLoS
 One 2009; 4:e4456.
- Zhang R, Wang YQ, Su B. Molecular evolution of a primate-specific microRNA family. Mol Biol Evol 2008; 25:1493-502.
- 19. Tsai KW, Kao HW, Chen HC, Chen SJ, Lin WC. Epigenetic control of the

expression of a primate-specific microRNA cluster in human cancer cells. Epigenetics 2009; 4:587-92.

- 20. Hutvagner G, Zamore PD. A microRNA in a multiple-turnover RNAi enzyme complex. Science 2002; 297:2056-60.
- Hu HY, Yan Z, Xu Y, Hu H, Menzel C, Zhou YH, et al. Sequence features associated with microRNA strand selection in humans and flies. BMC Genomics 2009; 10:413.
- 22. Pineles BL, Romero R, Montenegro D, Tarca AL, Han YM, Kim YM, et al.
 Distinct subsets of microRNAs are expressed differentially in the human placentas of patients with preeclampsia. Am J Obstet Gynecol 2007; 196:261 e1-6.
- Marsit CJ, Eddy K, Kelsey KT. MicroRNA responses to cellular stress. Cancer Res 2006; 66:10843-8.
- Avissar-Whiting M, Veiga K, Uhl K, Maccani M, Gagne L, Moen E, et al.
 Bisphenol A exposure leads to specific microRNA alterations in placental cells.
 Reprod Toxicol 2010.
- Meng F, Henson R, Wehbe-Janek H, Ghoshal K, Jacob ST, Patel T. MicroRNA-21 regulates expression of the PTEN tumor suppressor gene in human hepatocellular cancer. Gastroenterology 2007; 133:647-58.
- Tran N, McLean T, Zhang X, Zhao CJ, Thomson JM, O'Brien C, et al. MicroRNA expression profiles in head and neck cancer cell lines. Biochem Biophys Res Commun 2007; 358:12-7.
- 27. Rom S, Rom I, Passiatore G, Pacifici M, Radhakrishnan S, Del Valle L, et al.

CCL8/MCP-2 is a target for mir-146a in HIV-1-infected human microglial cells. FASEB J 2010.

- 28. Taganov KD, Boldin MP, Chang KJ, Baltimore D. NF-kappaB-dependent induction of microRNA miR-146, an inhibitor targeted to signaling proteins of innate immune responses. Proc Natl Acad Sci U S A 2006; 103:12481-6.
- 29. Hou J, Wang P, Lin L, Liu X, Ma F, An H, et al. MicroRNA-146a feedback inhibits RIG-I-dependent Type I IFN production in macrophages by targeting TRAF6, IRAK1, and IRAK2. J Immunol 2009; 183:2150-8.
- Curtale G, Citarella F, Carissimi C, Goldoni M, Carucci N, Fulci V, et al. An emerging player in the adaptive immune response: microRNA-146a is a modulator of IL-2 expression and activation-induced cell death in T lymphocytes. Blood 2010; 115:265-73.
- 31. Wang JF, Yu ML, Yu G, Bian JJ, Deng XM, Wan XJ, et al. Serum miR-146a and miR-223 as potential new biomarkers for sepsis. Biochem Biophys Res Commun 2010.
- 32. Pothof J, Verkaik NS, van IW, Wiemer EA, Ta VT, van der Horst GT, et al. MicroRNA-mediated gene silencing modulates the UV-induced DNA-damage response. EMBO J 2009; 28:2090-9.
- 33. Weber M, Baker MB, Moore JP, Searles CD. MiR-21 is induced in endothelial cells by shear stress and modulates apoptosis and eNOS activity. Biochem Biophys Res Commun 2010.
- 34. Wang P, Zou F, Zhang X, Li H, Dulak A, Tomko RJ, Jr., et al. microRNA-21 negatively regulates Cdc25A and cell cycle progression in colon cancer cells.

Cancer Res 2009; 69:8157-65.

- 35. Zhang L, Connor EE, Chegini N, Shiverick KT. Modulation by benzo[a]pyrene of epidermal growth factor receptors, cell proliferation, and secretion of human chorionic gonadotropin in human placental cell lines. Biochem Pharmacol 1995; 50:1171-80.
- Genbacev O, McMaster MT, Lazic J, Nedeljkovic S, Cvetkovic M, Joslin R, et al.
 Concordant in situ and in vitro data show that maternal cigarette smoking
 negatively regulates placental cytotrophoblast passage through the cell cycle.
 Reprod Toxicol 2000; 14:495-506.
- Crawford M, Batte K, Yu L, Wu X, Nuovo GJ, Marsh CB, et al. MicroRNA 133B targets pro-survival molecules MCL-1 and BCL2L2 in lung cancer. Biochem Biophys Res Commun 2009; 388:483-9.
- 38. Fabian D, Cikos S, Koppel J. Gene expression in mouse preimplantation embryos affected by apoptotic inductor actinomycin D. J Reprod Dev 2009; 55:576-82.
- Esibizione D, Cui CY, Schlessinger D. Candidate EDA targets revealed by expression profiling of primary keratinocytes from Tabby mutant mice. Gene 2008; 427:42-6.
- 40. Qian Y, Commane M, Ninomiya-Tsuji J, Matsumoto K, Li X. IRAK-mediated translocation of TRAF6 and TAB2 in the interleukin-1-induced activation of NFkappa B. J Biol Chem 2001; 276:41661-7.
- 41. Wang Q, Dziarski R, Kirschning CJ, Muzio M, Gupta D. Micrococci and peptidoglycan activate TLR2-->MyD88-->IRAK-->TRAF-->NIK-->IKK-->NFkappaB signal transduction pathway that induces transcription of interleukin-8.

Infect Immun 2001; 69:2270-6.

- 42. Vabulas RM, Ahmad-Nejad P, da Costa C, Miethke T, Kirschning CJ, Hacker H, et al. Endocytosed HSP60s use toll-like receptor 2 (TLR2) and TLR4 to activate the toll/interleukin-1 receptor signaling pathway in innate immune cells. J Biol Chem 2001; 276:31332-9.
- 43. Li L, Kang J, Lei W. Role of Toll-like receptor 4 in inflammation-induced preterm delivery. Mol Hum Reprod 2010; 16:267-72.
- 44. Nakao S, Ogata Y, Sugiya H. Nicotine stimulates the expression of cyclooxygenase-2 mRNA via NFkappaB activation in human gingival fibroblasts. Arch Oral Biol 2009; 54:251-7.
- 45. Perry MM, Williams AE, Tsitsiou E, Larner-Svensson HM, Lindsay MA.
 Divergent intracellular pathways regulate interleukin-1beta-induced miR-146a and miR-146b expression and chemokine release in human alveolar epithelial cells. FEBS Lett 2009; 583:3349-55.
- Russell MA, Jarvis M, Iyer R, Feyerabend C. Relation of nicotine yield of cigarettes to blood nicotine concentrations in smokers. Br Med J 1980; 280:972-6.
- Waggoner SE, Wang X. Effect of nicotine on proliferation of normal, malignant, and human papillomavirus-transformed human cervical cells. Gynecol Oncol 1994; 55:91-5.
- 48. Drukteinis JS, Medrano T, Ablordeppey EA, Kitzman JM, Shiverick KT. Benzo[a]pyrene, but not 2,3,7,8-TCDD, induces G2/M cell cycle arrest, p21CIP1 and p53 phosphorylation in human choriocarcinoma JEG-3 cells: a distinct signaling pathway. Placenta 2005; 26 Suppl A:S87-95.

- 49. Lall S, Grun D, Krek A, Chen K, Wang YL, Dewey CN, et al. A genome-wide map of conserved microRNA targets in C. elegans. Curr Biol 2006; 16:460-71.
- 50. Kruger J, Rehmsmeier M. RNAhybrid: microRNA target prediction easy, fast and flexible. Nucleic Acids Res 2006; 34:W451-4.

Table 1. Demographics of the study population.			
	Total	Maternal Smoking During Pregnancy	No Maternal Smoking During Pregnancy
Total, n (%)	25 (100)	8 (32)	17 (68)
Gender, n (%)			
Female	13 (52)	3 (37.5)	10 (59)
Male	12 (48)	5 (62.5)	7 (41)
Gestational Age in weeks, mean (SD)	38.7 (0.88)	38.4 (1.20)	38.9 (0.66)
Birthweight in g, mean (SD)	3080 (540)	3091 (682)	3071 (484)
Maternal Age in years, mean (SD)	27.3 (6.66)	24.3 (7.94)	28.8 (5.68)



Figure 1. Maternal cigarette smoking during pregnancy is associated with downregulation of *miR-16*, *miR-21*, and *miR-146a*. Quantitative RT-PCR analysis was used to examine the expression of the mature forms of *miR-16* (A), *miR-21* (B), *miR-146a* (C), and *miR-182* (D), in primary placenta tissue samples from infants whose mothers smoked cigarettes (CS) during pregnancy (n=8) and infants whose mothers did not smoke during pregnancy (No CS, n=17). * indicates p<0.01 and ** indicates p<0.0001 as determined by t-test.

Table 2. Multivariable linear regression model of individual miRNA expression.						
	miR-16		miR-21		mi R-146 a	
	Reg. coeff.	р	Reg. coeff.	р	Reg. coeff.	Р
Maternal Cigarette Smoking During Pregnancy						
No	Reference		Reference		Reference	
Yes	-0.058	< 0.0001	-0.094	< 0.0001	-0.039	0.003
Gender						
Female	Reference		Reference		Reference	
Male	-0.008	0.4	-0.003	0.9	0.001	0.9
Gestational Age, per week	0.004	0.5	0.007	0.5	0.005	0.5
Birthweight, per gram	-8.85x10 ⁻ 06	0.4	-1.26 x10 ⁻ 06	0.9	-7.23 x10 ⁻	0.5
Maternal Age, per year	-0.0003	0.7	-0.002	0.1	-0.001	0.2
Note: Each miRNA is modeled individually, and the models are controlled for all co-						
variates in the table.						



Figure 2. *miR-146a* is downregulated in TCL-1 cells exposed to nicotine and benzo(a)pyrene. Three placental cell lines (HTR8, 3A, and TCL1) were exposed to increasing doses of nicotine (A) and benzo(A)pyrene (B) for 6 days, and the expression of *miR-146A* was determined through qRT-PCR. Error bars indicated standard error of the mean, * indicates a significant downregulation of *miR-146a* across doses of nicotine (ANOVA, p<0.03). ** indicates a significant downregulation of *miR-146a* across doses of benzo(a)pyrene (ANOVA, p<0.006).

Table 3. mRNA targets predicted and/or experimentally validated for miR-16, miR-21, and miR-146a.

miRNA	Predicted target mRNA	Minimum free energy of duplex (kcal / mol)	Experimentally confirmed target mRNA	Source of target confirmation
miR-16	BCL2L2 EDA	-23.1 -25.3	Caprin-1, HGMA-1	Kaddar <i>et al</i> . 2009
miR-21	PLAG1 SATB1	-22.7 -23.8	PTEN PLAG1	Meng <i>et al.</i> 2007 Tran <i>et al.</i> 2007
miR- 146a	TRAF6	-26.1	CCL8/MCP-2 TRAF6	Rom et al. 2010 Taganov, <i>et al.</i> 2006; Hou <i>et al.</i> 2009

Note: In order to be considered a "predicted target", the target must appear in the "top 100" targets for the respective miRNA using all of the following target prediction tools: miRanda (September 2008 Release), PicTar (based on Lall, et al., 2006), and TargetScan 5.1. Minimum free energies of the miRNA-mRNA duplexes were determined using *RNA-hybrid*.

Chapter Three

miR-16 and miR-21 Expression in the Placenta is Associated with Fetal Growth

Matthew A. Maccani, James F. Padbury, Carmen J. Marsit

Under Review

miR-16 and miR-21 Expression in the Placenta is Associated with Fetal Growth

Matthew A. Maccani¹, James F. Padbury², Carmen J. Marsit^{1,3}

¹Department of Pathology and Laboratory Medicine, Brown University, Providence, RI,

USA; ²Department of Pediatrics, Women and Infants Hospital, Providence, RI, USA:

³Center for Environmental Health and Technology, Department of Community Health, Brown University, Providence, RI, USA

Corresponding Author Carmen J. Marsit, Box G-E55 Providence, RI 02912 Telephone: (401) 863-6508 Fax: (401) 863-9008

Email: <u>Carmen_Marsit@brown.edu</u>

Matthew_Maccani@brown.edu; JPadbury@wihri.org; Carmen_Marsit@brown.edu The authors have declared that no conflict of interest exists.

ABSTRACT

Novel research has suggested that altered miRNA expression in the placenta is associated with adverse pregnancy outcomes and with potentially harmful xenobiotic exposures. We hypothesized that aberrant expression of miRNA in the placenta is associated with fetal growth, a measurable phenotype resulting from a number of intrauterine factors, and one which is significantly predictive of later life outcomes. We analyzed 107 primary, term, human placentas for expression of 6 miRNA reported to be expressed in the placenta and to regulate cell growth and development pathways: miR-16, miR-21, miR-93, miR-135b, miR-146a, and miR-182. The expression of miR-16 and miR-21 was markedly reduced in infants with the lowest birthweights (p < 0.05). Logistic regression models suggested that low expression of *miR-16* in the placenta predicts an over 4-fold increased odds of small for gestational age (SGA) status (p=0.009). Moreover, having both low *miR-16* and low *miR-21* expression in the placenta predicts a greater increase in odds for SGA than having just low *miR-16* or *miR-21* expression (p<0.02), suggesting an additive effect of both of these miRNA. Future research on miRNA whose expression is associated with *in utero* exposures and markers of fetal growth is essential for better understanding the epigenetic mechanisms underlying the developmental origins of health and disease.

INTRODUCTION

Fetal development represents a critical period during which perturbations to the intrauterine environment through various factors in the extrauterine environment can have major ramifications on not only the proper growth and development of the fetus but also on risk for disease later in life(1). Barker and Hales(1) hypothesized that fetuses receive a poor or rich maternal forecast depending, in part, on the intrauterine and extrauterine conditions during pregnancy, but that this maternal forecast may not always accurately predict the post-birth environment and that such mismatches give rise to disease risk later in life. This has played out in a number of epidemiologic studies linking low birthweight with morbidity and mortality in early infancy(2) as well as with an increased risk for certain diseases later in life, particularly coronary heart disease, diabetes mellitus type 2, and hypercholesterolemia(2).

The placenta is of critical importance to ensure the proper growth and development of the fetus while *in utero*. It is involved in providing the fetus with nutrients and is involved in waste and gas exchange. The placenta's metabolic activity is crucial for protecting the fetus from potentially harmful maternal factors and xenobiotic toxicants that may alter fetal growth and development. The environment during pregnancy is thought to impact the appropriate function of the placenta during development, thus identification of alterations to the placenta and to placental gene expression may serve as a record of *in utero* exposures and of the intrauterine and extrauterine environments during pregnancy(3).

The mechanisms by which in utero exposures may dysregulate the regulatory mechanisms of the placenta continue to be studied. One mode of alteration may be

through the aberrant expression of microRNA (miRNA), 21-25 nucloeotide long noncoding RNA involved in post-transcriptional gene regulation (4, 5). miRNA base-pair to the 3'-untranslated region of target mRNA and effectively silence gene expression by a mechanism of either translational repression or direct mRNA degradation. The particular mechanism of this post-transcriptional regulation depends greatly on the degree of complementarity of the miRNA to its mRNA target. Previous work has shown that partial complementarity of a miRNA to an mRNA target may result in effective repression of translation; therefore, a single miRNA can regulate a vast number of genes(5). Through this mechanism of post-transcriptional gene regulation, miRNA have been shown to regulate a number of key cellular functions including migration, invasion, growth, and death(6). miRNA exhibit tissue-specific expression and function and have been shown to be expressed in the placenta in addition to a variety of other tissues(7). Alterations to placental miRNA expression have been associated with *in utero* exposures(8, 9) and adverse pregnancy outcomes (10-13).

Since miRNA have been described as playing important roles in development and are susceptible to the environment, we sought to further characterize the expression of six candidate miRNA previously shown to be expressed in the placenta and involved in regulating key cell processes – miR-16(14), miR-21(15), miR-93(12, 13), miR-135b(11), miR-146a(16), and miR-182(17) – in a large series of human placentas for associations with fetal growth.

RESULTS

One hundred seven human placenta samples were analyzed for the expression of candidate miRNA previously shown to be expressed in the placenta and involved in regulating cell growth and developmental processes, specifically, miR-16, miR-21, miR-93, miR-135b, miR-146a, and miR-182. The demographics of the sample population are given in **Table 1.** Of note, the study population was oversampled for small for gestational age (SGA) infants, defined as infants whose birth weight was <10th percentile for their gestational age, calculated as in Fenton 2003(18). Placentas from SGA neonates comprised approximately 30% of the study population. Table 2 describes the expression of the 6 candidate miRNA in all of the 107 samples, based on qRT-PCR and absolute quantification from a standard curve. The range of expression values differed for the miRNAs, with *miR-146a* and *miR-182* exhibiting expression in the 0.001-3.95 amol range, while *miR-16* and miR-21 expression was 2 orders of magnitude greater with expression ranging from 3.53-434.74 amol range. Within each miRNA, the majority of the samples exhibited relatively homogenous low levels of expression, but the distributions of expression exhibited a right skew, suggesting there were still a number of samples with relatively moderate to high expression. As these data were not normally distributed, and as we hypothesized that aberrant high or low expression may be associated with exposures or outcomes, the expression profiles were split into quartiles, and birthweight percentile was plotted by quartile of miRNA expression (Figure 1). Kruskal-Wallis tests were used to assess differences in birthweight percentile across quartiles of miRNA expression (Figure 1). Analysis revealed that birthweight percentile

significantly differed across quartiles of miR-16 and miR-21 expression, p=0.04 and p=0.02, respectively.

Observing that expression in the lowest quartiles of *miR-16* and *miR-21* was associated with reduced birthweight percentile, we more specifically examined the association between low expression (\leq median vs. >median) of the miRNA and infants considered small for gestational age (SGA), using logistic regression to control for potential confounders. These models (**Table 3**) demonstrate that low *miR-16* expression in the placenta predicts an odds of 4.13 for SGA (95% CI=1.42, 12.05) compared to infants with high placenta *miR-16* expression, controlled for confounders. Additionally, low *miR-21* expression in the placenta predicts a marginally-significant, 2.43-fold increased odds for SGA (95% CI=0.93, 6.37) compared to infants with high placenta *miR-21* expression controlled for confounders. Maternal cigarette smoking during pregnancy, as expected, also demonstrated greatly increased risks for growth restriction in both models.

As both miRNA demonstrated independent correlations with SGA status, we further examined if there was an interaction between expression of these 2 miRNA in their association with infant growth outcome. This model (**Table 4**) demonstrated that compared to infants having high expression (>median) of both miRNA, infants with low *miR-21* only or low *miR-16* only had non-significant elevation in SGA risk, but infants exhibiting reduced expression of both *miR-16* and *miR-21* were significantly more likely to be classified as SGA (OR 5.38, 95% CI 1.52, 19.01). Although a likelihood ratio test suggested no significant multiplicative interaction between *miR-16* and *miR-21* (p>0.05), there was a significant trend for increased risk of being classified as SGA from having

only *miR-21* reduced or only *miR-16* reduced in expression to having both reduced in expression (p<0.02).

DISCUSSION

We have demonstrated the expression of key candidate miRNA in a large, population-based series of primary human placenta samples and their association with poor fetal growth, specifically identifying that reduced expression of *miR-16* and *miR-21* are strongly associated with growth restriction. A number of groups have previously described placental miRNA expression associations with maternal conditions such as preeclampsia(17), with maternal cigarette smoking during pregnancy(8), and as markers of pregnancy itself(11). Furthermore, more is being uncovered about the role of *miR-16* and *miR-21* in regulating key cellular processes, especially the involvement of *miR-16* in regulating cell cycle progression(14) and *miR-21*'s capability of regulating cell cycling and cell proliferation(15).

miR-21 has been described as an oncogene, plays a role in enhancing tumor phenotypes including proliferation and migration, and has been shown to target a number of key regulators of these processes, including but not limited to PLAG1(22) and PTEN(19, 20). Meng and coworkers showed that *miR-21* regulates PTEN in human hepatocellular cancer(20), and Lou and colleagues demonstrated that in ovarian epithelial carcinomas, *miR-21* promotes proliferation, invasion and migration abilities by inhibiting PTEN(19). As the function of the placenta, though, is to promote fetal growth through its own proliferation and invasion into the maternal decidua, downregulation of *miR-21* in the placenta could, through dysregulation of PTEN, result in decreased invasion of the maternal decidua, decreased migration, and decreased growth – the opposite of what has been observed to occur in the case of upregulated *miR-21*(20).

As with many miRNA, miR-16 exhibits tissue-specific function and expression. In a number of cancer cell lines, miR-16 has been shown to be involved in the induction of apoptosis by targeting BCL-2(23) and in cell cycle regulation by targeting CDK6(24), CDC27(25), and CARD10(26). In other cell types, miR-16 has different functions, such as targeting HMGA1 and Caprin-1(21), further suggesting that miR-16 may have celltype function and expression(21, 23). Dysregulation of miR-16 in the placenta may lead to aberrant expression of its targets and may lead to functional and developmental abnormalities in the placenta that might result in reduced infant birthweight. Mechanistic research using model systems is needed to further elucidate the pathways regulated by miR-16 and miR-21 and to better determine the functional consequence of downregulation of miR-16 and miR-21 in the placenta. Additionally, more work to determine the functional effects of miRNA crosstalk – that is, miRNAs whose differential expression may have additive associations with risk for phenotype or disease – will be necessary to better understand the complex regulatory networks involved.

Poor fetal growth associated with adverse intrauterine conditions continues to be characterized. Fetal malnutrition linked to growth restriction has been shown occur in a variety of conditions, including but not limited to poverty, pregnancy in women with eating disorders, and pregnancy in high altitude(27). Maternal cigarette smoking during pregnancy is associated with an increased risk of fetal growth restriction(28, 29), and exposure to environmental toxicants *in utero*, such as those found in cigarette smoke, is associated with increased placental aberrations(30) and decreased placental function (31). A number of animal models have been generated to further study low birthweight in a controlled, experimental system (27, 32), and it will be important to consider the role of

miRNA in these animal models. Because poor maternal forecasts leading to low birthweight may not accurately predict the post-birth environment and because low birthweight increases one's risk for a number of diseases later in life(2), more research is necessary to more fully understand the pathways whose dysregulation may ultimately affect birthweight.

In summary, our data suggesting that low expression of *miR-16* and *miR-21* in the placenta is associated with poor fetal growth may have many important implications. Future work to determine the roles of miRNA in specific pathways leading to altered fetal growth will be key to better understanding fetal growth as both a marker of the intrauterine environment as well as a developmental outcome and in better comprehending the developmental origins of health and disease.

MATERIALS AND METHODS.

Placenta samples. All placenta samples used were collected as part of the Rhode Island Child Health Study (RICHS), an ongoing, population-based birth cohort at Women and Infants' Hospital in Providence, RI. In brief, for each sample, 12 biopsies of placenta tissue, 3 from each of 4 quadrants (totaling approximately 1 g of tissue) were excised, from the maternal side of the placenta 2 cm from the umbilical cord insertion site, free of maternal decidua. The samples were placed immediately in RNAlater and stored at 4°C. At least 72 hours later, placenta samples were removed from RNAlater, blotted dry, snapfrozen in liquid nitrogen, homogenized using a mortar and pestle, and stored in sample tubes at -80°C until needed for examination. A structured chart review was used to collect information from the maternal inpatient medical record from delivery, and mothers were subjected to an interviewer-administered structured questionnaire to obtain information on the lifestyle, demographics, and exposure histories of the participants. All samples and information were collected under appropriate protocols approved by the Institutional Review Boards for Women and Infants' Hospital and Brown University. *RNA extraction.* RNA was extracted from placenta samples and cultured cells using the miRvana miRNA Isolation Kit (Ambion) and manufacturer protocols as described previously(8). For tissue samples, 200 mg of homogenized tissue was used for extraction. Extracted RNA was quantified using a Nanodrop spectrophotometer and then aliquoted into single-use aliquots and stored at -80°C.

Quantitative RT-PCR (qRT-PCR) for mature miRNA. Expression of mature miRNAs was measured using commercially available TaqMan microRNA Assays or TaqMan Gene Expression Assays (Applied Biosystems, Valencia, CA) on an Applied Biosystems

7900HT Real-Time PCR system and analyzed with 7900HT System Software. Absolute quantitation of miRNA was calculated using a standard curve generated from serial dilutions of miRNA-specific pre-miR oligonucleotides (Ambion) run on each plate for each miRNA of interest, as previously described for array analyses by Bissels and colleagues (33). All reactions were run in triplicate on 384-well plates, and RNU-44 was used as an internal control for each sample to assess sample performance. No-RT controls were also run for each sample on each plate to assure samples were free of genomic DNA.

Statistical analysis. Birthweight percentile was calculated using the Fenton growth chart(18). SGA was defined as the lowest 10% of birthweight percentile. Kruskal-Wallis tests were used to determine if birthweight percentile significantly differed across quartiles of miRNA expression. Logistic regression models were used to analyze if differential expression of miRNA predicted increased risk for SGA while also considering potential confounders. All analyses were conducted in SAS 9.2 (SAS Institute, Cary, NC).

ACKNOWLEDGMENTS

Many thanks to Michele Avissar-Whiting, Charlotte Wilhelm-Benartzi, Devin Koestler, and Luc Gagne for their helpful comments and criticisms during the preparation of this manuscript and to Carolyn Banister, Gilda Ferro, Joyce Lee, Keila Veiga, Andrew Rainey, and Alison Migliori for placenta sample, medical history, and questionnaire data collection and processing.

FUNDING

This work was funded by National Institutes of Health (NIH) grants from the NIEHS

[P42ES013660] and [T32ES007272 to M.A.M.] and the NCRR [P20 RR018728].

CONFLICT OF INTEREST STATEMENT

The authors have no conflict of interest to declare.

REFERENCES

- 1 Hales, C.N. and Barker, D.J. (1992) Type 2 (non-insulin-dependent) diabetes mellitus: the thrifty phenotype hypothesis. *Diabetologia*, 35, 595-601.
- 2 Argente, J., Mehls, O. and Barrios, V. (2010) Growth and body composition in very young SGA children. *Pediatr Nephrol*, 25, 679-685.
- 3 Maccani, M.A. and Marsit, C.J. (2009) Epigenetics in the placenta. *Am J Reprod Immunol*, 62, 78-89.
- Lee, R.C., Feinbaum, R.L. and Ambros, V. (1993) The C. elegans heterochronic gene lin-4 encodes small RNAs with antisense complementarity to lin-14. *Cell*, 75, 843-854.
- 5 Du, T. and Zamore, P.D. (2007) Beginning to understand microRNA function. *Cell Res*, 17, 661-663.
- 6 Miska, E.A. (2005) How microRNAs control cell division, differentiation and death. *Curr Opin Genet Dev*, 15, 563-568.
- Bentwich, I., Avniel, A., Karov, Y., Aharonov, R., Gilad, S., Barad, O., Barzilai,
 A., Einat, P., Einav, U., Meiri, E. *et al.* (2005) Identification of hundreds of
 conserved and nonconserved human microRNAs. *Nat Genet*, 37, 766-770.
- Maccani, M.A., Avissar-Whiting, M., Banister, C.E., McGonnigal, B., Padbury,
 J.F. and Marsit, C.J. (2010) Maternal cigarette smoking during pregnancy is
 associated with downregulation of miR-16, miR-21 and miR-146a in the placenta.
 Epigenetics, 5.
- Avissar-Whiting, M., Veiga, K.R., Uhl, K.M., Maccani, M.A., Gagne, L.A.,
 Moen, E.L. and Marsit, C.J. (2010) Bisphenol A exposure leads to specific

microRNA alterations in placental cells. Reprod Toxicol, 29, 401-406.

- 10 Montenegro, D., Romero, R., Pineles, B.L., Tarca, A.L., Kim, Y.M., Draghici, S., Kusanovic, J.P., Kim, J.S., Erez, O., Mazaki-Tovi, S. *et al.* (2007) Differential expression of microRNAs with progression of gestation and inflammation in the human chorioamniotic membranes. *Am J Obstet Gynecol*, 197, 289 e281-286.
- Chim, S.S., Shing, T.K., Hung, E.C., Leung, T.Y., Lau, T.K., Chiu, R.W. and Lo,
 Y.M. (2008) Detection and characterization of placental microRNAs in maternal
 plasma. *Clin Chem*, 54, 482-490.
- Mouillet, J.F., Chu, T., Hubel, C.A., Nelson, D.M., Parks, W.T. and Sadovsky, Y.
 (2010) The levels of hypoxia-regulated microRNAs in plasma of pregnant women with fetal growth restriction. *Placenta*, 31, 781-784.
- Mouillet, J.F., Chu, T., Nelson, D.M., Mishima, T. and Sadovsky, Y. (2010) MiR-205 silences MED1 in hypoxic primary human trophoblasts. *FASEB J*, 24, 2030-2039.
- Liu, Q., Fu, H., Sun, F., Zhang, H., Tie, Y., Zhu, J., Xing, R., Sun, Z. and Zheng,
 X. (2008) miR-16 family induces cell cycle arrest by regulating multiple cell
 cycle genes. *Nucleic Acids Res*, 36, 5391-5404.
- Wang, P., Zou, F., Zhang, X., Li, H., Dulak, A., Tomko, R.J., Jr., Lazo, J.S.,
 Wang, Z., Zhang, L. and Yu, J. (2009) microRNA-21 negatively regulates
 Cdc25A and cell cycle progression in colon cancer cells. *Cancer Res*, 69, 8157-8165.
- 16 Hou, J., Wang, P., Lin, L., Liu, X., Ma, F., An, H., Wang, Z. and Cao, X. (2009)

MicroRNA-146a feedback inhibits RIG-I-dependent Type I IFN production in macrophages by targeting TRAF6, IRAK1, and IRAK2. *J Immunol*, 183, 2150-2158.

- 17 Pineles, B.L., Romero, R., Montenegro, D., Tarca, A.L., Han, Y.M., Kim, Y.M., Draghici, S., Espinoza, J., Kusanovic, J.P., Mittal, P. *et al.* (2007) Distinct subsets of microRNAs are expressed differentially in the human placentas of patients with preeclampsia. *Am J Obstet Gynecol*, 196, 261 e261-266.
- 18 Fenton, T.R. (2003) A new growth chart for preterm babies: Babson and Benda's chart updated with recent data and a new format. *BMC Pediatr*, 3, 13.
- 19 Lou, Y., Yang, X., Wang, F., Cui, Z. and Huang, Y. (2010) MicroRNA-21 promotes the cell proliferation, invasion and migration abilities in ovarian epithelial carcinomas through inhibiting the expression of PTEN protein. *Int J Mol Med*, 26, 819-827.
- Meng, F., Henson, R., Wehbe-Janek, H., Ghoshal, K., Jacob, S.T. and Patel, T.
 (2007) MicroRNA-21 regulates expression of the PTEN tumor suppressor gene in human hepatocellular cancer. *Gastroenterology*, 133, 647-658.
- Kaddar, T., Rouault, J.P., Chien, W.W., Chebel, A., Gadoux, M., Salles, G.,
 Ffrench, M. and Magaud, J.P. (2009) Two new miR-16 targets: caprin-1 and
 HMGA1, proteins implicated in cell proliferation. *Biol Cell*, 101, 511-524.
- Tran, N., McLean, T., Zhang, X., Zhao, C.J., Thomson, J.M., O'Brien, C. and
 Rose, B. (2007) MicroRNA expression profiles in head and neck cancer cell lines.
 Biochem Biophys Res Commun, 358, 12-17.
- 23 Cimmino, A., Calin, G.A., Fabbri, M., Iorio, M.V., Ferracin, M., Shimizu, M.,

Wojcik, S.E., Aqeilan, R.I., Zupo, S., Dono, M. *et al.* (2005) miR-15 and miR-16 induce apoptosis by targeting BCL2. *Proc Natl Acad Sci U S A*, 102, 13944-13949.

- 24 Linsley, P.S., Schelter, J., Burchard, J., Kibukawa, M., Martin, M.M., Bartz, S.R., Johnson, J.M., Cummins, J.M., Raymond, C.K., Dai, H. *et al.* (2007) Transcripts targeted by the microRNA-16 family cooperatively regulate cell cycle progression. *Mol Cell Biol*, 27, 2240-2252.
- Calin, G.A., Cimmino, A., Fabbri, M., Ferracin, M., Wojcik, S.E., Shimizu, M.,
 Taccioli, C., Zanesi, N., Garzon, R., Aqeilan, R.I. *et al.* (2008) MiR-15a and miR-16-1 cluster functions in human leukemia. *Proc Natl Acad Sci U S A*, 105, 5166-5171.
- Chen, R.W., Bemis, L.T., Amato, C.M., Myint, H., Tran, H., Birks, D.K.,
 Eckhardt, S.G. and Robinson, W.A. (2008) Truncation in CCND1 mRNA alters
 miR-16-1 regulation in mantle cell lymphoma. *Blood*, 112, 822-829.
- 27 Tamashiro, K.L. and Moran, T.H. (2010) Perinatal environment and its influences on metabolic programming of offspring. *Physiol Behav*, 100, 560-566.
- Shea, A.K. and Steiner, M. (2008) Cigarette smoking during pregnancy. *Nicotine Tob Res*, 10, 267-278.
- 29 Esposito, E.R., Horn, K.H., Greene, R.M. and Pisano, M.M. (2008) An animal model of cigarette smoke-induced in utero growth retardation. *Toxicology*, 246, 193-202.
- 30 Abel, E.L. (1984) Smoking and pregnancy. *J Psychoactive Drugs*, 16, 327-338.

- 31 Shiverick, K.T. and Salafia, C. (1999) Cigarette smoking and pregnancy I: ovarian, uterine and placental effects. *Placenta*, 20, 265-272.
- 32 Desai, M., Gayle, D., Han, G. and Ross, M.G. (2007) Programmed hyperphagia due to reduced anorexigenic mechanisms in intrauterine growth-restricted offspring. *Reprod Sci*, 14, 329-337.
- Bissels, U., Wild, S., Tomiuk, S., Holste, A., Hafner, M., Tuschl, T. and Bosio, A.
 (2009) Absolute quantification of microRNAs by using a universal reference. *RNA*, 15, 2375-2384.

Birthweight, grams, mean (sd)	3308 (742)
Gestational Age, weeks, mean (sd)	38.9 (1.2)
Small for Gestational Age (SGA) Status, n (%)	
SGA	32 (29.9%)
Non-SGA	75 (70.1%)
Infant Gender, n (%)	
Female	60 (56.1%)
Male	47 (43.9%)
Mode of delivery, n (%)	
Caesarian section	39 (36.4%)
Vaginal delivery	67 (62.6%)
Maternal Age, years, mean (sd)	29.4 (6.0)
Maternal Ethnicity, n (%)	
Non-white	24 (22.4%)
White	80 (74.8%)
Unknown	3 (2.8%)

Table 1. Demographics of the study population (n=107)

* One sample was missing mode of delivery information.

Table 2. Expression of *miR-16*, *miR-21*, *miR-93*, *miR-135b*, *miR-146a*, and *miR-182*determined through qRT-PCR in 107 primary human term placenta samples.

	Median (amol)	Range (amol)
miR-16	18.64	3.53-399.50
miR-21	54.86	5.25-434.74
miR-93	4.26	0.14-118.39
miR-135b	4.72	0.13-216.92
miR-146a	0.1	0.002-3.95
miR-182	0.23	0.001-3.63


Figure 1. Distribution of infant birthweights (y-axis) by primary term human placenta miRNA expression quartiles (x-axis) determined through qRT-PCR for (A) *miR-16* (p=0.04), (B) *miR-21* (p=0.02), (C) *miR-93* (p=0.88), (D) *miR-135b* (p=0.84), (E) *miR-146a* (p=0.46), and (F) *miR-182* (p=0.55). Black bars indicate median of birthweight percentile within each quartile. Kruskal-Wallis tests revealed that birthweight percentile significantly differed across quartiles of *miR-16* and *miR-21* expression (p<0.05).

Table 3. Logistic regression for the association between individual miRNAexpression and SGA status.

95% Wald

Effect	Odds Ratio	Confidence Limits	р
miR-16 Expression in Placenta, n (%)			
High n=52 (51%)	Reference		
Low n=50 (49%)	4.13	1.42-12.05	0.009
Maternal Smoking During Pregnancy, n (%)			
No n=97 (95%)	Reference		
Yes n=5 (5%)	22.18	1.72-286.86	0.018
		95% Wald	
Effect	Odds Ratio	Confidence Limits	р
miR-21 Expression in Placenta, n (%)			
High n=52 (51%)	Reference		
Low n=50 (49%)	2.43	0.93-6.37	0.069
Maternal Smoking During Pregnancy, n (%)			
No n=97 (95%)	Reference		
$V_{00} = -5(50/)$			
1 es II=5 (5%)	10.82	0.82-143.58	0.071

*Also included in models: Relative Weight Gained During Pregnancy, Maternal Ethnicity, Maternal Age, Delivery Method, Insurance, and Infant Gender

Table 4. Logistic regression to examine the interaction of low *miR-16* and low *miR-21* expression on the association with SGA status

Effect	Odds Ratio	95% Wald Confidence Limits
<i>miR-21</i> and <i>miR-16</i> , n (%)		
Both High, n=36 (35%)	Reference	
Low <i>miR-21</i> only, n=16 (16%)	1.54	0.29-8.21
Low <i>miR-16</i> only, n=16 (16%)	3.35	0.69-16.32
Both Low, n=34 (33%)	5.38	1.52-19.01

*Also included in model: Relative Weight Gained During Pregnancy, Maternal Ethnicity, Maternal Age, Delivery Method, Insurance, and Infant Gender

**p for trend (p<0.02).

Chapter Four

Functional Consequences of Overexpression of miRNA in Placental Cell Lines

Matthew A. Maccani and Carmen J. Marsit

ABSTRACT

Key cell processes have been shown to be regulated, in part, by miRNA. Previous work in our lab has suggested that *in utero* exposures and poor fetal growth are associated with the differential expression of several miRNA in the placenta. Using HTR8, 3A, and TCL-1 placental cell lines, representing early and later stages of placental development, we sought to characterize effects of dysregulation of *miR-16*, *miR-21*, and miR-146a – three miRNA previously shown by our lab and others to be expressed in the placenta and to be involved in the regulation of key cell processes – on placental cell viability, growth, proliferation, invasion, and migration. Additionally, we investigated effects of dysregulated miRNA expression on experimentally confirmed targets, specifically PTEN for *miR-21* and TRAF6 and STAT-1 for *miR-146a*. Our data suggested that 3A cells overexpressing *miR-21* exhibited a trend to increased migration compared to negative control cells (p < 0.08). TCL-1 cells overexpressing *miR-21* had 50% less PTEN than cells transfected with negative control (p<0.05). 3A cells overexpressing *miR-146a* had 26% less STAT-1 than cells transfected with negative control (p<0.05). Taken collectively our data suggest that miRNA play important roles in regulating cell processes crucial for the proper function of placental cells during various stages of pregnancy.

INTRODUCTION

The placenta is of critical importance to ensure the proper growth and development of the fetus while *in utero*. It is involved in providing the fetus with nutrients and is involved in waste and gas exchange. The placenta's metabolic activity is crucial for protecting the fetus from potentially harmful toxicants that may alter fetal growth and development. The placenta expresses compounds capable of metabolizing toxicants that would otherwise harm the fetus or lead to fetal demise. In order to perform this multitude of important functions, the placenta needs to differentiate, invade maternal decidua, illicit vascularization and immune changes, while still growing and performing the task of maintaining proper fetal growth and development. Alterations to the placenta and to placental gene expression may alter these important tasks required for appropriate function, and may therefore illicit downstream effects on the fetus (1).

The mechanisms by which *in utero* exposures may dysregulate the regulatory mechanisms of the placenta continue to be studied. miRNA base-pair to the 3'untranslated region of target mRNA and effectively silence gene expression by a mechanism of either translational repression or direct mRNA degradation. The particular mechanism of this post-transcriptional regulation depends greatly on the degree of complementarity of the miRNA to its mRNA target. Previous work has shown that partial complementarity of a miRNA to an mRNA target may result in effective repression of translation; therefore, a single miRNA can regulate a vast number of genes(2). Through their mechanism of post-transcriptional gene regulation, miRNA have been shown to regulate a number of key cell pathways including migration, invasion, growth, and proliferation(3). miRNA exhibit tissue-specific expression and function and

have been shown to be expressed in the placenta in addition to a variety of other tissues(4).

Work in our lab and elsewhere has investigated the association of aberrant placental miRNA expression with *in utero* exposures(5, 6) and adverse intrauterine and extrauterine environment(7-9). Work in our lab has demonstrated that downregulation of *miR-16, miR-21,* and *miR-146a* in the placenta is associated with maternal cigarette smoking during pregnancy(5), that low *miR-16* and *miR-21* expression is associated with fetal growth (Maccani, et al., *submitted*), and that bisphenol A (BPA) exposure dysregulates miRNA expression in placental cell lines(6). Pineles and colleagues investigated miRNA expression associated with preeclampsia and suggested that differential expression of *miR-210* and *miR-182* was associated with preeclampsia versus control patients(7). Others have shown dysregulated miRNA expression in trophoblasts exposed to varying concentrations of oxygen(9-11). This research suggests the importance of miRNA in regulating processes crucial for proper function of the placenta and of placental cells, thereby ensuring the proper health and development of the fetus *in utero*.

As miRNA can regulate key cell functions and as these functions are critical to the proper growth, development and function of the placenta, we hypothesized that overexpression of *miR-16*, *miR-21*, and *miR-146a* in placental cell lines may alter the growth, viability, proliferation, migration, and invasion of these cells. Western blots were used to detect the differential expression of previously predicted and empirically confirmed targets for miRNA, specifically PTEN(12, 13) for *miR-21* and STAT-1(14) and TRAF6(15) for *miR-146a*. Taken collectively, our work attempts to further elucidate

some of the pathways that are regulated by miRNA and whose perturbation in the placenta could have major ramifications for both placental function and proper health and development of the fetus.

RESULTS

Overexpression of miRNA and assessment of transfection efficiency

Effects of overexpression of *miR-16*, *miR-21*, and *miR-146a* on placental cell viability, growth, proliferation, invasion, and migration were investigated using a variety of *in vitro* strategies. HTR8, 3A, and TCL-1 placental cell lines were cultured as described previously(5). These three placental cell lines represent three different stages and aspects of placental development and were selected to examine the effects of dysregulated miRNA expression on models of different timepoints during pregnancy. In brief, 3A cells are first-trimester villous cells, HTR8 are first trimester extravillous cells, and TCL-1s are third trimester extravillous cells. Placental cells were transfected using the pre-miR miRNA precursors (Ambion) system and siPORT NeoFX transfection agent (Ambion) following manufacturer's protocols. The Silencer siRNA Transfection Kit (Ambion) and KDalert GAPDH Assay (Ambion) were used to determine transfection efficiency. **Figure 1** shows that an approximately 45% knockdown of GAPDH enzyme activity was observed in HTR8, 3A, and TCL-1 placental cells transfected with an siRNA to GAPDH, suggesting that a transfection efficiency of approximately 45% exists for these three cell lines using this system of transfection.

Growth

To determine the effects of overexpression of miRNA on cell growth, cell growth was characterized using cell counts. Cells were enumerated at time 0, 24, and 48 hours post-transfection using a Coulter counter. Differential growth of cells overexpressing *miR-16, miR-21,* or *miR-146a* was determined by comparing cell counts to those of negative control cells. **Figure 2** presents results from cell growth analyses using

transfected cells and cell counts to investigate whether overexpression of miRNA results in differential cell growth. The data suggest that overexpression of *miR-16*, *miR-21*, and *miR-146a* in HTR8, 3A, and TCL-1 does not alter cell growth compared to negative control.

Viability

To ensure that cell death was not being increased as a result of overexpression of *miR-16, miR-21*, or *miR-146a*, cell viability tests were conducted. At 0 hours, 24 hours, and 48 hours post-transfection, cell viability was assessed using the trypan blue exclusion test of cell viability. Cells were deemed viable if they successfully excluded trypan blue. **Figure 3** displays results from cell viability assays. The results suggest that overexpression of *miR-16, miR-21*, and *miR-146a* in HTR8, 3A, and TCL-1 does not alter cell viability compared to negative control.

Proliferation

To confirm the cell growth data by quantifying metabolically active cells, cell proliferation was assessed using the MTS assay at 0 hours, 24 hours, and 48 hours post-transfection. MTS is bioreduced by cells into a formazan product that is soluble in tissue culture medium. The conversion of MTS into the aqueous soluble formazan product is accomplished by dehydrogenase enzymes found in metabolically active cells. The quantity of formazan product as measured by the amount of 490 nm absorbance is directly proportional to the number of metabolically active cells in culture. Differential proliferation of cells was determined by comparing number of proliferating cells overexpressing *miR-16*, *miR-21*, or *miR-146a* versus number of proliferating cells transfected with negative control. **Figure 4** shows results from proliferation assays

conducted on HTR8, 3A, and TCL-1 overexpressing *miR-16*, *miR-21*, and *miR-146a* or negative control cells. The data suggest that overexpression of *miR-16*, *miR-21*, and *miR-146a* does not alter the proliferation of HTR8, 3A, and TCL-1 cells.

Migration

Placental cell migration is a key process ensuring proper placental development and function. Migration assays using a modified protocol from Shaw, et al, 1996(16) were conducted to determine whether overexpression of *miR-16*, *miR-21*, and *miR-146a* in placental cells leads to aberrant cell migration. Migration was assessed using an MTS assay as a more conservative, first-pass assay and the technique of staining with hematoxylin and counting described previously(17) was used as a confirming assay that is more sensitive to smaller changes in migration. **Figure 5** shows data from migration assays conducted with 3A, HTR8, and TCL-1 cells overexpressing *miR-16*, *miR-21*, or *miR-146a* or negative control cells. 3A cells overexpressing *miR-21* exhibited a trend of increased invasion, 250% versus control (p<0.08). Overexpression of the other miRNA in other cell types showed no differences in migration compared to controls.

Supplemental Figure 1 shows data from migration assays using the technique of staining with hematoxylin and cell counting described previously(17) to confirm results trending significance (p<0.2) as determined using MTS to quantify migrated cells.

Supplemental Figure 1 shows that 3A cells overexpressing *miR-21* exhibited a trend of increased invasion, 150% versus control (p<0.08), confirming previous results shown in **Figure 5** using MTS to quantify migrated cells.

Invasion

Placental cell invasion of the maternal decidua is a crucial process in not only anchoring the placenta but also ensuring the proper growth and development of the placenta. Invasion assays using a modified protocol from Shaw, et al, 1996(16) were utilized to investigate if overexpression of *miR-16*, *miR-21*, or *miR-146a* leads to altered placental cell invasion. Invasion was assessed using an MTS assay as a more conservative, first-pass assay and the technique of staining with hematoxylin and counting described previously(17) was used as a confirming assay that is more sensitive to smaller changes in invasion. **Figure 6** shows data from invasion assays conducted with 3A, HTR8, and TCL-1 cells overexpressing *miR-16*, *miR-21*, or *miR-146a* or negative control cells. Analysis revealed there was no significant difference in invasion in cells overexpressing any of the three miRNA versus negative control cells.

Empirical validation of miRNA gene targets

Table 1 displays an informative, though not exhaustive, list of predicted and empirically validated gene targets of *miR-16*, *miR-21*, and *miR-146a* reported in the literature. Predicted targets were predicted using the target prediction strategy reported previously(5). Targets on this list were considered for empirical validation in placental cells overexpressing *miR-21* and *miR-146a*. PTEN was selected as a target for validation due to its important role as a regulator of the PI3K pathway, a pathway important for cell survival, migration, and Akt signaling. STAT-1 was selected as a target for validation due to its role as a transcription factor and its involvement in the JAK/STAT signaling pathway. TRAF6 was selected for further validation in placental cells because it was predicted as a target using a previously-reported target strategy(5) and has been previously identified as an important molecule responsible for signaling through the TLR4 pathway. All of the selected targets have been previously empirically validated as targets in non-placental cell types (**Table 1**), and as suggested, are involved in regulating potentially crucial cell processes in the placenta.

Protein from HTR8, 3A, and TCL-1 placental cells overexpressing *miR-21* or *miR-146a*, or transfected with negative control, was harvested and Western blots were used to investigate whether bioinformatically-predicted and/or previously empirically-validated targets of *miR-21* and *miR-146a* are in fact targeted in placental cells. **Figure 7A** shows that TCL-1 cells overexpressing *miR-21* express approximately 50% less PTEN protein than TCL-1 cells transfected with negative control (p<0.05). **Figure 7B** shows that 3A cells overexpressing *miR-146a* express approximately 26% less STAT-1 protein than 3A cells transfected with negative control (p<0.05). Overexpression of *miR-146a* in HTR8, 3A, and TCL-1 did not significantly alter levels of TRAF6 compared to cells transfected with negative control (**Figure 7C**).

DISCUSSION

Our work sought to investigate the effects of miRNA overexpression on a number of key placental cell processes, specifically cell viability, growth, proliferation, invasion, and migration, as well as further confirm previously empirically validated targets of miRNA in placental cells. Previous findings have suggested that these miRNA are significantly altered in human placentas associated with a number of conditions and exposures(5). Taken collectively, our work contributes to the better understanding of how dysregulated *miR-16*, *miR-21*, and *miR-146a* may alter key cellular processes and expression of target genes.

As previous studies have suggested, one of the more challenging aspects of miRNA research is target prediction and validation(5). By using *in silico* bioinformatic approaches combined with empirical techniques, targets of specific miRNA can be predicted and validated. A number of target prediction strategies utilizing one or more of a number of published and publicly-available target prediction algorithms have predicted targets for *miR-16*(5, 18), *miR-21*(5, 12, 19), and *miR-146a*(5). Several of these predicted targets have also been empirically confirmed through experimental techniques demonstrating that the specific miRNA is capable of downregulating or, in some cases, silencing, its target (18, 19).

As with many miRNA, *miR-16* exhibits tissue-specific function and expression. In a number of cancer cell lines, *miR-16* has been shown to be involved in the induction of apoptosis by targeting *BCL-2*(20) and in cell cycle regulation by targeting CDK6(21), CDC27(22), and CARD10(23). In other cell types, *miR-16* has different functions, such as targeting HMGA1 and Caprin-1(18), further suggesting that *miR-16* may have celltype-specific function and expression (18, 20). Dysregulation of *miR-16* in the placenta may lead to aberrant expression of its targets and may lead to dysregulated placental cell processes. While our data suggest that overexpression of *miR-16* does not lead to significant differences in cell viability, growth, proliferation, invasion, or migration, future studies addressing potential effects on apoptosis and target gene expression may prove to be especially useful in better determining the biological ramifications of dysregulated *mir-16* expression in the placenta, which may ultimately lead to abnormalities in the placenta that might result in reduced infant birthweight.

miR-21 has also been shown to target a number of key genes, including but not limited to and PTEN(12, 13). Meng and coworkers showed that *miR-21* regulates PTEN in human hepatocellular cancer(12), and Lou and colleagues demonstrated that in ovarian epithelial carcinomas, *miR-21* promotes proliferation, invasion and migration abilities by inhibiting PTEN(13). Our results that *miR-21* targets PTEN in the TCL-1 placental cell line further suggest that *miR-21* may be one mechanism regulating PTEN expression in the third trimester placenta. Lack of targeting of PTEN by *miR-21* in 3A and HTR8 cells suggests that *miR-21* may not be the key or major regulator of PTEN in first trimester villous and extravillous placental cells, but more work is necessary to further investigate the epigenetic regulation of PTEN in these cell types. Downregulation of *miR-21* in the placenta could thus lead to a dysregulation of PTEN in the placenta and could result in decreased invasion of the maternal decidua, decreased migration, and decreased growth the opposite of what has been observed to occur in the case of upregulated miR-21(12). Our results suggesting that overexpression of *miR-21* in 3A cells leads to a trend towards increased migration compared to cells transfected with negative control may further

underscore the involvement of *miR-21* in regulating placental cell migration in first trimester villous cells. Lack of differential migration in HTR8 and TCL-1 cells overexpressing *miR-21* may suggest that *miR-21* may not be a major modulator of migration in these cell types. Future investigations are necessary to better determine mechanisms controlling migration throughout the various stages of pregnancy and are essential to generating a more complete understanding of *miR-21*'s regulation of placental cell processes.

Our *in-silico* analysis predicted TRAF6 as a target for *miR-146a*, suggesting that downregulation of *miR-146a* could result in an upregulation of TRAF6, a protein also important for signaling through both the NF $\kappa\beta$ signaling pathway(24) as well as a mediator of inflammation in the toll-like receptor 4 (TLR4) pathway (25, 26). Previous reports using different target prediction strategies have hypothesized TRAF6 as a target of miR-146a(5, 27) and have validated TRAF6 as a target in murine macrophages(15). TLR4 mediates the inflammatory response, and aberrant TLR-4 signaling is associated with inflammation-induced preterm delivery(28). Such a consequence could be the result of dysregulated TRAF6 production as a result of downregulation of miR-146a. TRAF6 overexpression due to downregulation of *miR-146a* could result in overactive TLR signaling which could have a number of downstream consequences for both placenta and fetus. Unfortunately, our Western blot analysis suggested that TRAF6 protein levels were not significantly different in placental cells overexpressing miR-146a versus cells transfected with negative control. This may be due, in part, to a more complex regulation of TRAF6 in these placental cell lines than is currently understood. While *miR-146a* may be capable of regulating TRAF6, it does not appear to be the major or most important regulator of TRAF6 in these placental cell lines. Further investigation into such relationships and regulatory networks would prove vital into better understanding both the role of TRAF6 in TLR4 signaling, as well as the association of aberrant expression of *miR-146a* with adverse pregnancy outcomes.

Previous work has demonstrated that STAT-1 is targeted by miR-146a(14). Our Western blot analysis revealed that 3A cells overexpressing miR-146a had 26% less STAT-1 protein than did cells transfected with negative control, a potentially important result since STAT-1 plays a role as a key transcription factor. STAT-1 is activated upon interferon (IFN) binding to its receptor. Once phosphorylated by JAKs, STAT-1 forms a homodimer and translocates to the nucleus where it binds the Interferon-Gamma activation site (GAS) in the promoter of interferon stimulated genes (ISG)(29). Our data suggest that STAT-1 may be regulated, in part, by miR-146a in 3A placental cells. 3A cells are first trimester villous cells, and lack of STAT-1 regulation by miR-146a in HTR8 or TCL-1 cells may suggest that time in pregnancy (i.e. first trimester vs. third trimester) as well as cell type (i.e. villous vs. extravillous) may play a role in determining targeting by miR-146a. More work is necessary to better understand the factors underlying this differential targeting of STAT-1 by miR-146a. Dysregulation of miR-146a may thereby lead to aberrant production of STAT-1 and dysregulated transcription of ISG, potentially leading to a cascade of events which may negatively alter placental function and ultimately lead to altered fetal programming.

While *in vitro* studies are important for proof-of-principle aims, such as targeting of particular genes by miRNA or analyzing the potential effects of dysregulated miRNA expression on one particular cell type, much of the complexity of an *in vivo* system is lost

when using an *in vitro* cell culture model alone. The use of an *ex vivo* model, such as harvesting primary, term, human placental tissue for experiments where one overexpresses a miRNA of interest and then characterizes effects on cell growth, viability, or migration, may prove worthwhile in better demonstrating the effects of dysregulated miRNA expression on primary tissue. Some challenges in such a system could be the difficulty in transfecting cells harvested from primary, term, human placental tissue, as well as the challenge of maintaining a healthy culture for the time period needed to perform the suite of assays aimed at characterizing effects of dysregulated miRNA expression on cell function. Additionally, the possibility of characterizing the effects of aberrant miRNA expression on cells from pre-term placental tissue, especially cells derived from human placental tissue from the first trimester, would be nearly impossible due to the practical difficulty of obtaining placental tissue from first trimester human placenta.

Model systems may prove useful in determining the effects of dysregulated miRNA on both the placenta as well as the developing fetus. Animal models have been generated to further study low birthweight (30, 31), and it will be important to consider the role of dysregulated miRNA expression in these animal models. Mouse lines generated to have tissue-specific (such as placenta-specific) dysregulation of miRNA may be especially useful in determining the effects of differential miRNA expression on both the placenta as well as the fetus at various stages of pregnancy. Placenta-specific knockouts of miRNA might be generated using a combined approach of the Cre/LoxP system and integrase-defective lentiviral (IDLV) vectors to help reduce the otherwise observed toxic effects of Cre expression in miRNA-specific floxed mice; while this system has not been applied to miRNA to date, it has been described previously (32). Of course, such a system presents a number of challenges, including difficulties in developing a miRNA-specific floxed mouse line not to mention the possibility that a total knock-out of a miRNA in the placenta might result in lethality. Nonetheless, future work in such systems will be necessary to further demonstrate and understand the effects of aberrant miRNA expression on placental function and fetal programming.

Collectively, dysregulation of these miRNA can lead to aberrant expression of their targets and may affect a number of key placental cell processes, ultimately leading to aberrant cell cycle regulation, growth, immunomodulation, and development in the placenta. These changes in target gene expression may have further effects downstream for both placenta and fetus, ultimately resulting in altered fetal programming. Future studies are ongoing to further characterize the effects of environmental exposures on placental miRNA, on understanding the downstream phenotypes of aberrant miRNA expression, and in investigating the placenta as a record of the intrauterine environment.

MATERIALS AND METHODS

Cell culture and transfection. HTR8, 3A, and TCL-1 placental cell lines were cultured as described previously(5). Placental cells were transfected using the pre-miR miRNA precursors (Ambion) system and siPORT NeoFX transfection agenet (Ambion) following manufacturer's protocols. The *Silencer* siRNA Transfection Kit (Ambion) and KDalert GAPDH Assay (Ambion) were used to determine transfection efficiency.

Growth assays. Effects on cell growth were characterized using cell counts enumerated by a Coulter Counter. Briefly, cells were transfected as above and plated in triplicate, approximately 250,000 cells per well in 6-well dishes. Cells were enumerated at time 0, 24, and 48 hours post-transfection using a Coulter counter. Differential growth of cells overexpressing *miR-16*, *miR-21*, or *miR-146a* was determined by comparing cell counts to those of negative control cells.

Viability assays. Cells were transfected as above. At the noted timepoints, cells were trypsinized, spun down, and washed 2x with PBS before being resuspended in PBS. 15 μ L of cells resuspended in PBS were mixed with 15 μ L of Trypan blue (Sigma-Aldrich) and incubated at room temperature for 3 minutes prior to counting using a hemacytometer (Hausser). Cells were deemed viable if they successfully excluded Trypan blue (hence, the Trypan Blue Exclusion Test of Cell Viability).

Proliferation assays. Cells were transfected as above and seeded into 96 well plates in triplicate. Twenty-four hours following transfection, transfection medium was removed and replaced with 100 μ L fresh medium and 20 μ L MTS reagent (Promega). Following incubation at 37 degrees C for 3 hours, absorbance at 490 nm was observed using a

spectrophotometer and differential proliferation of cells was determined by comparing number of proliferating cells overexpressing *miR-16*, *miR-21*, or *miR-146a* versus number of proliferating cells transfected with negative control.

Migration assays. Migration assays were conducted using a modified protocol based on Shaw et al, 1996(16). In brief, placental cells were transfected and serum-starved as above. Following serum-starvation, 200,000 cells in 100 μ L serum-free medium were seeded on filter-inserts in Boyden chambers (Corning). As above, complete medium with serum was placed in the lower chamber below the filter-insert and was used as a chemoattractant. After incubation at 37 degrees C for 20 hours, cells that had migrated to the lower side of the filter-insert were quantified using MTS assay (Promega) and manufacturer protocols. MTS assay was utilized as a more conservative, first-pass assay for quantifying cells that had migrated and the technique of staining with hematoxylin and cell counting described previously(17) that is more sensitive to smaller changes in migration was used to confirm results trending significance (p<0.2) as determined by MTS.

Invasion assays. Placental cells were transfected as above. Invasion assays were conducted using a modified protocol by Shaw, et al, 1996(16). Twenty-four hours after transfection with either miRNA-specific pre-miR precursor or negative control, media containing transfection reagents and molecules was removed, cells were washed once with PBS, and serum-free medium was added to serum-starve cells for 24 hours. Following serum-starvation, 200,000 cells in 100 μ L serum-free medium were seeded on solidified growth-factor reduced Matrigel (BD Biosciences) poured onto Costar

Transwell filter-inserts in Boyden chambers (Corning). Complete medium with serum was added to the lower chamber below the inserts and used as a chemoattractant. After incubation at 37 degrees C for 20 hours, cells that had invaded through the Matrigel and to the lower side of the filter-insert were quantified using MTS assay (Promega) and manufacturer protocols. MTS assay was utilized as a more conservative, first-pass assay for quantifying cells that had invaded and the technique of staining with hematoxylin and cell counting described previously(17) that is more sensitive to smaller changes in invasion was used to confirm results trending significance (p<0.2) determined by MTS assay. However, since no data trending significance (p<0.2) were determined using MTS, no follow-up invasion assays with hematoxylin staining and counting were performed.

Western blot. Western blots were used to confirm that overexpression of miRNA resulted in decreased protein levels of previously confirmed targets of miRNA. Following cell harvest and lysate collection, total protein was quantified using the BCA assay (Thermo Scientific) and manufacturer protocols. Cell lysates were then separated electrophoretically using 10% Tris-HCl gels (Bio-Rad). Proteins were transferred to Immuno-Blot PVDF membranes (Bio-Rad) through overnight night transfer at 4 degrees C and following transfer, membranes were washed with TBSt (Boston Bioproducts) and blocked with 5% milk in TBSt. For Western blot analysis, antibodies against PTEN (Invitrogen), STAT-1 (Cell Signaling), TRAF6 (Santa Cruz), and gamma-Tubulin (Cell Signaling) were used to visualize corresponding proteins following established protocols for staining and washing. Proteins were visualized using the Amersham ECL Western Blotting Analysis System (GE Healthcare) following the manusfacturer's protocol. Resulting images were scanned and protein quantitation was performed using Image J software.

Statistical analysis. ANOVA for repeated measures was used to determine if overexpression of candidate miRNA resulted in differential growth, viability, and proliferation of cells compared to negative control. Student's T-tests were used to determine if overexpression of candidate miRNA resulted in differential invasion and migration of cells compared to negative control. Student's T-tests were used to determine if overexpression of candidate miRNA resulted in differential protein levels as determine if overexpression of candidate miRNA resulted in differential protein levels as

ACKNOWLEDGMENTS

Many thanks to Jim Padbury, Joe Bliss, and Jared Robins for their helpful guidance in selecting experiments and their constructive criticism of results. A special thanks to Surendra Sharma and Satyan Kalkunte for generously providing their cell invasion protocol and to Jennifer Joukhadar for instructing us on cell invasion and microscopy techniques.

FUNDING

This work was funded by National Institutes of Health (NIH) grants from the NIEHS [P42ES013660] and [T32ES007272 to M.A.M.] and the NCRR [P20 RR018728].

CONFLICT OF INTEREST STATEMENT

The authors have no conflict of interest to declare.

REFERENCES

- 1 Maccani, M.A. and Marsit, C.J. (2009) Epigenetics in the placenta. *Am J Reprod Immunol*, **62**, 78-89.
- Du, T. and Zamore, P.D. (2007) Beginning to understand microRNA function.
 Cell Res, 17, 661-663.
- 3 Miska, E.A. (2005) How microRNAs control cell division, differentiation and death. *Curr Opin Genet Dev*, **15**, 563-568.
- Bentwich, I., Avniel, A., Karov, Y., Aharonov, R., Gilad, S., Barad, O., Barzilai,
 A., Einat, P., Einav, U., Meiri, E. *et al.* (2005) Identification of hundreds of
 conserved and nonconserved human microRNAs. *Nat Genet*, **37**, 766-770.
- Maccani, M.A., Avissar-Whiting, M., Banister, C.E., McGonnigal, B., Padbury,
 J.F. and Marsit, C.J. (2010) Maternal cigarette smoking during pregnancy is
 associated with downregulation of miR-16, miR-21 and miR-146a in the placenta.
 Epigenetics, 5.
- Avissar-Whiting, M., Veiga, K.R., Uhl, K.M., Maccani, M.A., Gagne, L.A.,
 Moen, E.L. and Marsit, C.J. (2010) Bisphenol A exposure leads to specific
 microRNA alterations in placental cells. *Reprod Toxicol*, 29, 401-406.
- Pineles, B.L., Romero, R., Montenegro, D., Tarca, A.L., Han, Y.M., Kim, Y.M.,
 Draghici, S., Espinoza, J., Kusanovic, J.P., Mittal, P. *et al.* (2007) Distinct subsets
 of microRNAs are expressed differentially in the human placentas of patients with
 preeclampsia. *Am J Obstet Gynecol*, **196**, 261 e261-266.

- 8 Chim, S.S., Shing, T.K., Hung, E.C., Leung, T.Y., Lau, T.K., Chiu, R.W. and Lo,
 Y.M. (2008) Detection and characterization of placental microRNAs in maternal
 plasma. *Clin Chem*, 54, 482-490.
- Mouillet, J.F., Chu, T., Hubel, C.A., Nelson, D.M., Parks, W.T. and Sadovsky, Y.
 (2010) The levels of hypoxia-regulated microRNAs in plasma of pregnant women with fetal growth restriction. *Placenta*, **31**, 781-784.
- 10 Donker, R.B., Mouillet, J.F., Nelson, D.M. and Sadovsky, Y. (2007) The expression of Argonaute2 and related microRNA biogenesis proteins in normal and hypoxic trophoblasts. *Mol Hum Reprod*, **13**, 273-279.
- Mouillet, J.F., Chu, T., Nelson, D.M., Mishima, T. and Sadovsky, Y. (2010) MiR-205 silences MED1 in hypoxic primary human trophoblasts. *FASEB J*, 24, 2030-2039.
- Meng, F., Henson, R., Wehbe-Janek, H., Ghoshal, K., Jacob, S.T. and Patel, T.
 (2007) MicroRNA-21 regulates expression of the PTEN tumor suppressor gene in human hepatocellular cancer. *Gastroenterology*, **133**, 647-658.
- 13 Lou, Y., Yang, X., Wang, F., Cui, Z. and Huang, Y. (2010) MicroRNA-21 promotes the cell proliferation, invasion and migration abilities in ovarian epithelial carcinomas through inhibiting the expression of PTEN protein. *Int J Mol Med*, **26**, 819-827.
- Lu, L.F., Boldin, M.P., Chaudhry, A., Lin, L.L., Taganov, K.D., Hanada, T.,
 Yoshimura, A., Baltimore, D. and Rudensky, A.Y. (2010) Function of miR-146a
 in controlling Treg cell-mediated regulation of Th1 responses. *Cell*, **142**, 914-929.

- Hou, J., Wang, P., Lin, L., Liu, X., Ma, F., An, H., Wang, Z. and Cao, X. (2009)
 MicroRNA-146a feedback inhibits RIG-I-dependent Type I IFN production in
 macrophages by targeting TRAF6, IRAK1, and IRAK2. *J Immunol*, 183, 2150-2158.
- 16 Shaw, L.M., Chao, C., Wewer, U.M. and Mercurio, A.M. (1996) Function of the integrin alpha 6 beta 1 in metastatic breast carcinoma cells assessed by expression of a dominant-negative receptor. *Cancer Res*, **56**, 959-963.
- Chrostowski, M.K., McGonnigal, B.G., Stabila, J.P. and Padbury, J.F. (2010)
 Role of the L-amino acid transporter-1 (LAT-1) in mouse trophoblast cell
 invasion. *Placenta*, **31**, 528-534.
- Kaddar, T., Rouault, J.P., Chien, W.W., Chebel, A., Gadoux, M., Salles, G.,
 Ffrench, M. and Magaud, J.P. (2009) Two new miR-16 targets: caprin-1 and
 HMGA1, proteins implicated in cell proliferation. *Biol Cell*, **101**, 511-524.
- 19 Tran, N., McLean, T., Zhang, X., Zhao, C.J., Thomson, J.M., O'Brien, C. and Rose, B. (2007) MicroRNA expression profiles in head and neck cancer cell lines. *Biochem Biophys Res Commun*, **358**, 12-17.
- 20 Cimmino, A., Calin, G.A., Fabbri, M., Iorio, M.V., Ferracin, M., Shimizu, M.,
 Wojcik, S.E., Aqeilan, R.I., Zupo, S., Dono, M. *et al.* (2005) miR-15 and miR-16 induce apoptosis by targeting BCL2. *Proc Natl Acad Sci U S A*, **102**, 13944-13949.
- Linsley, P.S., Schelter, J., Burchard, J., Kibukawa, M., Martin, M.M., Bartz, S.R.,
 Johnson, J.M., Cummins, J.M., Raymond, C.K., Dai, H. *et al.* (2007) Transcripts

targeted by the microRNA-16 family cooperatively regulate cell cycle progression. *Mol Cell Biol*, **27**, 2240-2252.

- Calin, G.A., Cimmino, A., Fabbri, M., Ferracin, M., Wojcik, S.E., Shimizu, M.,
 Taccioli, C., Zanesi, N., Garzon, R., Aqeilan, R.I. *et al.* (2008) MiR-15a and miR-16-1 cluster functions in human leukemia. *Proc Natl Acad Sci U S A*, **105**, 5166-5171.
- Chen, R.W., Bemis, L.T., Amato, C.M., Myint, H., Tran, H., Birks, D.K.,
 Eckhardt, S.G. and Robinson, W.A. (2008) Truncation in CCND1 mRNA alters
 miR-16-1 regulation in mantle cell lymphoma. *Blood*, **112**, 822-829.
- Qian, Y., Commane, M., Ninomiya-Tsuji, J., Matsumoto, K. and Li, X. (2001)
 IRAK-mediated translocation of TRAF6 and TAB2 in the interleukin-1-induced activation of NFkappa B. *J Biol Chem*, 276, 41661-41667.
- Wang, Q., Dziarski, R., Kirschning, C.J., Muzio, M. and Gupta, D. (2001)
 Micrococci and peptidoglycan activate TLR2-->MyD88-->IRAK-->TRAF- >NIK-->IKK-->NF-kappaB signal transduction pathway that induces
 transcription of interleukin-8. *Infect Immun*, 69, 2270-2276.
- Vabulas, R.M., Ahmad-Nejad, P., da Costa, C., Miethke, T., Kirschning, C.J.,
 Hacker, H. and Wagner, H. (2001) Endocytosed HSP60s use toll-like receptor 2
 (TLR2) and TLR4 to activate the toll/interleukin-1 receptor signaling pathway in innate immune cells. *J Biol Chem*, 276, 31332-31339.
- 27 Taganov, K.D., Boldin, M.P., Chang, K.J. and Baltimore, D. (2006) NF-kappaBdependent induction of microRNA miR-146, an inhibitor targeted to signaling

proteins of innate immune responses. *Proc Natl Acad Sci U S A*, **103**, 12481-12486.

- 28 Li, L., Kang, J. and Lei, W. (2010) Role of Toll-like receptor 4 in inflammationinduced preterm delivery. *Mol Hum Reprod*, 16, 267-272.
- 29 Katze, M.G., He, Y. and Gale, M., Jr. (2002) Viruses and interferon: a fight for supremacy. *Nat Rev Immunol*, 2, 675-687.
- 30 Desai, M., Gayle, D., Han, G. and Ross, M.G. (2007) Programmed hyperphagia due to reduced anorexigenic mechanisms in intrauterine growth-restricted offspring. *Reprod Sci*, **14**, 329-337.
- 31 Tamashiro, K.L. and Moran, T.H. (2010) Perinatal environment and its influences on metabolic programming of offspring. *Physiol Behav*, **100**, 560-566.
- 32 Morioka, Y., Isotani, A., Oshima, R.G., Okabe, M. and Ikawa, M. (2009) Placenta-specific gene activation and inactivation using integrase-defective lentiviral vectors with the Cre/LoxP system. *Genesis*, **47**, 793-798.
- Baudry, A., Mouillet-Richard, S., Schneider, B., Launay, J.M. and Kellermann, O.
 (2010) miR-16 targets the serotonin transporter: a new facet for adaptive
 responses to antidepressants. *Science*, **329**, 1537-1541.
- Qi, L., Bart, J., Tan, L.P., Platteel, I., Sluis, T., Huitema, S., Harms, G., Fu, L.,
 Hollema, H. and Berg, A. (2009) Expression of miR-21 and its targets (PTEN,
 PDCD4, TM1) in flat epithelial atypia of the breast in relation to ductal carcinoma in situ and invasive carcinoma. *BMC Cancer*, 9, 163.

Sinha, M., Ghose, J., Das, E. and Bhattarcharyya, N.P. (2010) Altered
 microRNAs in STHdh(Q111)/Hdh(Q111) cells: miR-146a targets TBP. *Biochem Biophys Res Commun*, 396, 742-747.

Table 1. Predicted and empirically-confirmed gene targets of *miR-16*, *miR-21*, and *miR-146a*. Predicted targets were predicted using the target prediction method described byMaccani, et al., 2010(5).

miRNA	Predicted Targets	Confirmed Targets
miR-16	BCL2L2	Caprin-1(18)
	EDA	HGMA-1 (18)
		SERT (33)
miR-21	PLAG-1	PTEN (12)
	SATB1	PDCD4 (34)
		TM1 (34)
miR-146a	TRAF6	TRAF6 (15)
		STAT-1 (14)
		TBP (35)



Figure 1. HTR8, 3A, and TCL-1 cells transfected with siRNA targeting GAPDH exhibit a 45% knockdown of GAPDH enzyme activity relative to cells transfected with negative control. Error bars show standard error from five replicates.



Figure 2. Overexpression of *miR-16*, *miR-21*, or *miR-146a* in HTR8 (A), 3A (B), and TCL-1 (C) cells does not lead to significantly different cell growth compared to negative control cells.



Figure 3. Overexpression of miR-16, miR-21, or miR-146a in HTR8 (A), 3A (B), and

TCL-1 (C) cells does not lead to significantly different cell viability compared to negative control cells.



Figure 4. Overexpression of *miR-16*, *miR-21*, or *miR-146a* in HTR8 (A), 3A (B), and TCL-1 (C) cells does not lead to significantly different cell proliferation compared to negative control cells.


Figure 5. Overexpression of *miR-16*, *miR-21*, or *miR-146a* in HTR8 (A), 3A (B), and TCL-1 (C) and investigation of differential cell migration compared to negative control cells. Overexpression of *miR-21* in 3A cells results in a trend to increased migration (p<0.08).



Supplemental Figure 1. Migration assays using the technique of staining with hematoxylin and cell counting described previously(17) to confirm results trending significance (p<0.2) as determined using MTS to quantify migrated cells (see Figure 5).
3A cells overexpressing *miR-21* exhibited a trend of increased invasion, 150% versus control (p<0.08), confirming previous results using MTS to quantify migrated cells.



Figure 6. Overexpression of *miR-16*, *miR-21*, or *miR-146a* in HTR8 (A), 3A (B), and TCL-1 (C) cells does not lead to significantly different cell invasion compared to negative control cells.



Figure 7. Western blots for PTEN and STAT-1 demonstrate targeting by *miR-21* and *miR-146a*, respectively. **A**) TCL-1 cells overexpressing *miR-21* express approximately 50% less PTEN protein than TCL-1 cells transfected with negative control (p<0.05). **B**) 3A cells overexpressing *miR-146a* express approximately 26% less STAT-1 protein than 3A cells transfected with negative control (p<0.05). **C**) 3A, TCL-1, and HTR8 cells overexpressing *miR-146a* have no differential levels of TRAF6 compared to cells transfected with negative control.

Chapter Five:

Discussion

This thesis aimed to analyze changes in miRNA expression in the human placenta associated with *in utero* exposure and fetal growth and sought to determine the cellular consequences of such changes in miRNA expression to ultimately understand potential downstream effects on infant growth and development. In this concluding chapter, I summarize my key findings from the previous chapters, attempt to put them into the context of previous and current research, and provide the reader with examples of future experiments important for better understanding links between exposures and fetal growth and miRNA expression in the placenta.

Central to growing interest in the field of epigenetics is the hope that more can be learned about the epigenetic regulatory mechanisms underlying processes of human development and disease. Important links have been examined between environmental exposures and altered expression of miRNA. miRNA expression profiles in the placenta may aid in determination of *in utero* exposures, as well as future disease risk assessment. Many of these important advances in placental epigenetics were discussed in Chapter one of the thesis.

Chapter two detailed the findings from a study which sought to characterize the modulation of 4 candidate placental miRNA by maternal cigarette smoking during pregnancy. Taken as a whole, the work identified the association of maternal cigarette smoking during pregnancy with the downregulation of *miR-16*, *miR-21*, and *miR-146a*, miRNA reported to be involved in critical cell processes. Previous work had demonstrated that nicotine (1) and benzo(a)pyrene (2) interact with the placenta and may affect placental growth and development and the effects of toxicants and agents of cell stress on expression of *miR-146a*, (3-6) as well as on *miR-16* (7) and *miR-21*. (8, 9)

As noted in the discussion section of Chapter two, our observations were limited by incomplete information regarding the duration of cigarette smoking during pregnancy, cigarette per day usage, or more extensive environmental exposure information (such as alcohol usage, environmental pollutant exposure, or secondhand cigarette smoke exposure). Despite these limitations, our data comprised an important first step in determining associations between maternal cigarette smoking during pregnancy and aberrant miRNA expression in the placenta.

Three placental cell lines from different stages of placental development were then utilized to further investigate the effects of components of cigarette smoke, namely nicotine and benzo(a)pyrene. Downregulation of *miR-146a* in TCL-1 cells treated with nicotine and benzo(a)pyrene suggested that *miR-146a* may be especially sensitive to agents of cellular stress. Moreover, this result suggests that two components of cigarette smoke which affect the expression of *miR-146a* in term placentas may be nicotine and benzo(a)pyrene. The lack of differential expression of *miR-16* and *miR-21* in cells treated with nicotine or benzo(a)pyrene does not necessarily rule out that their expression is not modulated by cigarette smoke. Other components of cigarette smoke may modulate the expression of these miRNA. Furthermore, the effect of the mixed exposures encountered with the myriad chemicals in cigarette smoke may be truly responsible for the modulation of expression of these miRNA. Experiments conducted with different components of cigarette smoke and complex mixtures of these components may provide more extensive information on the mechanistic effects of cigarette smoke on miRNA expression.

A final, but nonetheless important, contribution of Chapter two was the design and implementation of a miRNA target prediction strategy. As has been widely

reported, one of the more difficult challenges is miRNA target prediction; complementarity between miRNAs and their target mRNAs is generally far from perfect, and is not required for miRNA to functionally silence the expression of a protein product. As a result of this complexity, a simple search for sequence complementarity between the miRNA and its mRNA target, the basis of many target prediction algorithms, can be expected to produce many false-positive hits. Additionally, most miRNAs are thought to have potentially hundreds of targets, some of which will be targeted more strongly than others, and in cell-type specific contexts. Thus, it is essential to devise a strategy which predicts targets based not solely on sequence homology, but which also incorporates additional characterizations when choosing which targets should be examined with further testing.

Our target prediction strategy was as follows. Three target prediction algorithms were used to predict targets for miRNA of interest, and a fourth algorithm was used to evaluate predicted targets for base-pairing and minimum free energy. The three algorithms used for target prediction were *miRanda* (September 2008 release, available online at http://www.microrna.org/microrna/home.do), PicTar (as cited by (10) and available online at <u>http://pictar.mdc-berlin.de/cgi-bin/new PicTar vertebrate.cgi</u>), and *TargetScan 5.1* (available online at <u>http://www.targetscan.org/</u>). In order to be considered a predicted target for further investigation, the target must have appeared in the top 100 targets in all three prediction algorithms. The RNA-hybrid algorithm was used to evaluate predicted targets for information on secondary structure and thermodynamic stability of the miRNA-mRNA duplex (11). RNA-hybrid is available online at http://bibiserv.techfak.uni-bielefeld.de/rnahybrid/submission.html. The design and

implementation of target prediction algorithms and strategies are important for determining which gene targets to empirically validate using proteomic approaches as targets of miRNA.

A number of future experiments aimed at further characterizing the modulation of miRNA by cigarette smoke are needed. Future experiments conducted using cigarette smoke condensate as reported by Crane-Godreau and colleagues (12) may prove especially useful in characterizing the effects of complex mixtures of components of cigarette smoke on placental cells. Of course, such experiments also have the caveat of presenting challenges in determining the specific composition of the complex mixture as well as physiological relevance of exposure to such a mixture on placental cells. Other more extensive experiments aimed at better understanding the more complex effects of organism-wide cigarette smoke exposure might be designed using a mouse model of chronic environmental tobacco smoke exposure as described previously (13). In such an experimental design, pregnant mice would be exposed to environmental tobacco smoke throughout their entire gestation and euthanized at key timepoints in gestation, with placentas and other key organs harvested for measurement of miRNA expression. Findings would elaborate on work done to characterize the association of maternal cigarette smoking during pregnancy with the dysregulation of placental miRNA. Additionally, such findings using an *in vivo* model system may provide key mechanistic data that may further strengthen current epidemiologic association studies with findings from a model system experimental paradigm.

In Chapter three, I showed data from an investigation of the expression of key candidate miRNA in a large, population-based series of primary human placenta samples

and their association with poor fetal growth, specifically identifying that reduced expression of *miR-16* and *miR-21* are strongly associated with growth restriction. While I controlled for a number of confounders in my analysis and while my data suggest that reduced *miR-16* and *miR-21* expression in the placenta may be associated with SGA status, there remains a somewhat difficult challenge of data interpretation. Because this study was performed using human tissues, it is a retrospective association study – one that is limited by the fact that it can reveal associations between expression of miRNA and fetal growth but it cannot reveal causality. Furthermore, the study cannot adequately answer the question of whether dysregulated miRNA expression in the placenta leads to growth restriction (i.e. SGA status) or growth restriction leads to dysregulated miRNA expression in the placenta. Future animal models designed to study growth restriction (14, 15) will be essential in answering this "chicken and egg" causality dilemma.

A number of groups had previously described placental miRNA expression associations with maternal conditions such as preeclampsia (16), with maternal cigarette smoking during pregnancy (17), and as markers of pregnancy itself (18). Furthermore, more is being uncovered about the role of *miR-16* and *miR-21* in regulating key cellular processes, especially the involvement of *miR-16* in regulating cell cycle progression (19) and *miR-21*'s capability of regulating cell cycling and cell proliferation (9).

miR-21 had been shown to target a number of key regulators of cell proliferation and migration, including but not limited to PLAG1 (20) and PTEN (21, 22). Meng and coworkers showed that *miR-21* regulates PTEN in human hepatocellular cancer (21), and Lou and colleagues demonstrated that in ovarian epithelial carcinomas, *miR-21* promotes proliferation, invasion and migration abilities by inhibiting PTEN (22). Like *miR-21* and many miRNA, *miR-16* exhibits tissue-specific function and expression. In a variety of cancer cell lines, previous work has shown that *miR-16* is involved in the induction of apoptosis by targeting *BCL-2* (23) and in cell cycle regulation by targeting CDK6 (24), CDC27 (25), and CARD10 (26). In other cell types, *miR-16* has different functions, such as targeting HMGA1 and Caprin-1 (27), further suggesting that *miR-16* may have cell-type-specific function and expression (23, 27).

Mechanistic research using model systems is needed to further elucidate the pathways regulated by *miR-16* and *miR-21* and to better determine the functional consequence of downregulation of miR-16 and miR-21 in the placenta. A number of animal models have been generated to further study low birthweight in a controlled, experimental system (14, 15), and it will be important to consider the role of miRNA in these animal models. Recently, Pernaute and colleagues have made important steps in characterizing the roles of miRNA in controlling cell cycle processes in the developing mouse embryo as well as extra-embryonic tissues (28). Future work may include studying the effects of viral or bacterial infections at different stages during pregnancy on placental miRNA expression in the mouse. Such experiments could be accomplished by exposing mice to sub-lethal doses of Influenza A as described by Mackenzie and colleagues (29) or LPS as described by Clark (30) and investigating subsequent effects on miRNA expression in placental tissues. One might hypothesize that miRNA particularly sensitive to infections and miRNA reported to be involved in regulating inflammatory pathways, such as *miR-146a*, might be especially sensitive to such infections. Additionally, the effects of Influenza A-induced or LPS-induced dysregulation of miRNA on rates of resorption, pre-term birth, and birthweight might be investigated to

further understand the roles of miRNA in mediating or responding to infection during pregnancy.

In summary, data from Chapter three suggesting that low expression of *miR-16* and *miR-21* in the placenta is associated with poor fetal growth may have many important implications. Future work to determine the roles of miRNA in specific pathways leading to altered fetal growth will be crucial to better understanding fetal growth as both a marker of the intrauterine environment as well as a developmental outcome and in better comprehending the developmental origins of health and disease.

Chapter four reported findings from important follow-up experiments in placental cell lines to determine effects of aberrant miRNA expression on important cell processes and function, as well as to determine the effects of dysregulated miRNA expression on protein levels of target genes. More specifically, our work sought to investigate the effects of miRNA overexpression on a number of key placental cell processes, specifically cell viability, growth, proliferation, invasion, and migration, as well as further confirm previously empirically validated targets of miRNA in placental cells. Taken collectively, our work contributes to the better understanding of how dysregulated *miR-16*, *miR-21*, and *miR-146a* may alter key cellular processes and expression of target genes and further explore how miRNA may control cell processes in placental cells from various stages of pregnancy.

As with many miRNA, *miR-16* exhibits tissue-specific function and expression. We hypothesized that dysregulation of *miR-16* in the placenta may lead to aberrant expression of its targets and may lead to dysregulated placental cell processes. While our data suggest that overexpression of *miR-16* does not lead to significant differences in cell

viability, growth, proliferation, invasion, or migration, future studies addressing potential effects on apoptosis and target gene expression may prove to be especially useful in better determining the biological ramifications of dysregulated *mir-16* expression in the placenta, which may ultimately lead to abnormalities in the placenta that might result in reduced infant birthweight. Future work using both cell lines as well as animal models will be important in better characterizing the effects of dysregulated *miR-16* on the placenta.

miR-21 has also been shown to target a number of key genes, including but not limited to PTEN (21, 22). Our results that *miR-21* targets PTEN in the TCL-1 placental cell line further suggest that *miR-21* may be one mechanism regulating PTEN expression in the placenta. Downregulation of *miR-21* in the placenta could thus lead to a dysregulation of PTEN and could result in decreased invasion of the maternal decidua, decreased migration, and decreased growth – the opposite of what has been observed to occur in the case of upregulated *miR-21* (21). Our results suggesting that overexpression of *miR-21* in 3A cells leads to a trend towards increased migration compared to cells transfected with negative control may further underscore the involvement of *miR-21* in regulating placental cell migration, especially in first trimester placental cells. Future work using *in vivo* animal models of dysregulated placental miRNA expression will be crucial in determining the causal effects aberrant placental *miR-21* expression on placental cell migration.

Our *in-silico* analysis predicted TRAF6 as a target for *miR-146a*, suggesting that downregulation of *miR-146a* could result in an upregulation of TRAF6, a protein also important for signaling through the NF $\kappa\beta$ signaling pathway (31) as well as a mediator of

inflammation in the toll-like receptor 4 (TLR4) pathway (32, 33). Previous reports using different target prediction strategies have hypothesized TRAF6 as a target of miR-146a (17, 34) and have validated TRAF6 as a target in murine macrophages (35). TLR4 mediates the inflammatory response, and aberrant TLR-4 signaling is associated with inflammation-induced preterm delivery (36). Such a consequence could be the result of dysregulated TRAF6 production as a result of downregulation of miR-146a. TRAF6 overexpression due to downregulation of *miR-146a* could result in overactive TLR signaling which could have a number of downstream consequences for both placenta and fetus. Unfortunately, our Western blot analysis suggested that TRAF6 protein levels were not significantly different in placental cells overexpressing *miR-146a* versus cells transfected with negative control. This may be due, in part, to a more complex regulation of TRAF6 in these placental cell lines than is currently understood. While *miR-146a* may be capable of regulating TRAF6, it does not appear to be the major or most important regulator of TRAF6 in these placental cell lines. Further investigation into such relationships and regulatory networks using *in vivo* animal models would prove vital into better understanding both the role of TRAF6 in TLR4 signaling, as well as the association of aberrant expression of *miR-146a* with adverse pregnancy outcomes.

Previous work has demonstrated that STAT-1 is targeted by *miR-146a* (37). Our Western blot analysis revealed that 3A cells overexpressing *miR-146a* had 25% less STAT-1 protein than did cells transfected with negative control, a potentially important result since STAT-1 plays a role as a key transcription factor. STAT-1 is activated upon interferon (IFN) binding to its receptor. Once phosphorylated by JAKs, STAT-1 forms a homodimer and translocates to the nucleus where it binds the Interferon-Gamma activation site (GAS) in the promoter of interferon stimulated genes (ISG) (38). Our data suggest that STAT-1 may be regulated, in part, by *miR-146a* in 3A placental cells. Dysregulation of *miR-146a* may thereby lead to aberrant production of STAT-1 and dysregulated transcription of ISG, potentially leading to a cascade of events which may negatively alter placental function and ultimately lead to altered fetal programming. Once again, future work using animal models of aberrant placental miRNA expression will be important for further determining the role of dysregulated *miR-146a* expression in the placenta.

Whenever one is working with placental cell lines, one must be aware of the potential limitations involved. Cell lines have important utility as tools for proof-ofconcept experiments but cannot serve as adequate substitutes for animal model systems used to answer questions of causality. One recent study pointed out several limitations regarding human placental cell lines, including but not limited to unexpected effects of transformation, unanticipated cell contamination resulting from site of initial derivation, and varying immunological characteristics of different human placental cell lines (39). In addition, varying degrees of transfection efficiency can impact the likelihood of observing hypothesized changes in cell culture. For example, transfection efficiency for HTR8, 3A, and TCL-1 cells using our transfection protocol was estimated at approximately 45%; this equates to only 45% of the cells in culture actually being transfected with the pre-miR mimic or negative control of interest. Higher transfection efficiencies may have led to more cells actually being transfected with the pre-miR mimic of interest and may have resulted in increasing the likelihood of observing phenotypic changes to placental cells and cell function in culture. Collectively,

considering these limitations is important when determining the scope of conclusions from data using human placental cell lines; nonetheless, human placental cell lines can serve as important tools for proof-of-concept experiments, providing researchers basis for following up such experiments in primary cell culture or in *in vivo* animal models.

Collectively, dysregulation of these miRNA can lead to aberrant expression of their targets and may affect a number of key placental cell processes, ultimately leading to aberrant cell cycle regulation, growth, immunomodulation, and development in the placenta. These changes in target gene expression may have further effects downstream for both placenta and fetus, ultimately resulting in altered fetal programming. Additionally, more work to determine the functional effects of miRNA crosstalk – that is, miRNAs whose differential expression may have additive associations with risk for phenotype or disease – will be necessary to better understand the complex regulatory networks involved. Also, future studies comparing the severity of effects resulting from the downregulation of a particular miRNA versus the upregulation of that miRNA will be important in further understanding the boundaries of what might be considered the range of specific miRNA expression that allows for biological equilibrium; this range of miRNA expression may be miRNA-specific as well as cell type-specific. Future studies are ongoing to further characterize the effects of environmental exposures on placental miRNA, on understanding the downstream phenotypes of aberrant miRNA expression and aberrant miRNA target gene expression, and in investigating the placenta as a record of the intrauterine environment. In such a way, the environment can act through the alteration of miRNA which may lead to aberrant mediation of placental function, potentially leading to downstream effects on infant growth and development.

Understanding how the variety of stimuli that comprise the intrauterine environment elicit their effects on the placental epigenome and how these epigenetic effects mechanistically alter placental function will be important in better characterizing the downstream effects of the intrauterine environment on fetal programming.

In conclusion, this work further elucidates the association of placental miRNA expression with *in utero* exposures and fetal growth and seeks to better understand how an adverse intrauterine environment acts through epigenetic mechanisms to alter infant and child health. Data have suggested that environmental exposures can alter miRNA expression in in-vitro systems (3) and various tissues (40) and that altered levels of miRNA expression can be associated with particular diseases or risk factor for disease (41). Research on diseases of pregnancy – such as finding miRNA whose expression may be associated with preeclampsia (16, 42) or gestational diabetes – might be expanded to include work to better understand associations between epigenetic factors in the placenta and diseases of childhood and adolescence, as well as alterations to normal placental immunology and function.

As has been reported previously (43), using the placenta as a reference tissue allows researchers to utilize an important residual tissue whose respective miRNA expression may prove to be powerful biomarkers possessing predictive capability for a number of diseases or disease progression. Modern advances in bioinformatics – such as pathway analysis tools and target gene prediction software – as well as advances in technology, such as microarray technology, have given researchers and clinicians tools to better detect patterns in a number of epigenetic mechanisms, including DNA methylation, imprinting, histone modification and miRNA expression, that are associated with

particular exposures and diseases. Aberrant patterns of miRNA expression or DNA methylation may ultimately serve as biomarkers for exposure, disease burden, or even as "early indicator" diagnostics of increased risk for developing future disease or disorders.

Research continues to elucidate a better understanding of the epigenetic regulatory mechanisms of in the placenta. Panels of miRNA determined to be associated with particular exposures, such as heavy metal exposure or other xenobiotic toxicants, may prove to be useful biomarkers in assessing a newborn's degree of *in utero* exposure to such toxicants; in this way, an aberrant miRNA expression profile previously associated with a heavy metal or toxicant exposure might give clinicians an important indication of exposure level in their patients. Such miRNA biomarkers may be especially important in circumstances where exposure assessments have not yet been conducted. miRNA biomarkers may prove essential for developing new diagnostics for early diagnosis of risk factor and levels of exposure; in this way, panels of miRNA whose altered expression is shown to be associated with a particular disease or risk for disease might be investigated to inform clinicians of a patient's particular disease state or risk for Additionally, these aberrant patterns of miRNA expression may identify disease. previously unknown pathways targeted for alteration, which, in turn, may serve as targets for novel drug treatment or prevention strategies. These epigenetic biomarkers can be brought from the benchtop to the bedside and will be useful in helping clinicians better diagnose and possibly prevent the onset of disease.

Because poor maternal forecasts leading to low birthweight may not accurately predict the post-birth environment and because low birthweight increases one's risk for a number of diseases later in life (44), more research is necessary to more fully understand

the pathways whose dysregulation may ultimately affect birthweight. In addition, it will be important to gain a better understanding of how *in utero* exposures and adverse intrauterine conditions mechanistically lead to changes which alter the placental epigenome and may thereby alter placental function, ultimately leading to impacts on fetal programming and future health. Future work to determine the roles of miRNA in pathways responsive to *in utero* exposures and in pathways leading to altered fetal growth will be key to better understanding fetal growth as both a marker of the intrauterine environment as well as a developmental outcome and in better comprehending the developmental origins of health and disease.

REFERENCES

- Lambers, D.S. and Clark, K.E. (1996) The maternal and fetal physiologic effects of nicotine. *Semin Perinatol*, **20**, 115-126.
- 2 Guyda, H.J., Mathieu, L., Lai, W., Manchester, D., Wang, S.L., Ogilvie, S. and Shiverick, K.T. (1990) Benzo(a)pyrene inhibits epidermal growth factor binding and receptor autophosphorylation in human placental cell cultures. *Mol Pharmacol*, **37**, 137-143.
- 3 Marsit, C.J., Eddy, K. and Kelsey, K.T. (2006) MicroRNA responses to cellular stress. *Cancer Res*, **66**, 10843-10848.
- Curtale, G., Citarella, F., Carissimi, C., Goldoni, M., Carucci, N., Fulci, V.,
 Franceschini, D., Meloni, F., Barnaba, V. and Macino, G. (2010) An emerging
 player in the adaptive immune response: microRNA-146a is a modulator of IL-2
 expression and activation-induced cell death in T lymphocytes. *Blood*, **115**, 265-273.
- Wang, J.F., Yu, M.L., Yu, G., Bian, J.J., Deng, X.M., Wan, X.J. and Zhu, K.M.
 (2010) Serum miR-146a and miR-223 as potential new biomarkers for sepsis.
 Biochem Biophys Res Commun.
- 6 Avissar-Whiting, M., Veiga, K., Uhl, K., Maccani, M., Gagne, L., Moen, E. and Marsit, C.J. (2010) Bisphenol A exposure leads to specific microRNA alterations in placental cells. *Reprod Toxicol*.
- 7 Pothof, J., Verkaik, N.S., van, I.W., Wiemer, E.A., Ta, V.T., van der Horst, G.T., Jaspers, N.G., van Gent, D.C., Hoeijmakers, J.H. and Persengiev, S.P. (2009)

MicroRNA-mediated gene silencing modulates the UV-induced DNA-damage response. *EMBO J*, **28**, 2090-2099.

- Weber, M., Baker, M.B., Moore, J.P. and Searles, C.D. (2010) MiR-21 is induced in endothelial cells by shear stress and modulates apoptosis and eNOS activity.
 Biochem Biophys Res Commun.
- Wang, P., Zou, F., Zhang, X., Li, H., Dulak, A., Tomko, R.J., Jr., Lazo, J.S.,
 Wang, Z., Zhang, L. and Yu, J. (2009) microRNA-21 negatively regulates
 Cdc25A and cell cycle progression in colon cancer cells. *Cancer Res*, 69, 8157-8165.
- Lall, S., Grun, D., Krek, A., Chen, K., Wang, Y.L., Dewey, C.N., Sood, P.,
 Colombo, T., Bray, N., Macmenamin, P. *et al.* (2006) A genome-wide map of
 conserved microRNA targets in C. elegans. *Curr Biol*, 16, 460-471.
- 11 Kruger, J. and Rehmsmeier, M. (2006) RNAhybrid: microRNA target prediction easy, fast and flexible. *Nucleic Acids Res*, **34**, W451-454.
- Crane-Godreau, M.A., Maccani, M.A., Eszterhas, S.K., Warner, S.L., Jukosky,
 J.A. and Fiering, S. (2009) Exposure to Cigarette Smoke Disrupts CCL20 Mediated Antimicrobial Activity in Respiratory Epithelial Cells. *Open Immunol J*,
 2, 86-93.
- Xiong, Z., Leme, A.S., Ray, P., Shapiro, S.D. and Lee, J.S. (2011) CX3CR1+
 Lung Mononuclear Phagocytes Spatially Confined to the Interstitium Produce
 TNF-{alpha} and IL-6 and Promote Cigarette Smoke-Induced Emphysema. *J Immunol*.

- 14 Desai, M., Gayle, D., Han, G. and Ross, M.G. (2007) Programmed hyperphagia due to reduced anorexigenic mechanisms in intrauterine growth-restricted offspring. *Reprod Sci*, 14, 329-337.
- 15 Tamashiro, K.L. and Moran, T.H. (2010) Perinatal environment and its influences on metabolic programming of offspring. *Physiol Behav*, **100**, 560-566.
- 16 Pineles, B.L., Romero, R., Montenegro, D., Tarca, A.L., Han, Y.M., Kim, Y.M., Draghici, S., Espinoza, J., Kusanovic, J.P., Mittal, P. *et al.* (2007) Distinct subsets of microRNAs are expressed differentially in the human placentas of patients with preeclampsia. *Am J Obstet Gynecol*, **196**, 261 e261-266.
- Maccani, M.A., Avissar-Whiting, M., Banister, C.E., McGonnigal, B., Padbury,
 J.F. and Marsit, C.J. (2010) Maternal cigarette smoking during pregnancy is
 associated with downregulation of miR-16, miR-21 and miR-146a in the placenta.
 Epigenetics, 5.
- Chim, S.S., Shing, T.K., Hung, E.C., Leung, T.Y., Lau, T.K., Chiu, R.W. and Lo,
 Y.M. (2008) Detection and characterization of placental microRNAs in maternal
 plasma. *Clin Chem*, 54, 482-490.
- Liu, Q., Fu, H., Sun, F., Zhang, H., Tie, Y., Zhu, J., Xing, R., Sun, Z. and Zheng,
 X. (2008) miR-16 family induces cell cycle arrest by regulating multiple cell
 cycle genes. *Nucleic Acids Res*, 36, 5391-5404.
- Tran, N., McLean, T., Zhang, X., Zhao, C.J., Thomson, J.M., O'Brien, C. and
 Rose, B. (2007) MicroRNA expression profiles in head and neck cancer cell lines.
 Biochem Biophys Res Commun, 358, 12-17.

- Meng, F., Henson, R., Wehbe-Janek, H., Ghoshal, K., Jacob, S.T. and Patel, T.
 (2007) MicroRNA-21 regulates expression of the PTEN tumor suppressor gene in human hepatocellular cancer. *Gastroenterology*, 133, 647-658.
- 22 Lou, Y., Yang, X., Wang, F., Cui, Z. and Huang, Y. (2010) MicroRNA-21 promotes the cell proliferation, invasion and migration abilities in ovarian epithelial carcinomas through inhibiting the expression of PTEN protein. *Int J Mol Med*, **26**, 819-827.
- Cimmino, A., Calin, G.A., Fabbri, M., Iorio, M.V., Ferracin, M., Shimizu, M.,
 Wojcik, S.E., Aqeilan, R.I., Zupo, S., Dono, M. *et al.* (2005) miR-15 and miR-16 induce apoptosis by targeting BCL2. *Proc Natl Acad Sci U S A*, **102**, 13944-13949.
- 24 Linsley, P.S., Schelter, J., Burchard, J., Kibukawa, M., Martin, M.M., Bartz, S.R., Johnson, J.M., Cummins, J.M., Raymond, C.K., Dai, H. *et al.* (2007) Transcripts targeted by the microRNA-16 family cooperatively regulate cell cycle progression. *Mol Cell Biol*, **27**, 2240-2252.
- Calin, G.A., Cimmino, A., Fabbri, M., Ferracin, M., Wojcik, S.E., Shimizu, M.,
 Taccioli, C., Zanesi, N., Garzon, R., Aqeilan, R.I. *et al.* (2008) MiR-15a and miR16-1 cluster functions in human leukemia. *Proc Natl Acad Sci U S A*, **105**, 51665171.
- Chen, R.W., Bemis, L.T., Amato, C.M., Myint, H., Tran, H., Birks, D.K.,
 Eckhardt, S.G. and Robinson, W.A. (2008) Truncation in CCND1 mRNA alters
 miR-16-1 regulation in mantle cell lymphoma. *Blood*, **112**, 822-829.

- Kaddar, T., Rouault, J.P., Chien, W.W., Chebel, A., Gadoux, M., Salles, G.,
 Ffrench, M. and Magaud, J.P. (2009) Two new miR-16 targets: caprin-1 and
 HMGA1, proteins implicated in cell proliferation. *Biol Cell*, 101, 511-524.
- 28 Pernaute, B., Spruce, T., Rodriguez, T.A. and Manzanares, M. (2011) miRNAmediated regulation of cell signaling and homeostasis in the early mouse embryo. *Cell Cycle*, 10.
- Mackenzie, J.S., Williams, K. and Papadimitriou, J. (1977) Influenza A virus and its influence on the outcome of pregnancy in the mouse. *Dev Biol Stand*, **39**, 489-496.
- Clark, D.A., Manuel, J., Lee, L., Chaouat, G., Gorczynski, R.M. and Levy, G.A.
 (2004) Ecology of danger-dependent cytokine-boosted spontaneous abortion in
 the CBA x DBA/2 mouse model. I. Synergistic effect of LPS and (TNF-alpha +
 IFN-gamma) on pregnancy loss. *Am J Reprod Immunol*, **52**, 370-378.
- Qian, Y., Commane, M., Ninomiya-Tsuji, J., Matsumoto, K. and Li, X. (2001)
 IRAK-mediated translocation of TRAF6 and TAB2 in the interleukin-1-induced activation of NFkappa B. *J Biol Chem*, 276, 41661-41667.
- Wang, Q., Dziarski, R., Kirschning, C.J., Muzio, M. and Gupta, D. (2001)
 Micrococci and peptidoglycan activate TLR2-->MyD88-->IRAK-->TRAF- >NIK-->IKK-->NF-kappaB signal transduction pathway that induces
 transcription of interleukin-8. *Infect Immun*, 69, 2270-2276.
- Vabulas, R.M., Ahmad-Nejad, P., da Costa, C., Miethke, T., Kirschning, C.J.,
 Hacker, H. and Wagner, H. (2001) Endocytosed HSP60s use toll-like receptor 2

(TLR2) and TLR4 to activate the toll/interleukin-1 receptor signaling pathway in innate immune cells. *J Biol Chem*, **276**, 31332-31339.

- Taganov, K.D., Boldin, M.P., Chang, K.J. and Baltimore, D. (2006) NF-kappaB dependent induction of microRNA miR-146, an inhibitor targeted to signaling
 proteins of innate immune responses. *Proc Natl Acad Sci U S A*, **103**, 12481 12486.
- Hou, J., Wang, P., Lin, L., Liu, X., Ma, F., An, H., Wang, Z. and Cao, X. (2009)
 MicroRNA-146a feedback inhibits RIG-I-dependent Type I IFN production in
 macrophages by targeting TRAF6, IRAK1, and IRAK2. *J Immunol*, 183, 2150-2158.
- 36 Li, L., Kang, J. and Lei, W. (2010) Role of Toll-like receptor 4 in inflammationinduced preterm delivery. *Mol Hum Reprod*, 16, 267-272.
- Lu, L.F., Boldin, M.P., Chaudhry, A., Lin, L.L., Taganov, K.D., Hanada, T.,
 Yoshimura, A., Baltimore, D. and Rudensky, A.Y. (2010) Function of miR-146a
 in controlling Treg cell-mediated regulation of Th1 responses. *Cell*, 142, 914-929.
- 38 Katze, M.G., He, Y. and Gale, M., Jr. (2002) Viruses and interferon: a fight for supremacy. *Nat Rev Immunol*, 2, 675-687.
- 39 Apps, R., Murphy, S.P., Fernando, R., Gardner, L., Ahad, T. and Moffett, A. (2009) Human leucocyte antigen (HLA) expression of primary trophoblast cells and placental cell lines, determined using single antigen beads to characterize allotype specificities of anti-HLA antibodies. *Immunology*, **127**, 26-39.
- 40 Reamon-Buettner, S.M., Mutschler, V. and Borlak, J. (2008) The next innovation cycle in toxicogenomics: environmental epigenetics. *Mutat Res*, **659**, 158-165.

- 41 Callis, T.E. and Wang, D.Z. (2008) Taking microRNAs to heart. *Trends Mol Med*,
 14, 254-260.
- 42 Zhu, X.M., Han, T., Sargent, I.L., Yin, G.W. and Yao, Y.Q. (2009) Differential expression profile of microRNAs in human placentas from preeclamptic pregnancies vs normal pregnancies. *Am J Obstet Gynecol*.
- 43 Maccani, M.A. and Marsit, C.J. (2009) Epigenetics in the placenta. *Am J Reprod Immunol*, **62**, 78-89.
- 44 Argente, J., Mehls, O. and Barrios, V. (2010) Growth and body composition in very young SGA children. *Pediatr Nephrol*, 25, 679-685.

Appendix:

Birthweight is associated with DNA promoter methylation of the glucocorticoid receptor in human placenta

Amanda C. Filiberto*, Matthew A. Maccani*, Devin Koestler, Charlotte Wilhelm-

Benartzi, Michele Avissar-Whiting, Carolyn E. Banister, Luc A. Gagne, and Carmen J.

Marsit

*Contributed equally to this work

Epigenetics, In Press

Birthweight is associated with DNA promoter methylation of the glucocorticoid receptor in human placenta

Amanda C. Filiberto^{1*}, Matthew A. Maccani^{1*}, Devin Koestler², Charlotte Wilhelm-Benartzi², Michele Avissar-Whiting¹, Carolyn E. Banister¹, Luc A. Gagne¹, and Carmen J. Marsit^{1,2}

¹Department of Pathology and Laboratory Medicine, ²Center for Environmental Health and Technology, Department of Community Health, Brown University, Providence, RI,

USA

Corresponding Author Carmen J. Marsit, Box G-E55 Providence, RI 02912

Telephone: (401) 863-6508 Fax: (401) 863-9008

Email: <u>Carmen_Marsit@brown.edu</u>

<u>Amanda_Filiberto@brown.edu;</u> <u>Matthew_Maccani@brown.edu;</u> <u>Devin_Koestler@brown.edu;</u> <u>Charlotte_Wilhelm@brown.edu;</u>

Michele_Avissar@brown.edu; Carolyn_Banister@brown.edu; Luc_Gagne@brown.edu;

Carmen_Marsit@brown.edu

The authors have declared that no conflict of interest exists.

*These authors have contributed equally to this work.

ABSTRACT

Birthweight has been associated with a number of health outcomes throughout life. Crucial to proper infant growth and development is the placenta, and alterations to placental gene function may reflect differences in the intrauterine environment which functionally contribute to infant growth and may ultimately affect the child's health. To examine if epigenetic alteration to the glucocorticoid receptor (GR) gene was linked to infant growth, we analyzed 480 human placentas for differential methylation of the GR gene, exon 1F, and examined how this variation in methylation extent was associated with fetal growth. Multivariable linear regression revealed a significant association (p<0.0001) between differential methylation of the GR gene and large for gestational age (LGA) status. Our work is one of the first to link infant growth as a measure of the intrauterine environment and epigenetic alterations to the GR and suggests that DNA methylation may be a critical determinant of placental function.

KEYWORDS: DNA methylation, placenta, fetal development, birthweight, epigenetics

RUNNING TITLE: Birthweight is associated with placental GR methylation

INTRODUCTION

The period of intrauterine development represents a sensitive period where disruption or modification of the intrauterine environment can influence fetal development as well as lead to programming of health throughout the life course.^{1, 2} In particular, environmental exposures during intrauterine development have been shown to be associated with several chronic diseases in adulthood³ and adverse intrauterine conditions have been shown to associate with fetal growth outcomes.^{2, 4-6} There is growing interest in determining the functional molecular basis of this fetal programming.

The metabolic and endocrine activity and ability to transport nutrients, water, gas and waste products marks the placenta as a vital organ for the growing fetus.⁷ Dysregulated placental gene expression may result from *in utero* exposures and thus represent a record of the intrauterine environment during pregnancy. These alterations may also play a critical role in establishing a fetal program which can influence health outcomes throughout life.⁷ Thus, as a functional marker of intrauterine environment,⁸ investigation of epigenetic alterations in the placenta may further the understanding of the molecular mechanisms behind many developmental outcomes which may be influenced by adverse intrauterine conditions.

Low birthweight has been associated with increased infant morbidity and mortality and with an increased risk for certain diseases later in life, particularly those comprising the metabolic syndrome.⁹ High birthweight, as well, has been shown to have serious adverse health outcomes in both developing children and later in life.¹⁰ Previous studies have shown high birthweight as a risk factor for insulin resistance,¹¹ obesity,¹² and cancers¹³ such as leukemia and breast, prostate and colon cancers. Studies to

determine the potential risk factors for mothers of large for gestational age (LGA) babies are ongoing,¹⁰ and more research is needed to identify potential risk factors behind this developmental outcome.

The glucocorticoid receptor (GR), a known mediator of glucocorticoid signaling, has been shown to be regulated, in part, by epigenetic mechanisms, specifically DNA methylation.¹⁴⁻¹⁹ Both rat and human studies have shown an association between methylation of the hippocampal GR gene and early postnatal outcomes.¹⁶ Early maternal care of rat pups leads to altered methylation status of the NGFI-A consensus binding site within the promoter of the GR gene in the hippocampus, and these changes have been linked to altered stress responses later in life, as well as to growth restriction in-utero.^{14,} ^{15, 20-22} Oberlander and colleagues suggested a potential association between methylation status of the analogous human NGFI-A binding site in promoter region 1F of the GR gene in infant cord blood and maternal mood.^{18, 20}

As birthweight is known to be associated with a variety of later life health outcomes and aberrant methylation of the GR gene has been linked with adverse developmental outcomes in animal models, we sought to investigate the associations between birthweight and GR promoter methylation in human placenta.

RESULTS

We examined the variation in the extent of methylation of the GR gene promoter 1F region in a large series of human placenta samples obtained from term infants of various birthweights. **Table 1** shows the demographics of the study population in total (n=480) and by birthweight grouping: small for gestational age (SGA, <10th percentile of birthweight for gestational age, n=102), appropriate for gestational age (AGA, 10-90th percentile of birthweight for gestational age, n=343), and large for gestational age (LGA, >90th percentile of weight for gestational age, n=35). Distributions of gestational age, infant gender, maternal race, prenatal vitamin use, and maternal age were not significantly different across birthweight groupings. As expected, maternal tobacco use during pregnancy was more common amongst SGA infants, and LGA infants were more often delivered through Caesarean section. There was a low prevalence of recreational drug use, alcohol use, and maternal gestational diabetes across all groups.

Figure 1 depicts the difference in the extent of methylation of the 13 CpG sites examined in the GR exon 1F region in SGA, AGA, and LGA placentas. Differential methylation was noted across groups at all CpG sites in the GR exon 1F but only CpG site 7 and the mean across all CpG sites showed a significantly different extent of methylation across the groups following a Bonferroni correction for multiple comparisons (p<0.0036).

As these sites are in close proximity, and as methylation likely occurs throughout the region in a coordinated fashion, we examined the correlation between the extent of methylation at the individual CpG sites, using pairwise correlations for methylation extent between each of the 13 CpG sites (**Figure 2**). Of note, there is a moderate to

strong correlation in methylation between many of the 13 CpG sites. Moreover, the correlation in methylation between CpG sites appears to be a function of the distance between sites, consistent with the findings of Nautiyal and colleagues.²³ This relatively high degree of correlation suggests that use of the mean may be a reliable marker of the extent of methylation across this region.

Examination of the correlation of each of the covariates listed in **Table 1** with log mean GR methylation extent was conducted (**Supplemental Table 1**). This examination revealed that birthweight was significantly correlated with mean GR methylation extent (r=0.16, P=0.0004). None of the other demographic or clinical characteristics of the subjects was significantly correlated with GR methylation. To further investigate the association between birthweight category and mean GR methylation, controlling for potential confounders, we performed multivariable linear regression (**Table 2**). Analysis revealed that in LGA infants, the log mean percent methylation of the GR increased significantly by 0.40 (p<0.0001), controlled for gestational age, maternal age, maternal ethnicity, tobacco use during pregnancy, alcohol use during pregnancy, recreational drug use during pregnancy, prenatal vitamin use, delivery method, maternal gestational diabetes, and infant gender. Being SGA was not a significant predictor of GR methylation extent, nor were any of the covariates included as confounders in the model, consistent with the univariate findings.

To demonstrate the functional role of promoter methylation on GR gene expression, the placenta choriocarcinoma cell line JEG3 was treated with the DNA methyltransferase inhibitor 5-aza-2'-deoxycytidine (5-AzaC). Pyrosequencing of bisulfite modified DNA from JEG-3 cells treated with 5-azaC showed decreased

methylation with increasing doses of 5-azaC across all 13 CpG sites in the GR (**Fig. 3a**). The mean methylation across this region demonstrated an almost 20% decrease in cells treated with 1.25 μ M 5-azaC, and a 50% decrease in methylation in cells treated with 2.5 μ M 5-azaC, compared to cells treated with DMSO alone. GR gene expression, measured by qRT-PCR, also showed a significant (p<0.004) dose-dependent increase with increasing exposure to 5-azaC (**Fig. 3b**). Our cell culture data suggests that JEG-3 cells do not express GR unless treated with 5-AzaC, and increasing doses of 5-AzaC resulted in a dose-dependent increase in GR gene expression.

DISCUSSION

Perturbations to placental gene expression may result in the altered metabolic and endocrine function of the placenta and may also affect the placenta's ability to transport water, gas, nutrients, and waste products crucial for the proper growth and survival of the fetus.⁷ Previous findings have reported that DNA methylation marks are stable and accessible for measurement in placenta,²⁴ and that the GR gene is expressed in placenta.²⁵ Taken collectively, our data combined with this previous work suggests that DNA methylation of the GR gene promoter may be important in regulating GR gene expression, and this epigenetic alteration is linked to infant birthweight. Specifically, we have demonstrated that infants considered LGA have significantly higher GR methylation and that birthweight classification as LGA explains a great deal of the variability in methylation of this promoter region. While there is a difference in the percent methylation observed in the placental choriocarcinoma cell lines versus that in tissue, the levels of GR methylation that we observed in tissue are consistent with previous findings in other primary human tissues.¹⁸ As birthweight is a multifactorial outcome as well as a predictor of later life health and disease, this finding begins to provide a mechanistic link between the role of the intrauterine environment on health throughout life.

Previous work has shown that altered methylation status of the NGF1-A consensus binding site within the rat exon 1-7 and analogous human exon 1F of the GR gene can lead to reduced NGF1-A binding, and reduced expression.^{14, 15, 19, 26} Our work is an extension of the previous work focused on the 13 CpG sites in exon 1F^{15, 18}, driven by the hypothesis that the environment may be playing a critical role in determining the DNA methylation status of this gene regulatory region. The NGFI-A binding site occurs

at CpG sites 3 and 4. Our results suggest elevated methylation at these sites in LGA compared to SGA and AGA infants, although the differences were not significant. Methylation at these sites in infant cord blood was previously associated with maternal mood and stress.¹⁸ Although we could not examine these associations in this population, our results suggest that the environment represented by birthweight may have greater impacts at other CpG sites in the 1F promoter than at the NGF1A sites. This difference may be related to differences in the environment influencing methylation, or may be related to differences in susceptibility of the tissues examined. Specific examinations of potential environmental influences on methylation of this region in placenta and other tissues are warranted to better explain this difference. There are also several alternative first exons of the GR involved in gene expression regulation in a tissue specific manner. Future studies aimed at exploring and more completely characterizing the complex regulatory mechanisms controlling GR gene expression are needed in the placenta and in other primary human tissues.

Our *in-vitro* work in human placenta choriocarcinoma cells suggests that methylation of this region of the gene is also associated with reduced gene expression in this cell type, thereby suggesting a relationship between the methylation of this region in human placentas. Our cell culture data suggests that JEG-3 cells do not express GR unless treated with 5-AzaC, and increasing doses of 5-AzaC resulted in a dose-dependent increase in GR gene expression. Because 5-AzaC acts as a non-specific DNA methyltransferase inhibitor, we cannot exclude the possibility that its effects were not directly related to changes in exon 1F, but these results are consistent with previous work performed in other cell types.²⁶ Future follow-up studies can be performed to more
completely characterize the molecular mechanisms underlying control of GR gene expression.

Rich maternal forecasts represented by LGA status can prove to be incorrect if the child is born into a nutrient poor environment. Moreover, high birthweight, especially birthweights that would categorize babies as LGA, is associated with an increased risk for disease in both children and adults. Previous work has shown high birthweight as a risk factor for insulin resistance¹¹ and obesity,¹² as well as for a number of cancers,¹³ and it is clear that glucocorticoid signaling can play a role in these pathologic processes. While normal physiological levels of glucocorticoids and GR are essential for metabolic control, altered glucocorticoid action has also been shown to be associated with a variety of metabolic diseases such as obesity and type 2 diabetes.²⁷ Altered methylation of the GR in the placenta and subsequent dysregulated expression of the GR and thereby altered glucocorticoid signaling may lead to an adult phenotypes, such as glucocorticoid resistance, as well as to increased risks of metabolic disorders, such as Type 2 diabetes and obesity.²⁸⁻³⁰ To date, associations between placental GR methylation and adult metabolic outcomes have yet to be elucidated. Ke and colleagues demonstrated an increase in expression of some forms of GR in the hippocampus of IUGR rats.¹⁷ Our data suggest an increase in DNA methylation of the exon 1F of the GR associated with LGA status, which would be consistent with the findings of Ke and colleagues. More research is needed to identify risk factors that may lead to the developmental outcome of high birthweight, as well as further elucidate the role of the glucocorticoid pathways as well as other critical molecular mediators leading to this phenotype and the associated later onset disease risk including metabolic disorders.

165

In summary, our study is one of the first to show an association between the methylation status of the promoter of the GR gene in human placenta and LGA status. This may serve as an early epigenetic marker of maternal exposures and may be involved in the pathway leading to the adverse health outcomes associated with increased birthweight. More work is necessary to further elucidate the mechanisms involved in this complex and multifactorial pathway and is crucial for better understanding the developmental origins of health and disease.

MATERIALS AND METHODS

Placenta samples. 480 placenta samples were collected at Women and Infants Hospital in Providence, Rhode Island, in accordance with protocols approved by the Institutional Review Boards of both Women and Infants Hospital and Brown University. Samples were collected from women in good physical health between the ages of 18-42, whose infants were at term and viable with no known genetic disorders. Approximately a 1g biopsy of placenta was excised, free of maternal decidua, from the maternal side of the placenta 2 cm from the umbilical cord insertion site within 4 hours of delivery, using a protocol designed to reduce potential sample degradation. The sample was immediately placed in RNAlater (Applied Biosystems, Inc.) and stored at 4°C. At least 72 hours later, placenta samples were removed from the RNAlater, blotted dry and stored at -80°C until further analysis. Medical information, such as infant's birth weight, length, gender, mode of delivery, gestational age, and maternal demographics was recorded from patient medical charts using a structured chart review. Birthweight percentiles were calculated using the method of Fenton (2003).³¹

DNA extraction and modification. DNA was extracted from the placenta samples using the QIAmp DNA Mini Kit (Qiagen, Inc.) following manufacturer's protocols. Purified DNA was quantified using a ND-1000 spectrophotometer (Nanodrop, Wilmington, DE), and DNA samples (1 μ g) were bisulfite-modified using the EZ DNA Methylation Kit (Zymo Research, CA, USA.) and stored at -20°C.

Bisulfite pyrosequencing DNA methylation analysis. Pyrosequencing was performed on PCR product amplified from bisulfite-modified DNA as described previously.¹⁸ Bisulfite conversion was performed using individual columns using the EZ DNA Methylation Kit

(Zymo Research) and manufacturer's protocol. In order for a sample's methylation extent to be called, it must exhibit at least a 93% bisulfite conversion rate, as assessed by pyrosequencing, and all samples examined exhibited a rate >95%. To prevent batch effects from bisulfite treatments interfering with the analysis, samples were randomized across batches. 10% of the samples were repeated independently and the R² for repeats is 0.98. The region of interest, exon 1F of the human GR (*NR3C1*) gene, has 13 CpG sites, and is considered the human homologue of the rat *NR3C1* gene, exon 1-7 previously shown to have differential methylation in response to maternal exposures.^{19, 32} In brief, HotStar Taq DNA Polymerase (Qiagen) and the following forward and biotinylated reverse primers were used: PMHumGCCRF, 5'-

TTTTTTTTGAAGTTTTTTA-3' and PMHumGCCRR, 5'-

CCCCCAACTCCCCAAAAA-3'(IDT Inc., Coralville, IA.) Cycling conditions were 94°C for 15 minutes followed by 50 cycles of 94°C for 1 minute, 55°C for 1 minute and 72°C for 1 minute with a final extension of 10 minutes at 72°C. PCR products were sequenced using a PyroMark MD system and the two following sequencing primers (IDT): PMHumGCCRS2, 5'GAGTGGGTTTGGAGT-3' and PMHumGCCRS3, 5'-AGAAAAGAATTGGAGAAATT-3'. The first sequencing primer was designed to sequence the first five CpG sites, and the second sequencing primer was designed to sequence the following eight CpG sites for a total of thirteen sites sequenced. The dispensation orders for the two assays were

GTCTGTCGAGTAGTCGGTCGAGAGCTATGTCGAG for the first assay and ATCGTGTTGATCTGTCGCTTAGAGAGAGACTATGTCAGTTCTGTCGTAGTCTGTC GTA for the second. The percent methylation at each CpG site was quantified using the Pyro Q-CpG software, version 1.0.9. (Qiagen). A >99% success rate of the pyrosequencing reaction was observed for the samples examined.

RNA extraction. RNA was isolated from cultured cells using the miRvana miRNA Isolation Kit (Ambion, Inc., Austin, TX) following the manufacturer protocol. Extracted RNA was quantified using a ND-1000 spectrophotometer and stored at -80°C.

Quantitative RT-PCR (qRT-PCR). Gene expression was measured using commercially available TaqMan Gene Expression Assays (Applied Biosystems, Valencia, CA) on an Applied Biosystems 7500 Real Time PCR system and analyzed with 7500 System Software. All reactions were run in triplicate, with GAPDH serving as a referent. In addition, a no-RT control was run with each plate.

Cell Culture. The human choriocarcinoma cell line JEG-3 was cultured in Eagle's minimal essential medium (MEM) (ATTC, Manassas, VA), supplemented with 10% fetal bovine serum (FBS) and 1% penicillin/streptomycin. JEG-3 cells were treated with the inhibitor of DNA methylation, 5-aza-2'deoxycytidine (Sigma-Aldrich), using doses of 1.25µM and 2.5µM in DMSO for 7 days, based on previously published studies.³³ Media containing DMSO vehicle only was used as the mock treatment. All experimental and mock exposures were done in biological triplicate.

Statistical Analysis. Differences in covariates between birthweight groups (SGA, AGA, and LGA) were assessed by ANOVA for continuous variables and chi-square test for categorical variables. Pearson correlation coefficients and corresponding p-values were used to examine the correlation of each of the covariates with log GR mean methylation extent. Univariate associations between birthweight status and methylation extent were assessed using the nonparametric Kruskal-Wallis test, with Bonferroni correction to

control for multiple comparisons. Controlling for potential confounders, multivariable linear regression modeling was used to examine the association between DNA methylation (log-transformed to fit the normality assumptions of the model) and birthweight. Covariates were added to the model because of their potential as confounders of birthweight. Two-tailed t-tests were used to analyze gene expression of 5aza-2'deoxycytidine exposed cells compared to control cells. Data were analyzed by SAS 9.1 and R.

ACKNOWLEDGMENTS

Many thanks to Gilda Ferro, Joyce Lee, and Keila Veiga for placenta sample and medical history collection and processing.

FINANCIAL SUPPORT

This work was supported by NIH grants from the COBRE for Perinatal Biology P20 RR018728.

REFERENCES

- Barker DJ, Clark PM. Fetal undernutrition and disease in later life. Reviews of reproduction. 1997 May;2(2):105-12.
- Hales CN, Barker DJ. Type 2 (non-insulin-dependent) diabetes mellitus: the thrifty phenotype hypothesis. Diabetologia. 1992 Jul;35(7):595-601.
- Curhan GC, Willett WC, Rimm EB, Spiegelman D, Ascherio AL, Stampfer MJ. Birth weight and adult hypertension, diabetes mellitus, and obesity in US men. Circulation. 1996 Dec 15;94(12):3246-50.
- 4. Narahara H, Johnston JM. Smoking and preterm labor: effect of a cigarette smoke extract on the secretion of platelet-activating factor-acetylhydrolase by human decidual macrophages. Am J Obstet Gynecol. 1993 Nov;169(5):1321-6.
- Shea AK, Steiner M. Cigarette smoking during pregnancy. Nicotine Tob Res. 2008 Feb;10(2):267-78.
- Esposito ER, Horn KH, Greene RM, Pisano MM. An animal model of cigarette smoke-induced in utero growth retardation. Toxicology. 2008 Apr 18;246(2-3):193-202.
- Maccani MA, Marsit CJ. Epigenetics in the placenta. Am J Reprod Immunol. 2009 Aug;62(2):78-89.
- Sood R, Zehnder JL, Druzin ML, Brown PO. Gene expression patterns in human placenta. Proc Natl Acad Sci U S A. 2006 Apr 4;103(14):5478-83.
- Argente J, Mehls O, Barrios V. Growth and body composition in very young SGA children. Pediatr Nephrol. 2010 Apr;25(4):679-85.

- Ng SK, Olog A, Spinks AB, Cameron CM, Searle J, McClure RJ. Risk factors and obstetric complications of large for gestational age births with adjustments for community effects: results from a new cohort study. BMC Public Health. 2010;10:460.
- Giapros V, Papadimitriou P, Challa A, Andronikou S. The effect of intrauterine growth retardation on renal function in the first two months of life. Nephrol Dial Transplant. 2007 Jan;22(1):96-103.
- Hediger ML, Overpeck MD, McGlynn A, Kuczmarski RJ, Maurer KR, Davis WW. Growth and fatness at three to six years of age of children born small- or large-for-gestational age. Pediatrics. 1999 Sep;104(3):e33.
- Ross JA. High birthweight and cancer: evidence and implications. Cancer Epidemiol Biomarkers Prev. 2006 Jan;15(1):1-2.
- 14. Weaver IC, La Plante P, Weaver S, Parent A, Sharma S, Diorio J, et al. Early environmental regulation of hippocampal glucocorticoid receptor gene expression: characterization of intracellular mediators and potential genomic target sites. Mol Cell Endocrinol. 2001 Dec 20;185(1-2):205-18.
- 15. Weaver IC, Diorio J, Seckl JR, Szyf M, Meaney MJ. Early environmental regulation of hippocampal glucocorticoid receptor gene expression: characterization of intracellular mediators and potential genomic target sites. Ann N Y Acad Sci. 2004 Jun;1024:182-212.
- 16. Herbeck YE, Gulevich RG, Amelkina OA, Plyusnina IZ, Oskina IN. Conserved methylation of the glucocorticoid receptor gene exon 1(7) promoter in rats

subjected to a maternal methyl-supplemented diet. Int J Dev Neurosci. 2010 Feb;28(1):9-12.

- Ke X, Schober ME, McKnight RA, O'Grady S, Caprau D, Yu X, et al.
 Intrauterine growth retardation affects expression and epigenetic characteristics of the rat hippocampal glucocorticoid receptor gene. Physiol Genomics. 2010 Jul 7;42(2):177-89.
- Oberlander TF, Weinberg J, Papsdorf M, Grunau R, Misri S, Devlin AM. Prenatal exposure to maternal depression, neonatal methylation of human glucocorticoid receptor gene (NR3C1) and infant cortisol stress responses. Epigenetics. 2008 Mar-Apr;3(2):97-106.
- Weaver IC, Cervoni N, Champagne FA, D'Alessio AC, Sharma S, Seckl JR, et al. Epigenetic programming by maternal behavior. Nat Neurosci. 2004 Aug;7(8):847-54.
- Turner JD, Schote AB, Macedo JA, Pelascini LP, Muller CP. Tissue specific glucocorticoid receptor expression, a role for alternative first exon usage?
 Biochem Pharmacol. 2006 Nov 30;72(11):1529-37.
- Breslin MB, Geng CD, Vedeckis WV. Multiple promoters exist in the human GR gene, one of which is activated by glucocorticoids. Mol Endocrinol. 2001 Aug;15(8):1381-95.
- 22. Presul E, Schmidt S, Kofler R, Helmberg A. Identification, tissue expression, and glucocorticoid responsiveness of alternative first exons of the human glucocorticoid receptor. J Mol Endocrinol. 2007 Feb;38(1-2):79-90.

- 23. Nautiyal S, Carlton VE, Lu Y, Ireland JS, Flaucher D, Moorhead M, et al. Highthroughput method for analyzing methylation of CpGs in targeted genomic regions. Proc Natl Acad Sci U S A. 2010 Jul 13;107(28):12587-92.
- Avila L, Yuen RK, Diego-Alvarez D, Penaherrera MS, Jiang R, Robinson WP. Evaluating DNA methylation and gene expression variability in the human term placenta. Placenta. 2010 Dec;31(12):1070-7.
- 25. Johnson RF, Rennie N, Murphy V, Zakar T, Clifton V, Smith R. Expression of glucocorticoid receptor messenger ribonucleic acid transcripts in the human placenta at term. J Clin Endocrinol Metab. 2008 Dec;93(12):4887-93.
- 26. McGowan PO, Sasaki A, D'Alessio AC, Dymov S, Labonte B, Szyf M, et al. Epigenetic regulation of the glucocorticoid receptor in human brain associates with childhood abuse. Nat Neurosci. 2009 Mar;12(3):342-8.
- Rose AJ, Vegiopoulos A, Herzig S. Role of glucocorticoids and the glucocorticoid receptor in metabolism: insights from genetic manipulations. J Steroid Biochem Mol Biol. 2010 Oct;122(1-3):10-20.
- 28. Charmandari E, Kino T, Souvatzoglou E, Vottero A, Bhattacharyya N, Chrousos GP. Natural glucocorticoid receptor mutants causing generalized glucocorticoid resistance: molecular genotype, genetic transmission, and clinical phenotype. J Clin Endocrinol Metab. 2004 Apr;89(4):1939-49.
- van Rossum EF, Lamberts SW. Glucocorticoid resistance syndrome: A diagnostic and therapeutic approach. Best Pract Res Clin Endocrinol Metab. 2006 Dec;20(4):611-26.

- 30. Chen P, Jiang T, Ouyang J, Cui Y, Chen Y. Epigenetic programming of diverse glucocorticoid response and inflammatory/immune-mediated disease. Med Hypotheses. 2009 Nov;73(5):657-8.
- 31. Fenton TR. A new growth chart for preterm babies: Babson and Benda's chart updated with recent data and a new format. BMC Pediatr. 2003 Dec 16;3:13.
- 32. Turner JD, Muller CP. Structure of the glucocorticoid receptor (NR3C1) gene 5' untranslated region: identification, and tissue distribution of multiple new human exon 1. J Mol Endocrinol. 2005 Oct;35(2):283-92.
- 33. Pospechova K, Rozehnal V, Stejskalova L, Vrzal R, Pospisilova N, Jamborova G, et al. Expression and activity of vitamin D receptor in the human placenta and in choriocarcinoma BeWo and JEG-3 cell lines. Mol Cell Endocrinol. 2009 Feb 27;299(2):178-87.

	Total	SGA	AGA	LGA	P *
	(n=480)	(n=102)	(n=343)	(n=35)	
Birthweight in g , mean (sd)	3035.6 (671.5)	2369.8 (396.1)	3118.7(553.5)	4161.1 (320.2)	<0.0001
Gestational age in wks, mean (sd)	38.3 (2.1)	38.0 (2.2)	38.3 (2.1)	38.7 (1.2)	0.185
Maternal age in yrs , nean (sd)	28.2 (5.8)	27.6 (6.0)	28.3 (5.8)	28.7 (4.9)	0.488
Infant gender, n (%)					0.102
Male	236 (49%)	43 (42%)	179 (52%)	14 (40%)	
Female	243 (51%)	59 (58%)	163 (48%)	21 (60%)	
Maternal race , n %)					0.215
White	280 (58%)	49 (48%)	209 (61%)	22 (63%)	
Black	50 (10%)	14 (14%)	33 (10%)	3 (9%)	
Hispanic	105 (22%)	27 (26%)	69 (20%)	9 (26%)	
Asian	27 (6%)	8 (8%)	19 (6%)	0 (0%)	
Other/Unknown	18 (4%)	4 (4%)	13 (3%)	1 (2%)	
Alcohol during pregnancy, n (%)					0.153
No	477 (99%)	100 (98%)	342 (100%)	35 (100%)	
Yes	3 (1%)	2 (2%)	1 (0%)	0 (0%)	
Fobacco during pregnancy, n (%)					0.007
No	433 (90%)	84 (82%)	315 (92%)	34 (97%)	
Yes	47 (10%)	18 (18%)	28 (8%)	1 (3%)	

Table 1. Demographics of the study population

	Total	SGA	AGA	LGA	P *
	(n=480)	(n=102)	(n=343)	(n=35)	
Recreational drugs during pregnancy, n (%)					0.511
No	471 (98%)	99 (97%)	337 (98%)	35 (100%)	
Yes	9 (2%)	3 (3%)	6 (2%)	0 (0%)	
Prenatal vitamin use, n (%)					0.249
No	95 (20%)	18 (18%)	74 (22%)	3 (9%)	
Yes	385 (80%)	84 (82%)	269 (78%)	32 (91%)	
Delivery method , n (%)					0.0004
C-section	165 (34%)	34 (33%)	108 (31%)	23 (66%)	
Vaginal	315 (66%)	68 (67%)	235 (69%)	12 (34%)	
Maternal gestational diabetes,					0.202
n (%)					0.202
No	460 (96%)	96 (94%)	332 (97%)	32 (91%)	
Yes	20 (4%)	6 (6%)	11 (3%)	3 (9%)	

Supplemental Table 1. Pearson correlation coefficients and corresponding P-values examining the correlation of each of the covariates with log GR mean methylation extent.

	Correlation coefficient	p-value
Birthweight in g	0.161	0.0004
Gestational age in wks	0.075	0.099
Maternal age in yrs	-0.064	0.158
Infant gender	-0.010	0.833
Maternal race	0.013	0.768
Alcohol during pregnancy	-0.058	0.204
Tobacco during pregnancy	-0.019	0.687
Recreational drugs during		
pregnancy	-0.024	0.606
Prenatal vitamin use	0.017	0.705
Delivery method	0.089	0.052
Maternal gestational diabete	es 0.033	0.468





Figure 1. Comparison of the mean extent of methylation of the 13 CpG sites in the GR exon 1F promoter region in SGA (n=102), AGA (n=343), and LGA (n=35) placentas determined by bisulfite pyrosequencing. Error bars represent standard error. * indicates p<0.05 and ** indicates significant differential methylation following Bonferroni correction (p<0.0036).





Figure 2. Correlation matrix describing pairwise Pearson correlations of methylation status among the 13 CpG sites within the exon 1F promoter region of the GR in all placenta samples (n=480). Numbers in () indicate bp distance of specific GR Position from GR Position 1.

Covariate	Ν	Estimate	Р
AGA Status	327	referent	
LGA status	35	0.40	< 0.0001
SGA status	101	-0.05	0.3504

Table 2. Multivariable linear regression model of log-

transformed mean GR methylation extent

Model is controlled for covariates in table as well as gestational age, maternal age, maternal ethnicity, tobacco, alcohol, and recreational drug use during pregnancy, prenatal vitamin use, method of delivery, infant gender, and diagnosed gestational diabetes.



Figure 3. Examination of the effects of 5-aza-2'deoxycytidine (5-azaC) on placenta choriocarcinoma cell line JEG-3. A) Extent of methylation at CpG sites in GR exon 1F region examined by bisulfite pyrosequencing with increasing doses of 5-azaC. B) Expression of GR mRNA examined by qRT-PCR with increasing doses of 5-azaC.